Buying Time with Climate Engineering?

An analysis of the buying time framing in favor of climate engineering

Zur Erlangung des akademischen Grades einer DOKTORIN DER PHILOSOPHIE (Dr. phil.)

von der KIT-Fakultät für Geistes- und Sozialwissenschaften des Karlsruher Instituts für Technologie (KIT) angenommene

DISSERTATION

von

Frederike Neuber

KIT-Dekan: Prof. Dr. Michael Schefczyk

- 1. Gutachter: Prof. Dr. Gregor Betz
- 2. Gutachter: Prof. Dr. Armin Grunwald

Tag der mündlichen Prüfung: 10.04.2018



This document is licensed under a Creative Commons Attribution-NoDerivatives 4.0 International License (CC BY-ND 4.0): https://creativecommons.org/licenses/by-nd/4.0/deed.en

Foreword

This work is the outcome of three and a half years of research in the DFG priority program SPP 1689 "Climate Engineering – Risks, Challenges, Opportunities?" as part of the project C-E-thics. The main objectives of C-E-thics was to sort, scrutinize and evaluate the main moral arguments about climate engineering. While my colleagues worked, among others, on the trade-off argument, the lesser evil argument, and the argument from political economy, I myself was concerned with the buying time argument – an argument in favor of potential climate engineering deployment. It was a challenging argument in many ways: Firstly, it was challenging to reconstruct a reasonable version of this argument, that adds to and clarifies the current discussion about a potential buying time deployment of climate engineering. Secondly, it challenged my own point of view about climate engineering. While intuitively, I would headlong reject any use of climate engineering, analyzing the buying time argument made me concede that there might be forms of deployment that actually could be beneficial and morally sound, albeit in very strict boundary-conditions. However, those very clear-cut and limited deployment schemes are only morally acceptable when embedded in a comprehensive climate portfolio including fast and far reaching emission cuts. In the end, the need for immediate and drastic mitigation cannot be reduced by the possibility of future climate engineering deployment, nor be postponed.

This thesis has been made possible by numerous people that supported and encouraged me in numerous ways. I would like to thank them, though I cannot possibly mention all of them here.

I thank my doctoral advisor, Gregor Betz, for giving me the opportunity to work in this fascinating project and to deepen research in my main areas of interests: climate, ethics and argumentation theory. He gave this thesis the main direction and the right 'drive'. I also thank him for being very supportive and understanding of my family situation, always knowing the importance of putting first things first.

I wish to thank my research group, the members of the shared research group LOBSTER and the members of the SPP, especially Christian Baatz for supporting me in all respects from discussing my ideas to conference organization to online paper research.

I wish to thank my family for their amazing support. Foremost, I thank my husband, Gregor Neuber. He is a wonderful husband and father, a loving man and an insightful iii

partner in discussing my work. This thesis would not have been possible without his constant encouragement, his constant belief in me and his constant and selfless help in letting me work long hours on the weekends.

I thank my father, Konrad Ott. He gave valuable insights to my thesis, greatly widening my viewpoints, and was always available for late-night near-breakdown phone calls. I thank my mother, Veronika Surau-Ott. Not only did she always show a keen interest in my research and discussed the thesis with me from a fresh and neutral standpoint, she also supported me and my family at home, allowing me to continue writing. I thank my brother, Immanuel Ott, for being a fierce critic and for teaching me to be an even fiercer advocate. I thank my mother in law, Irina Neuber, for being there in the last week of my writing, essentially enabling me to finish my thesis.

Last, but never least, I wish to thank my children, for being understanding, for always cheering me up when I come home, just for being the cutest and most awesome kids in the world, and for constantly reminding me of what is really important in life. I would dedicate this work to them. But... One result of my thesis is the rather bleak prospect, that future generations might have no choice but to engage in climate engineering strategies. We may find ourselves in a situation, in which the policy options we propose violate our own normative standards. This prospect is nothing I would wish to dedicate to anyone.

This thesis is written with a deep love for nature; a love for every single being and for everything there is. It is written with the hope and faith that the current and future generations will be able, against all difficulties, to stabilize temperature rise, to turn the wheel around and to stop polluting and exploiting our planet. I hope to teach my children to be conscious about nature, to be aware of the limits we live within and to be content – virtues that are so dearly needed in the anthropocene. I hope to show them the beauty of nature, the wondrous creatures and the mesmerizing places, from the smallest ant to the highest heights of the mountains – this is what I dedicate to my children.

Table of content

Summary o	f the argument and main results	1
Chapter 1	Introduction	9
Chapter 2	The Climate Engineering Controversy	
Chapter 3	The Buying Time Argument	
Chapter 4	Deployment Scenarios	73
Chapter 5	Evaluation of the Arguments	
Chapter 6	Deepened Analysis of the Buying Time Argument	
Chapter 7	Conclusion and Outlook	
References		

Full Table of content

Foreword	iii
Table of content	V
Full Table of content	vi
Abbreviations and Metrics	viii
List of Tables	ix
List of Figures	X

Summary of the argument and main results	1
Chapter 1 Introduction	9
1.1 The debate on climate engineering in context of climate change	9
1.2 Method	13
1.3 Outline and set-up of this work	17
Chapter 2 The Climate Engineering Controversy	21
2.1 Climate engineering and policy responses to climate change	21
2.2 Mitigation obligation	25
2.3 Main theses and arguments in the moral debate	28
2.4 Challenging the canonical distinction	32
2.5 Sulfate aerosol injection	39
2.6 Bioenergy with carbon capture and storage	43
2.7 Results	47
Chapter 3 The Buying Time Argument	49
3.1 Current state of research	
3.2 Establishing the buving time argument	
3.3 Instantiations and desirability of climate goal G	65
3.4 Results	70
Chapter 4 Deployment Scenarios	73
4.1 Reduced warming rate – buying time for adaptation	73
4.2 Flexible Emission Budget – minimizing pressure for mitigation	79
4.3 Results and complete argument matrix	83
Chapter 5 Evaluation of the Arguments	87
5.1 Plausible instantiations	88
5.1.2 Argument a3	89
5.1.2 Argument a2	94
5.2 Implausible instantiations	95
5.2.1 Arguments a1 and b1	96
5.2.2 Arguments A1, A2, A3 and A4	97
5.3 Lack of knowledge	98
5.4 A special case: Argument B4	. 101
5.5 Results	.107

Chapter 6 Deepened Analysis of the Buying Time Argumen	nt111
6.1 Weak buying time argument	
6.2 General moral constraints	
6.2.1 Hubris argument	
6.2.2 Techno-fix argument	
6.3 Results	
Chapter 7 Conclusion and Outlook	
References	

Abbreviations and Metrics

BECCS	Bioenergy with Carbon Capture and Storage
BT-argument	Buying Time argument
CDR	Carbon Dioxide Removal
CE	Climate Engineering
EJ	Exajoule, measuring unit to indicate energy consumption
GHG	Greenhouse Gases
Gt CO ₂ /yr	Annual emission of gigatonnes of Carbon Dioxide
Mt SO ₂ /yr	Annual emission of megatonnes of Sulfur Dioxide
NET	Negative Emission Technology
ppm	parts per million
RCP	Representative Concentration Pathway
SRM	Solar Radiation Management
SAI	Sulfate Aerosol Injection
W/m^2	Watt per square meter

List of Tables

Table 1. Argument matrix. The rows depict climate goals G; the columns depict portfolio options O. Each entry represents a version of the BT-argument
Table 2. Full table of instantiation of the BT-argument. The colors indicate the status of theargument (plausible, implausible, undecided and the special case of BECCS). In each field,the discussion of the argument is shortly summarized and the implausible premises, if any,are indicated.5
Table 3. Categorization of CE technologies according to Heyward 2013. 34
Table 4. Incomplete Table according to Boucher et al. 2014. Only the definition and examples are reprinted, while the original table of Boucher at al. also includes: Mapping onto existing terminology, Scale of Action, Scale of Impact, Impact on the global Commons, Trans-Boundary or Trans-National Side Effect and Permanence of the Effect35
Table 5. Argument matrix. The rows depict climate goals G, the columns depict portfolio options O. Each entry represents a version of the BT-argument
Table 6. Plausible Instantiations of the BT-argument. 88
Table 7. Plausible and implausible instantiations of the BT-argument. 95
Table 8. Implausible instantiations of the BT-argument. The implausible premises of the respective argument are shown in the fields. 98
Table 9. Plausible, implausible and undecided arguments. Light blue indicates a lack ofknowledge about the scenario, dark blue indicates two cases that are unclear with regard tothe underlying BT-principle
Table 10 (same as Table 2). Full table of instantiation of the BT-argument

List of Figures

Figure 1. Three possible strategies in a climate policy portfolio: mitigation, adaptation and climate engineering. For the latter some areas of research are indicated (Source: Kiel Earth Institute 2015)
Figure 2. Different climate engineering technologies, sorted by SRM and CDR approaches (source: Kiel Earth Insitute 2011)
Figure 3. SAI scheme injected by balloons (source: wikicommons)40
Figure 4. Schematic BECCS cycle (source: Canadell and Schulze 2014)45
Figure 5. Simplified peak shaving/ buying time deployment. The orange line indicates a temporary, modest deployment of CE that declines parallel to CO ₂ emissions53
Figure 6. CO ₂ concentration in the atmosphere will remain near stable or even continue to grow, even if CO2 emissions are cut drastically within this century (source: Van Vuuren 2011).54
Figure 7. The top panel shows the total radiative forcing for RCP4.5, and a radiative forcing profile in which the rate of change is halved. The second panel indicates the suggested SAI (SRM) profile. Effect on global mean temperature is shown in the third panel, indicating the time benefit, and the corresponding injection rates per decade are shown in the last panel (source: Keith and MacMartin 2015)
Figure 8. The top panel (a) shows SAI (SRM) being used for either maintaining constant global mean temperature ('0°C per decade) starting in 2020 or limiting the rate of change to 0.1°C per decade for an RCP4.5 pathway. Panel (b) shows the corresponding solar reduction potential of SRM as well as the length of time over which SAI (SRM) would have to be deployed (source: MacMartin et al. 2014, online version in color)77
Figure 9. The panels show the duration (a) and amount (b) of SAI that would be required to constrain the rate of temperature change for different RCP pathways. For RCP8.5, the amount and duration of SAI is significantly larger than for other pathways. If SAI (SRM) is used to maintain a constant temperature (zero rate of change), the maximum solar reduction is finite for each RCP, but duration is only finite for RCP2.6 (indicated by 'x' for the other RCPs). (MacMartin et al. 2014, online version in color.)
Figure 10. Four CO ₂ emission pathways with identical cumulative carbon emissions over the twenty-first century (a) and their corresponding temperature projections (b). The grey area indicates the uncertainty range (source: Rogelj et al. 2016)
Figure 11. Illustrative emission pathways based on carbon budget of 1600 GtCO ₂ and constant emission reduction rate. Panel (a) shows emission pathways going to zero emission with the respective reduction rate, panel (b) shows results if the emission reduction potential is limited, panel (c) shows the results if negative emissions via BECCS are assumed. Panel (d) summarizes the emission window for 2020, if emissions in 2100 are zero with a maximum reduction rate of 2-3% per year (source: Van Vuuren et al. 2013)
Figure 12. Emission trajectory of scenarios with and without BECCS leading to stabilization at 450 and 480ppm CO ₂ -eq (source: Van Vuuren et al. 2013)

Figure 13. Comparison of the solar reductions required to maintain a 0.1°C per decade rate of
temperature change for different emission pathways (source: MacMartin et al. 2014). See
also Figure 9, Section 4.192

Summary of the argument and main results

Can climate engineering help provide more time for an ambitious mitigation program? And if so, is a buying time deployment of climate engineering morally acceptable? The work at hand means to thoroughly scrutinize this specific argument of the climate engineering discourse – the buying time argument (BT-argument). The point of departure of this research is the notion that climate engineering (CE) is not inherently morally wrong. The guiding question is: Is there a possible buying time deployment of a climate engineering technology absent any general moral constraints?

The discussion enfolds in three parts: The first part (Chapter 1 and Chapter 2) is of introductory character and sets the stage for the debate. The second part (Chapter 3 to Chapter 5) constitutes the focus of the investigation. Here, the buying time argument is translated into a deductively valid premise-conclusion-structure and a strong buying time principle is developed. In particular, this principle implies that deployment of CE is to be finite and should not interfere with mitigation efforts. The BT-argument is discussed in light of four specific climate engineering deployment scenarios, two including sulfate aerosol injection (SAI) and two including bioenergy with carbon capture and storage (BECCS). In one of the scenario pairs it is assumed that CO_2 emissions to remain within the two-degree budget; in the other an overshoot is assumed. The third part (Chapter 6 and Chapter 7) stresses the moral dimension of the depicted deployment scenarios and gives an outlook for further research into policy choices.

A close reading of the current literature encompassing the buying time idea enables the distillation of the underlying implicit assumptions. Those assumptions of the BTargument shape the formulation of an elaborated argument principle:

Buying Time Principle (BT-Principle)

If: i. Climate goal G is desirable, ii. option O leads to climate goal G and is beneficial in so doing, iii. option O only reckons with finite CE deployment, iv. CE in O does not lead to less mitigation compared to mitigation in O without CE, v. there is no option O' which is maximally finite and which leads to G and which is better than O, vi. there are no general moral constraints on option O; then: option O should be adopted.

From this an argument¹ can be formulated.

Buying Time Argument (BT-Argument)

- 1. If: i. Climate goal G is desirable, ii. option O leads to climate goal G and is beneficial in so doing so, iii. option O only reckons with finite CE deployment, iv. CE in O does not lead to less mitigation compared to mitigation in O without CE, v. there is no option O' which is maximally finite and which leads to G and which is better than O, vi. there are no general moral constraints on option O;
- then: option O should be adopted.
- 2. Climate goal G is desirable (desirability thesis).
- 3. Option O leads to the desirable climate goal G (effectiveness thesis).
- 4. Option O only allows for finite CE deployment (finitude thesis).
- 5. CE in O does not lead to less mitigation compared to mitigation in O without CE (no-impediment thesis).
- 6. There is no option O' which is finite and which leads to G and which is better than O (no-better-option thesis).
- 7. There are no general moral constraints on option O (morality thesis).
- 8. THUS: Option O should be adopted.

I introduce two placeholders for the buying time argument: Climate goal G and policy option O. The discourse on CE as a means to buy time incorporates at least four different climate goals G: Two that aim at preventing dangerous climate impacts and two that relate to minimizing the social and economic costs of the inevitable transition towards a carbon-free economy. Those four goals are discussed in light of four policy options O, two for SAI and two for BECCS.

Climate Goal G

Climate Goal 1	Avoiding temperature-dependent tipping points
Climate Goal 2	Avoiding rate-dependent tipping points.
Climate Goal 3	Reducing adaptation pressure.
Climate Goal 4	Reducing mitigation pressure.

Climate Policy Option O

Climate Policy Option a	SAI + CO_2 emissions remaining within the two-
	degree budget (CO ₂ < 2° budget).

¹ An argument with placeholders is called an argument scheme, since its premises are not truthapt. For the sake of brevity, I will use the term 'argument' instead of 'argument scheme'.

Climate Policy Option A	SAI + CO_2 emissions exceeding the two-degree
	budget (CO ₂ > 2° budget).
Climate Policy Option b	BECCS+ CO_2 emissions remaining within the two-
	degree budget ($CO_2 < 2^\circ$ budget).
Climate Policy Option B	$BECCS\ +\ CO_2$ emissions exceeding the two-degree
	budget (CO ₂ > 2° budget).

The lower case letters a and b indicate that emissions remain within the twodegree budget; the upper case letters A and B signify an overshoot scenario. Climate goal G and policy option O are then plotted against each other, which yields a matrix of sixteen instantiations of the BT-argument.

Climate Goal G		Preventing climate tipping		Reducing pressure		
		points				
			Avoiding	Avoiding	Reducing	Reducing
			temperature	rate-	adaptation	mitigation
			- dependent	dependent	pressure	pressure
		tipping	tipping			
		points	points			
Policy Option O		1	2	3	4	
	$CO_2 < 2^\circ$	a	al	a2	a3	a4
	budget					
_	$CO_2 > 2^\circ$	Α	A1	A2	A3	A4
SA	budget					
	$CO_2 < 2^\circ$	b	b1	b2	b3	b4
	budget					
~		D	D1	D2	D2	D.4
ğ	$CO_2 > 2^\circ$	в	BI	B2	B3	В4
EC	budget					
В	1				1	

Table 1. Argument matrix. The rows depict climate goals G; the columns depict portfolio options O. Each entry represents a version of the BT-argument.

Of those sixteen versions, only two generate plausible premises (arguments a2 and a3). A further six instances can be shown to be implausible. Most notably, the sub-cases assuming an overshoot of CO_2 emissions yield either implausible or at least indecisive arguments. I believe I can demonstrate that the BT-argument has plausible premises only if emissions are assumed to remain within the two-degree budget. The remaining eight versions of the BT-argument are undecided in that I cannot pronounce on their plausibility. Scientific findings to validate the respective premises are inaccessible to me. There are two special cases, however, that pose quite a challenge for the BT-argument: Arguments B1 and B4. Those

cases assume BECCS deployment in an overshoot scenario, in which BECCS is used to realize the two-degree target and to reduce mitigation pressure. However, those instantiations violate at least two requirements of the BT-argument. If one still wants to hold on to this scenario as being a buying time instantiation, the respective BT-requirements may need to be attenuated. In particular, the request not to interfere with CO_2 emission cuts may turn out to be absurd for BECCS. If some premises of the initial BT-argument are weakened, the argument instantiations involving BECCS in an overshoot scenario may make for a plausible BT-deployment.

			Preventing climate tipping points		Reducing pressure	
			Avoiding temperature dependent	Avoiding rate-dependent	Reducing adaptation pressure	Reducing mitigation pressure.
			tipping points	tipping points.		
			1	2	3	4
SAI	CO ₂ < 2° budget	a	No-better-option thesis. This climate goal amounts to the two- degree target. If CO_2 emissions remain within the two-degree budget, SAI deployment is superfluous in reaching the two-degree target.	SAI might be able to prevent rate depended tipping points, while mitigation efforts serve to realize the two-degree target.	SAI might be able to reduce the pressure of adaptation, if it lessens the rate of temperature change which could generate more time for ecosystems and human systems to adapt.	Since mitigation is assumed to be ambitious in this case, it is unclear, in how far a lessening of the rate of temperature change might influence mitigation costs.
	CO ₂ > 2° budget	Α	No-impediment thesis If carbon emissions exceed the two-degree budget, SAI would be used in order to stabilize temperature. It would then be used as a substitute for mitigation, impeding mitigation efforts. <i>Finitude thesis</i> . Because SAI would function as a substitute for mitigation efforts, it would have to be used continuously in order to stabilize temperature. Under the assumption of exceeding carbon emissions, SAI would not be finite.			
	$CO_2 < 2^\circ$ budget	b	No-better-option thesis. This climate goal amounts to the two- degree target. If CO_2 emissions remain within the two-degree budget, BECCS deployment is superfluous in reaching the two-degree target.	Whether BECCCS can influence the rate of temperature change, cannot be decided within the scope of this work.	If additional BECCS can influence the rate of change, it might also reduce adaptation pressure.	While emission would remain within the two-degree budget, BECCS could be used to further reduce the time pressure for mitigation. Research for BECCS together with sufficient mitigation has not been reviewed for this research thesis.
BECCS	$CO_2 > 2^\circ$ budget	B	The special case of BECCS. BECCS in light of insufficient mitigation might be used to stabilize temperature change. It might however violate several BT-requirements. A weak version of the BT-argument could incorporate this case.	Especially in light of insufficient mitigation, it cannot be decided here, whether BECCCS can influence the rate of temperature change.	Especially in light of insufficient mitigation, it is unclear, whether BECCS can reduce adaptation pressure.	The special case of BECCS. BECCS in light of insufficient mitigation might be used to reduce mitigation pressure as it enhances the emission budget. It might however violate several BT-requirements. A weak version of the BT-argument could incorporate this case.

Table 2. Full table of instantiation of the BT-argument. The colors indicate the status of the argument (plausible, implausible, undecided and the special case of BECCS). In each field, the discussion of the argument is shortly summarized and the implausible premises, if any, are indicated.

Plausible Versions (green)

Two plausible versions of the BT-argument are been identified: Arguments a2 and a3. In those cases, SAI is deemed to be finite, beneficial and effective, reducing pressure for adaptation while not interfering with mitigation efforts. This form of moderate and strictly purpose-bound SAI may realize the BT-requirements.

This result is somewhat surprising, because SAI especially has been criticized frequently on moral grounds. Nevertheless, a beneficial buying time deployment of SAI might be conceivable within the given boundaries. It is to be noted that the plausibility of this version prominently depends on the effectiveness of mitigation efforts. Only if emissions remain within the two-degree budget, a plausible SAI deployment is possible. Thus, in addition to the moral obligation to mitigate, it can be shown that CE deployment in a BT-framing is only *plausible* with sufficient mitigation.

Implausible Versions (red)

Five additional SAI-instantiations and one BECCS-instantiation can be shown to be implausible. When relating to goal 1, both SAI and BECCS turn out to be superfluous in the respective policy option a and b (argument a1 and b1). All other SAI-scenarios which assume emissions to exceed the two-degree budget, become implausible (arguments A1, A2, A3 and A4) – they violate both the finitude- and as the no-impediment requirement. If this result holds true, it may influence the decision making-process as to warn against a supposed 'buying time deployment' of SAI in light of insufficient mitigation.

Undecided Versions (light blue)

For several scenarios, I cannot provide further scientific back-up. Those BT-versions must remain undecided in the current discussion. Out of six undecided scenarios, five belong to BECCS scenarios (arguments b2, B2, b3, B3 and b4). This might indicate that there is much more uncertainty regarding BECCS than there is regarding SAI, and highlights the need for comprehensive research on BECCS, before it is mutually assumed as a potential future negative emission technology.

The special case of BECCS (dark blue)

A BECCS deployment to artificially enhance the emission budget (argument B1 and B4) turns out to violate at least two requirements of the BT-argument: The finitudeand the no-impediment requirement, and it may very well also violate the morality requirement. Exactly this case, though, is the *raison d'être* for the contemplation and research of BECCS deployment scenarios. Discussion in Chapter 5 adheres to the strong BT-requirements, which lead to a rejection of BECCS in light of insufficient mitigation efforts. Chapter 6 traces another route by asking whether the BTrequirements might be too strong for the special case of BECCS.

Nevertheless, I argue for a strong version of the BT-argument. I do so by stressing one of the initial assumptions of the BT-framing – the inevitable decarbonization of society. I mean to show that while there might be morally acceptable and even necessary BECCS deployment scenarios, those are no instantiations of the BT-argument. Finally, I wish to discuss general moral concerns that address the plausible BT-instantiations and give an outlook for further research.

To summarize, the main objectives and the main results of my research are given in bullet points:

Main objectives

- Introduction of the moral controversy about climate engineering;
- Introduction of two central CE technologies: SAI and BECCS;
- Close analysis of the buying time framing;
- Reconstruction of the buying time argument and discussion of its premises;
- Discussion of SAI and BECCS instantiations of the BT-argument;
- Critique of the BT-argument with two general moral arguments;
- Demonstration of how argument reconstruction helps to structure and evolve a complex debate.

Main results

• Ambitious mitigation is necessary for the BT-argument to be plausible. SAI can only serve as means to buy time, if CO₂ emissions remain within the two-degree budget. Not only is drastic mitigation morally obligatory, but also necessary in order to render the SAI-instantiation of the BT-argument plausible.

- **BT-deployment needs to be finite**. If any CE technology is anticipated to be deployed for a potentially infinite or uncontrollably long period of time, it cannot be called to be in line with the BT-framing.
- The plausible BT-instantiations are not rebutted by general moral arguments. Cases that meet the strong BT-requirements do not fall prey to at least two general moral constraints: The hubris-argument and the techno-fix-argument.
- Case-specific arguments are imperative in evaluating the moral scope of CE deployment. Each technology should be evaluated as part of a comprehensive climate portfolio, where both the extent of its planned deployment as well as the additional climatic options are taken into account.

Chapter 1 Introduction

This chapter gives the framework for the research thesis at hand. First, the significance of climate engineering (CE) will be illustrated in the context of urgent climate policies. Climate engineering is a risky technology that comes with an array of problems - technical, political and moral. The way CE deployment is framed influences the judgment of those problems. Especially if CE is seen only as a temporary stopgap measure that serves to 'buy time' until efficient mitigation politics take hold, CE deployment is likely to be positively assessed (Section 1.1).

The critical examination of the buying time framing in favor of CE deployment constitutes the core of this research thesis. The buying time framing is embedded in a socio-economic discourse. complex political. and moral with many interdependencies to other fields of research. Additionally, research on CE deals with great uncertainties. In particular research on climate engineering has been conducted to a large extend with argumentative analysis contributing to the so-called 'argumentative turn' in policy analysis (Hansson and Hirsch Hadorn 2016a). Argument reconstruction proofs to be a viable tool for certain kinds of political debates that are faced with deep uncertainty. It allows for the assessment of complex decisions absent any clear probabilities or results (Section 1.2). By means of argument reconstruction, the buying time argument (BT-argument) will be formulated and assessed for validity. The set-up and main results of the research thesis at hand will be outlined in Section 1.3.

1.1 The debate on climate engineering in context of climate change

In November 2015, the most recent major climate conference took place in Paris. It was the 21st Conference of the Parties (COP21) to the United Nations Framework Convention on Climate Change. All 197 participating states mutually agreed on the danger and urgency of climate change being the foremost threat to human society. The final *Paris Agreement* (UNFCCC 2016) aims at reducing global warming to "well below 2°C above pre-industrial levels" (UNFCCC 2016, p. 3). In order to reach this goal, global GHG emissions should reach net zero by the end of the twenty-first century, asking for drastic emission cuts in the next decades for both developed and developing countries. The *nationally determined contribution* (NDC), which each

country formulates individually, is a nation's share to realizing the climate goal. The NDCs are expected to amplify every five years and the nations are expected eventually to abandon the use of fossil fuels (UNFCCC 2016, p. 3). At the time of writing (July 2017), 153 out of 197 parties have ratified the commitment – with the astounding exception of the U.S.A, which withdrew from the Paris Agreement in June 2017.

The Paris Agreement marks the preliminary end of the long discussion on anthropogenic climate change. For decades now, but at the latest with the ratification of the Kyoto Protocol in 1997, the need to cut emissions in order to stabilize temperature at 2°C above pre-industrial level has been present to policy makers and the public alike.

However, existing pledges are not sufficient to reach the two-degree target, but rather would allow for a 2.8°C warming (CAT 2017). The progression of the pledges is intended, but meeting the climate goals keeps getting harder if the pledges continue to range at the lowest possible formulation. Moreover, the mechanism to implement the NDCs is argued to be feeble, and the power of the treaty is supposed to rely mainly on a country's good reputation (Jacquet and Jamieson 2016).

A grand, large-scale and global effort to reduce CO_2 emission² is needed in order to meet the commonly agreed climate targets. The first signs of climate change are already visible (IPCC 2014b), some of which are irreversible (Solomon 2009) and both human and natural systems suffer from the impacts. The threat of a 'climate emergency' looms behind current climate predictions, while scientists repeatedly correct their predictions to ever more dire scenarios (e.g. Hansen et al. 2016).

It is in this situation that the idea of climate engineering (also known as geoengineering, climate manipulation or climate intervention³, hereafter referred to as CE) has gained interest. Paul Crutzen has opened up the discussion with his 2006 editorial remark, urging for "active scientific research" (Crutzen 2006, p. 217) on stratospheric aerosol injection (SAI) – a technology that could possibly reduce radiative forcing by injecting sulfur particles into the stratosphere. Climate engineering has since been defined as the "deliberate large-scale manipulation of the planetary environment to counteract anthropogenic climate change" (Royal Society 2009, p. 77). Deployment of climate engineering technologies might avert dangerous

² There are several other Greenhouse Gases (GHGs) that contribute to global warming. Because the main driver of global warming is CO_2 I will focus on CO_2 solely.

³ Even the wording of the technology is subject of discussion, see also Section 2.5.

climate change in face of insufficient mitigation. And with the latest Intergovernmental Panel on Climate Change's (IPCC) report having included CE as an (admittedly theoretical) option to lessen climate change impacts, CE has been hoisted into the realm of acceptable climate policy options (IPCC 2013).

Climate policy options (or portfolios) contain different strategies on how to cope with climate change. They include options to reduce emission (mitigation), adjust to climate impacts (adaptation) and, as of late, technically averting climate effects (CE).



Figure 1. Three possible strategies in a climate policy portfolio: mitigation, adaptation and climate engineering. For the latter some areas of research are indicated (Source: Kiel Earth Institute 2015).

However, the endeavor of climate engineering comes with an array of problems – technologically, politically and morally. With research on CE still being in its infancy, the scope and effectiveness of CE is but a possibility. There are grave, maybe even insurmountable uncertainties about side effects of CE deployment, since both the technologies as well as the climate system are still not sufficiently understood. Social impacts of developing and advertising CE technologies are unclear and there is no viable political framework that could enable a global governance of CE technologies with respect to liability or compensation. The debate

about CE is faced with incomplete and uncertain information impeding robust decision-making.

On a more general note, many philosophers have argued that the enterprise of technically counteracting climate change is as such morally wrong (Robock 2008, Gardiner 2010, Gardiner 2013). Those authors argue, for instance, that pursuing drastic emission cuts is the foremost priority of developed nations, and CE might divert from this obligation. CE then appears to show a hubristic stance towards nature and our place in it, as it offers a cheap techno-fix for our failed emission cuts. Between the support of large-scale CE as a strategy to realize climate goals in the face of insufficient mitigation on the one hand, and the devaluation of CE from a moral point of view on the other hand, an apparent middle ground has been sought. In this view, CE deployment should not amount to an alternative to mitigation efforts. Rather, CE presents itself as a stopgap measure that allows for more time, while in the interim period an ambitious mitigation program will be underway in order to push the decarbonization of society. The deployment of CE could help developed nations to buy time for initiating a successful climate policy. This way, CE might act as a bridge-technology that could be ramped down as soon as mitigation efforts show sufficient effect. This line of thinking amounts to the buying time framing of climate engineering.

The research question of the thesis at hand is this: Are there possible time buying deployments absent any general moral constraints? This question assumes that CE is not inherently morally wrong – if that were the case, CE deployment would have to be rejected no matter the mode of deployment. In contrast, the BT-argument is consequentialist; hence, it assesses the deployment of CE on grounds of its results.⁴ The BT-framing now advocates a limited CE deployment in both its intensity as well as its duration. If the relevant requirements are met, a decent CE deployment seems imaginable. This shows that the framing of CE as a means to buy time makes a positive evaluation more likely. I will follow that thought, while critically examining the underlying assumptions. I will translate the vague buying time framing into an

argument - the buying time argument (BT-argument) - and discuss the possibility of

⁴ The evaluation of the BT-argument has significant intersections to the field of technology assessment. Grunwald (2008, 2009) gives a comprehensive introduction to the method and scope of technology assessment (TA). The thesis at hand, however, uses the method of argument reconstruction and evaluates the premises of the BT-argument thusly.

morally acceptable CE deployment in light of this argument. In doing so, I adhere to what has been called the 'argumentative turn' in policymaking: Debates that are faced with incomplete and uncertain information can be substantiated with the help of argument analysis.

Evaluation of the moral status of a certain technology, however, remains incomplete, if it is not embedded within a comprehensive policy mix that also accounts for a mitigation pathway. Therefor, the thesis at hand uses climate portfolios as instantiations for the BT-argument.

1.2 Method

In both our daily life and politics, many decision problems are characterized by great (or deep) uncertainty: The lack of probabilistic foresight, the lack of information about relevant consequences and the lack of a clear ranking of our values⁵. Climate change and climate engineering are paradigmatic cases for decisions under great uncertainty.

"(...) climate change is associated with conditions of deep uncertainty, where decisionmakers do not know or cannot agree on: (i) the system models, (ii) the prior probability distributions for inputs to the system model(s) and their interdependencies, and/or (iii) the value system(s) used to rank alternatives."

(Lempert et al. 2004, p. 2)

Traditional decision theories (like cost-benefit-analysis) cannot deal very well with great, or deep, uncertainty, because they rely on a concrete value of their input variables. But in political context, there may be 'hard choices' that cannot be resolved unambiguously, having to be made nevertheless.

A way of addressing issues inflicted with great uncertainty lies in argument analysis. Argument analysis is a diverse field of philosophical research that can help clarifying a complex and uncertain situation by making the arguments pro and contra a certain decision explicit. Especially normative principles that underlie certain assumptions are made visible by argument reconstruction. This is obtained by analytical thinking in the best sense: Creating transparency, avoiding ambiguity, and highlighting

⁵ This is not intended to be a definition of 'deep uncertainty', rather a colloquial understanding of what it amounts to.

inferential relations. While there is no final result to be gained from this approach, decision makers might see more clearly what the arguments are about, how to evaluate them in light of their own values and what would follow from them. The argumentative turn has been characterized by Hansson:

"This is a 'widened rationality approach' that scrutinizes inferences from what is known and what is unknown for the decision at hand. It recognizes and includes the normative components and makes them explicit. This is what we mean by the argumentative turn in decision support and uncertainty analysis."

(Hansson and Hirsch Hadorn 2016b, p. 29)

Betz has exemplified this methodological approach to the problem of CE in his 2012 paper on the arming-the-future argument (AF-argument) (Betz 2012a). Betz analyses this prominent argument in favor of CE research and discusses its the premises as well as the objections to it, thereby pinpointing the critical issues at stake. Furthermore, Betz indicates what a proponent of the AF-argument needs to ensure if her argument is to remain plausible (e.g. installing mechanisms that prevent automatic CE deployment).

Elliott has described climate engineering as the "poster child" of argumentative research (Elliot 2016, p. 305), not least because Betz and Cacean (2012) have brought forward a comprehensive argument map on the moral controversy about CE. Their study is a textbook example for how valuable the method of argument analysis can be for the discourse on climate engineering. The many different uncertainties surrounding the research and deployment of CE may be taken hold of by means of presenting them as arguments which themselves assume certain normative principles. That way, the discussion may shift from the empirical question of how to reduce uncertainty to the normative question of how to rationally deal with this uncertainty. Elliot stresses that especially in the case of CE, the ethical discussion precedes the natural scientific one.

"Since there are such pervasive uncertainties associated with climate geoengineering, it is a fool's errand to try to quantify the likely costs and benefits associated with various climate geoengineering schemes with precision in an effort to determine a rational choice. (...) Many of the foundational documents discussing climate geoengineering highlight the necessity of thinking through its ethical ramifications". (Elliot 2016, p. 313) For the BT-argument, argument analysis proves to offer a tool for evaluating different deployment scenarios under an ethical perspective. Certain implicit assumptions, like the inevitable decarbonization by the end of the century (Section 3.1), can thus be problematized and discussed.

An essential part of argument analysis is argument reconstruction. Argument reconstruction aims at translating different claims of a debate into deductively valid arguments and picturing their dialectical inferences. While a debate consists of two features – the dynamic process of giving and taking arguments on the one side, and the static 'freeze frame' of the arguments and their interrelations – argument reconstruction focuses on the latter aspect of the debate (Betz 2010).

Argument reconstruction can either focus on the reconstruction of single arguments, or on the reconstruction of the whole debate. Single argument reconstruction takes the semantic-syntactical inferences of sentences as theses into account and translates them into premise-conclusion-structures (i.e. arguments); debate reconstruction highlights dialectical relations like support and attack between single arguments and theses. Those two tasks are intertwined when considering argumentative texts.

The practice of argument reconstruction is a hermeneutic process of analyzing scientific texts, whose results are deductively valid⁶, semi-formal premise-conclusion-structures, which make their dialectical interrelation visible. Argument reconstruction is also an interpretative task, since the text under investigation usually does not present arguments in a formal manner. Mostly, arguments in texts are found in the form of 'enthymemes' – arguments that are incomplete in that "(...) a premise or the conclusion has been 'left implicit' " (Betz and Brun 2016, p. 50). Additionally, many arguments found in scientific texts cannot be assessed directly, for they contain vague or ambiguous wording. It is the chore of argument reconstruction to transform those incomplete arguments into structures with clear and truth-apt premises. Betz and Brun summarize the practice of argument reconstruction as follows:

"A reconstruction of an individual argument takes an argumentative text as its input and aims at delivering an inference as its output. The guiding principles are the hermeneutic

⁶ Every inductive argument can be transformed into a deductive argument via premise inclusion (Betz 2010).

maxims of accuracy and charity as well as the ideal of clarity with its aspects of explicitness, precision, and transparency." (Betz and Brun 2016, p. 47)

An argument resulting from argument reconstruction should be explicit, precise and transparent: All relevant premises should be made explicit and they should consist in complete, truth-apt sentences. The argument should also be precise, i.e. it should not contain any ambiguous wording or ill-defined concepts, whereby the decision on what concepts need further explication is given by the scope of interpretation and the current context of the debate. On the other hand, certain concepts may be left undefined as to serve as a placeholder. In the reconstruction of the BT-argument, two placeholders are used: Climate goal G and policy option O.

Some concepts are essential to the argument itself, so that one does not understand the argument without understanding those concepts. In such cases, vague or ambiguous wording needs to be as concrete as possible in order to make the argument as strong as possible. In the work at hand, the vague notion of 'buying time' is replaced by the presumably more tangible concept of 'reducing mitigation pressure'.

Finally, the internal structure of the argument needs to be made transparent. The benefit of argument analysis arguably lies in the presentation of a comprehensive and transparent structure of the argument, facilitating a well-founded evaluation and judgment. If the argument itself was opaque in that one could not understand its meaning or its logical structure, the goal of clarity of argument reconstruction might not be achieved.

Arguments relate to other arguments in a debate – they support or attack each other. Those interdependencies can be called 'dialectical relations' and can be defined accordingly⁷:

Support: Argument A supports argument B, if the conclusion of argument A is equivalent to one premise in argument B.

⁷ A normal-language definition of the following concepts is given which have been laid out in detail in Betz 2009, Betz 2010 and Betz 2012b.

Attack: Argument A attacks argument B, if the conclusion of argument A is in contradiction to one premise in argument B.

Independence: If argument A and B neither attack nor support each other, argument A and argument B are independent.

Argument reconstruction gives a preliminary viewpoint of a given debate. A debate is defined as a structure, consisting of theses and arguments with the respective dialectical relations between them. It can be denoted by the Greek letter τ . In the work at hand, the method of argument reconstruction and analysis serves the following purposes:

- Structuring the debate surrounding the BT-argument and ultimately structuring the work itself;
- Formulation of a comprehensive BT-argument and the discussion of its premises;
- Elucidating the thought process behind the formulation of the BT-argument and accounting for some premise variation;
- Evaluation of the BT-argument by means of exemplary deployment scenarios;
- Discussion of possible objections to the BT-argument.

The result of this work will not consist in a definite decision about which deployment scenario of a given CE technology is morally acceptable – this is not what can be gained through argument reconstruction. Rather, I wish to shed some light on *what kind of* considerations are at stake (economic, social, etc.) and *what kind of* assumptions are to be made (substitution, fairness, etc.), if CE deployment is accepted as being a morally acceptable option. Finally, argument analysis allows for formulating two versions of the BT-argument, a weak and a strong one which might be the basis for further political decision making.

1.3 Outline and set-up of this work

This thesis means to thoroughly discuss one specific argument of the climate engineering discourse – the buying time argument. But of course, no argument stands for itself, each argument belongs to a universe of discourse and relates to and draws 17 from other arguments. Taking this into account, Chapter 2 serves to introduce the debate on climate engineering. First, establishing the canonical distinction of solar radiation management (SRM) and carbon dioxide removal (CDR) further deepens the concept of CE. If CE is to be deployed, it will be embedded in a complex policy mix consisting in mitigation, adaptation and (possibly) CE. Thus, CE is discussed as a possible part of a comprehensive climate policy portfolio (Section 2.1).

Having introduced a broad definition of CE, the next section sets the basic corner stones for the moral controversy on CE. In climate ethics, the pivotal moral obligation of the current generation is the obligation to pursue drastic emission cuts (mitigation obligation) (Section 2.2). This obligation is the moral epicenter around which the further debate revolves. In light of this obligation, the moral controversy on CE is further portrayed (Section 2.3). Following Betz and Cacean (2012), the main theses and arguments pro and contra research and deployment of CE are given. The BT-argument, however, does not figure prominently in this early exposition, indicating that the BT-argument has only recently played a major role.

The argument reconstruction of Betz and Cacean (2012) uses the placeholder T for a specific CE technology. The authors reason that some arguments only apply to a certain technology or a group of technologies. Arguments about *CE in general* usually fall short, since the diverse features of individual technologies might not be addressed by a generalizing argument. Hence, the placeholder T ought to be specified when evaluating the arguments. This resonates with the objection that not even the broad categories of CDR and SRM do justice to the technology assessment. In discussing arguments that clearly do not apply to certain (groups of) technologies, I will make a case for a technology- and even portfolio-specific evaluation of the BT-argument (Section 2.4). Consequently, the following investigation will focus on two specific technologies. The subsequent sections will introduce sulfate aerosol injection (SAI) (Section 2.5) and bioenergy with carbon capture and storage (BECCS) (Section 2.6).

Chapter 3 to 5 form the centerpiece of the thesis. First, the current discourse is recapitulated on the basis of a selected corpus of literature representing the buying time idea (Section 3.1). From those quotes, several implicit assumptions of the buying time argument are derived. Two kinds of implicit assumption can be distinguished: Assumptions that relate to a desirable climate goal, and assumptions

that refer to future political and social development. Those assumptions shape the formulation of the BT-argument.

The buying time argument is then translated into a deductively valid premiseconclusion-structure (Section 3.2). The central principle (buying time principle) is designed so as to capture the different implicit assumptions identified in the previous section.

The BT-principle works with two placeholders – climate goal G and policy option O. I identify two overarching climate goals that are associated with the buying time idea: Avoiding dangerous climate impacts and reducing pressure. Those goals are further specified with two respective sub-cases, resulting in four instantiations for climate goal G, the desirability of which is also established in Section 3.3.

In order to evaluate any version of the BT-argument, both placeholders need to be replaced. In Chapter 4, four specific climate engineering scenarios are introduced as instances of policy option O: Two for sulfate aerosol injection (SAI) and two for bioenergy with carbon capture and storage (BECCS), with one scenario of each assuming CO₂ emissions to remain within the two-degree budget, and one assuming an overshoot. The scenarios have been chosen so as to resonate with the buying time idea, and to discuss whether the different climate goals associated with the BT-argument might be realized through option O. The first scenarios depict a moderate SAI deployment in order to lessen the rate of temperature change (Section 4.1). The latter scenarios are BECCS scenarios that technically increase the two-degree emission budget (Section 4.2). The results from Chapter 3 and 4 are connected, and an argument matrix of sixteen fields is established by plotting the four instantiations each of placeholders O and G against each other (Section 4.3). Those constitute the sixteen versions of the BT-argument that are checked for plausibility in Chapter 5.

Chapter 5 then evaluates the different versions of the BT-argument. Surprisingly, only two arguments can instantly be called plausible (Section 5.1): Moderate SAI plus sufficient mitigation could help avoid rate-dependent tipping points as well as help reduce adaptation pressure (argument a2 and $a3^8$).

Other combinations of G and O yield at least one implausible premise, which quite easily can be detected (Section 5.2). The remaining versions are neither obviously plausible nor obviously implausible (Section 5.3). On the one hand, the evaluation of

⁸ For the numbering of the arguments see Summary or Section 4.3.

at least six arguments is impossible in the scope of this work, since they rely on scientific findings that are inaccessible to me.

On the other hand, I identify two versions of the BT-argument (B1 and B4, of which only B4 is discussed), whose status is unclear, because the design of the BTargument itself may be questionable. Both argument versions adopt the prominently discussed case of BECCS as a means to buy time in face of less than sufficient mitigation which turns out to be at odds with at least two buying time requirements: The finitude requirement and the no-impediment requirement. This scenario, however, may be necessary from a practical point of view. The BT-requirements might turn out to be too rigid for this case, hinting at a possible revision of the BTargument.

The next chapter will try to reformulate the BT-argument so as to capture the BECCS scenario in question (Section 6.1). Especially the possibility of acceptable substitution will be further examined. But, as the discussion will show, a weak version of the BT-argument which does not include a finitude requirement and reckons with acceptable substitution is no buying time argument at all. That is not to say, however that the respective deployment scenario is morally wrong – only that it is no buying time deployment.

Finally and lastly, the research thesis at hand scrutinizes general moral constraints on CE deployment. This will be discussed by means of the two plausible SAI-instantiations (argument a2 and a3). The question is: If each of the strong BT-requirements is plausibly met, could CE deployment still be morally wrong? I will focus on two general moral objections to CE deployment – the hubris- and the techno-fix argument.

In the last chapter (Chapter 7), the numerous questions that still remain unresolved are resumed and an outlook on possible areas of further research is given. Those include argumentation theory, virtue ethics and degrowth.

Chapter 2 The Climate Engineering Controversy

This chapter serves to set the scene for the following research. First, establishing the canonical distinction of solar radiation management (SRM) and carbon dioxide removal (CDR) further deepens the concept of CE. If CE is to be deployed, it will be embedded in a complex policy mix consisting in mitigation, adaptation and (possibly) CE. Thus, CE is discussed as a possible part of a comprehensive climate policy portfolio (Section 2.1).

Having introduced a broad definition of CE, the next section sets the basic corner stones for the moral controversy on CE. In climate ethics, the pivotal moral obligation of the current generation is the obligation to pursue drastic emission cuts (mitigation obligation) (Section 2.2). In light of this obligation, the moral controversy on CE is further portrayed (Section 2.3). Following Betz and Cacean (2012), the main theses and arguments pro and contra research and deployment of CE are given.

The argument reconstruction of Betz and Cacean (2012) uses the placeholder T for a CE technology. The authors reason that some arguments only apply to a certain technology or a group of technologies. Arguments about *CE in general* usually fall short, since the diverse features of individual technologies might not even be touched by the argument at stake. Hence, the placeholder T ought to be specified when evaluating the arguments. This resonates with the objection that not even the broad categories of CDR and SRM do the technology assessment justice. In discussing arguments that clearly do not apply to certain (groups of) technologies, I will make a case for a technology- and even portfolio-specific evaluation of the BT-argument (Section 2.4).

Consequently, only two technologies will be of focus in the following investigation. The subsequent sections will introduce sulfate aerosol injection (SAI) (Section 2.5) and bioenergy with carbon capture and storage (BECCS) (Section 2.6).

2.1 Climate engineering and policy responses to climate change

Ever since climate change was recognized a global problem, responses to this problem were categorized as being either part of a mitigation or of an adaptation strategy. The 2007 Assessment Report of the International Panel on Climate Change (IPCC) knows only those climate strategies, as well, and states that "societies can

respond to climate change by adapting to its impacts and by reducing GHG emissions (mitigation)" (IPCC 2007, p. 56). Mitigation efforts can be determined in resemblance to the four Representative Emission Pathways (RCPs) that have been modeled to illustrate emission trajectories and corresponding effects (van Vuuren 2011). RCP2.6, for example, is deemed to be coherent with limiting global temperature rise to two degree Celsius above preindustrial level. But both recent political development as well as natural inertia (many effects of past emissions cannot be reversed (Solomon 2009)) give rise to the belief that limiting global warming by political and social means only is out of reach.

"Even if an aggressive global mitigation program is undertaken, substantial reductions in greenhouse gas levels would not be realized for several decades, and the halting or reversing of some of the detrimental effects already built into the climate system (e.g., ocean warming, ocean acidification, polar ice melting, sea level rise) would not follow for many decades or even centuries beyond that."

(McNutt et al. 2015a, p. 18)

Consequently, the need for other, additional solutions is voiced. One set of technologies to battle global warming is known as climate engineering (CE). Paul Crutzen has openly discussed CE as a third option besides mitigation and adaptation (Crutzen 2006). It has since been advocated as part of a reasonable climate policy portfolio (e.g. Keith 2013, Long 2016).

Climate engineering itself is defined as the "deliberate large-scale manipulation of the planetary environment to counteract anthropogenic climate change" (Royal Society 2009, p. 77). It aims at delaying or even offsetting climate change impacts technically by either manipulating the global mean temperature directly or by removing CO_2 from the atmosphere. According to those two approaches, one can distinguish two kinds of CE-technologies: 1. solar radiation management (SRM) and 2. carbon dioxide removal (CDR).



Climate Engineering

Figure 2. Different climate engineering technologies, sorted by SRM and CDR approaches (source: Kiel Earth Insitute 2011).

1. SRM. In order to control and stabilize the global mean temperature, Solar Radiation Management (SRM)-technologies influence the energy balance of the earth by reflecting the incoming sunlight and thus influence the radiative forcing. Reflecting SRM schemes might be situated in the atmosphere (*reflective aerosols*) or even in outer space (*space mirrors*). While the idea of space mirrors would actually amount to reflective particles orbiting earth, this idea is not very prominently discussed due to technical and financial difficulties. Some more promising strategies include the enhancement of cloud formation (*cloud seeding*), which shield a significant amount of sunlight. However, the physics behind cloud formation is still very little understood which may lead to great uncertainty regarding the side effects of this set of technologies (Royal Society 2009).

Injection of reflective aerosols, mostly sulfate aerosol (*sulfate aerosol injection*, SAI), has been researched into at large. There is quite a corpus of literature that focuses on different aspects of this technology, ranging from the technical and physical aspects to moral issues and governmental challenges. The latest comprehensive overview on the different aspects of this technology can be found in

the National Academy of Science's 2016 study (McNutt et al. 2015a and McNutt et al. 2015b). It summarizes the state of the art on SAI research and is used as the main resource for the thesis at hand.

The common aspect of Solar Radiation Management strategies is their ability to influence the global mean temperature directly. SRM technologies are deemed to be quick and effective (see Section 2.3), but they don't address the root cause of climate change – the concentration of CO_2 in the atmosphere. This is also reason for a mostly negative stance towards SRM technologies, since a symptomatic approach towards climate change is associated with the reprehensive attitude of 'techno-fixing' climate change (see also Section 2.3 and Section 6.2).

2. *CDR*. The second group of climate engineering strategies aims at directly removing carbon dioxide from the atmosphere (Carbon Dioxide Removal, CDR). This may happen through mechanical or technical *carbon air capture*, for example via *artificial trees* or *biochar*, or by enhancing natural CO₂ sinks, such as *ocean fertilization*, enhancement of *oceans alkalinity*, and *afforestation*. A combination of those approaches is the technology bioenergy with carbon capture and storage (BECCS). Biomass is harvested and combusted for energy production, while the resulting CO₂ emission is captured and stored underground. BECCS is seen as a very promising approach in climate policies, and is the second technology that this thesis focuses on.

Even though CDR-technologies treat the root cause of climate change, i.e. the concentration of carbon dioxide in the atmosphere, they too have great influence on natural cycles. It is not clear, for example, how the enhancement of the ocean's alkalinity affects the ecosystem. Afforestation struggles with problems like land use conflicts, influences on wildlife habitat or a trade-off with the earth's albedo (Rickels et al. 2011, Royal Society 2009). Besides these problems, CDR technologies are generally more expensive than SRM schemes, and take much longer to show effect. As such they might not be able to prevent a sudden and catastrophic change in climate.

The best way to manage climate change is arguably a portfolio of different strategies. CE could be part of a climate policy portfolio, adding to mitigation and adaptation (Wagner and Zeckhauser 2012, Rayner et al. 2013, Morrow 2014). However, the portfolio approach to climate politics might oversimplify the problems at stake. Since

CE and mitigation might have interdependencies, and CE might inflict a reduction in mitigation efforts, a purely additive concept of a portfolio might be myopic (Gardiner 2013, Morrow 2014). The difficulties with the portfolio approach will be further addressed in context of allowed substitution (Section 2.2, 3.2, 5.4 and 6.1). For now, it suffices to take CE as another option in a potential climate policy mix.

2.2 Mitigation obligation

The fundamental believe in climate ethics is that future generations have the same moral rights as the current generation. In particular, future generations have the right to have access to an array of lifestyle options equal to (or at least comparable to) our current choices.

One influential model, which argues for equal options for future generations, is the 'equity model' of Edith Weiss (Weiss 1989, Weiss 1992). She coined this model as opposed to the opulence model, in which the current generation raises its standard of living and accumulates maximum wealth by consuming as much natural capital as possible. According to Weiss, the opulence model has serious shortcomings, since it overlooks the long-term effect of nature degradation. The equity model, on the other hand, states that future generations should have at least the choice to pursue the same standard of living as the current one by the same means. For the current generation, this implies a moral obligation to preserve such choices, especially with respect to natural capital.

Regarding climate change, irreversible climate impacts, occurring at so-called climate tipping points, would limit the options of future generations inadmissibly. Tipping points are climatic situations that tilt the sensitive balance of the climate system beyond recovery, with possibly disastrous consequences. Tipping points are characterized by the fact that the initial state cannot be restored, even if the initial parameters were to re-emerge. There is a number of climate tipping points, such as the Amazon rainforest dieback, the loss of the Polar ice packs and melting of the Antarctic ice sheets (Lenton et al. 2008).

Since tipping points may limit the options of future generations and possibly impair their standard of living, they ought to be avoided. If, on the other hand, the twodegree target is realized, there is the chance of avoiding most (if not all) of those tipping points. In this sense, the two-degree target is the quantitative measure to
realize the equity model of climate ethics. It also is the commonly agreed upon climate goal in global climate politics. The Paris Agreement (see Section 1.1) even speaks about stabilizing global warming at "well below 2°C". As of July 2017, 153 out of 197 Parties have ratified the commitment.⁹

At the same time, mitigation – the drastic reduction in GHG emissions – has been the preferred option for realizing the two-degree target. Since its emergence, the IPCC report has highlighted the importance of emission reduction. The Kyoto Protocol, which was launched in 1992, has committed its parties to the reduction of GHG emissions (UNFCCC 1998). National declarations such as the WBGU Guidelines (WBGU 2014) declare mitigation to be the priority in climate protection.¹⁰

The central position of mitigation to achieve the two-degree goal, however, is wavering when other options reach the same goal. While in the 1980s and 1990s discussion on CE was an academic niche, the 2006 article by Paul Crutzen (Crutzen 2006) introduced CE and in particular SAI to the discussion about the best climate policy.

The moral obligation to mitigate is therefore no longer absolute, but must be assessed relative to all other options that achieve the same goal. The obligation to reduce emissions must be defended against SAI and other CE options. This shall be done at this point, thus establishing a *prima facie* obligation for mitigation as opposed to CE. Whether the mitigation obligation still holds in face of other measures, which also reach the two-degree target, depends on whether these measures can replace mitigation. Can SRM and CDR be regarded a substitute for mitigation? Baatz and Ott (Baatz and Ott 2016) define a substitute as follows:

"Y is a substitute for X if Y achieves the same aim as X. (...) Strategy Y is a perfect substitute for mitigation if it avoids all negative climatic effects resulting from GHG emissions, without thereby creating other harms or risks."

(Baatz and Ott 2016, p. 99f)

⁹ In June 2017, President Trump has declined to ratify the Paris Agreement. The consequences of his choice may prove devastating for global climate protection. On the other hand, in July 2017 at the G20 summit in Hamburg (Germany), all other 19 states have agreed to further pursue the Paris Agreement, isolating the US climate policy. What effect this development has for potential CE deployment is unclear.

¹⁰ Other obligations constituted by climate ethics transgress the mitigation obligation, like the obligation to help developing countries to adapt, or to obligation to compensate for current climate damages. Those obligations are connected with the mitigation obligation, but cannot be further addressed in the scope of this work.

The authors argue that SRM and CDR are no substitutes for mitigation, since they either do not eliminate all negative effects of climate change (e.g. ocean acidification), or generate additional risks. In the case of CDR measures, this conclusion may be weakened and the authors admit that some small-scale, moderate CDR may very well be a perfect substitute for mitigation. However, due to the cost intensity and the land use restrictions for these measures (see Section 2.6), large-scale CDRs cannot be considered a perfect substitute. Since CE measures, if they were to be applied, would not substitute mitigation efforts, the moral obligation to mitigate still holds.

Even an optimal portfolio mix cannot relieve the current generation from the basic obligation to drastically mitigate. From this the authors conclude that mitigation is the *primary* obligation. Adaptation, SRM and CDR are *secondary* obligations that result from the inadequate realization of the primary commitment. They follow from the primary obligation; they do not make it obsolete.

Baatz (2016) provides another strong argument as to why mitigation is preferable to CE. He further advances the dilemma-argument (Ott 2012) and argues that, in particular, SAI can lead to a dilemma for future generations, if it is carried out without parallel aggressive mitigation measures. According to Baatz, future generations could find themselves between scylla and charyptis: They would either have to continue to operate SAI and would have to bear the possible negative consequences of its deployment, or they would have to end SAI abruptly and thus would have to experience accelerated climate change. Baatz argues that, the more mitigation is pushed, the lower is the probability for future generations to face this dilemma will be. If the dilemma-argument holds, mitigation is not only a *prima facie* obligation, but is compulsory especially in face of SAI deployment.

Gardiner (2013) has argued in a more fundamental fashion against deploying CE headlong. He has elaborated on the moral 'schizophrenia' which is present in demanding CE to be a moral obligation, promoting it as a viable or single alternative to mitigation, while ignoring the primary obligation to mitigate first. In his view, this moral schizophrenia is reason not to pursue CE development and deployment, but rather focus strongly on the mitigation obligation.

All in all, the mitigation obligation is can be established as an *all things considered* obligation, that holds even in face of possible CE. The reasons for this strong

obligation are the imperfect substitutability between CE and mitigation, the possible unmanageable side effects of CE and the dilemma generated by certain forms of CE, like SAI. From an existential moral point of view, the suspension of the mitigation obligation shows signs of moral corruption, and even schizophrenia (Gardiner 2013). However, this does not define the extent of mitigation. In an ideal world, the twodegree target should be achieved with mitigation alone (if this is still possible¹¹). The mitigation obligation could then be defined quantitatively, e.g. as a certain emission trajectory like the RCP 2.6 (van Vuuren 2011), or as a carbon emission budget (Schellnhuber 2009). In a non-ideal world there are manifold trade-offs to other legitimate political objectives (e.g. economic security, infrastructure development, social order, etc.). So, at this point, only a qualitative determination of the mitigation obligation is adopted. Some qualifiers are used synonymously in the work at hand: fast and far reaching (Baatz and Ott 2016), aggressive, ambitious, drastic, sufficient mitigation, decarbonization, change in modes of production and consumption, and so on.

A determination regarding the timespan of mitigation can be incorporated nevertheless: Mitigation and decarbonization respectively, is a task of the near future and cannot be postponed (passed down) indefinitely. However drastic mitigation may look like, it ought to be realized within the next few decades.

For the continuation of this work, the mitigation obligation is defined as follows:

(Mitigation Obligation) The current generation has the moral obligation to drastically reduce their CO_2 emissions within the coming decades.

This broad definition is the moral starting point of the following investigation. The moral obligation to mitigate is an independent obligation that ought to be met regardless of possible CE deployment.

2.3 Main theses and arguments in the moral debate

The first comprehensive argument-analysis on the moral controversy about CE has been conducted by Betz and Cacean (Betz and Cacean 2012, also Rickels et al.

¹¹ Baatz seems to think that the two-degree target can still be realized with mitigation efforts alone (Baatz 2016).

2011)¹². In their landmark report they map the normative issues surrounding CE. They distinguish arguments regarding the deployment and those regarding the research of CE technologies. Moreover, they use the placeholder T for any specific CE technology. They reason that some arguments only apply to a certain technology or a group of technologies. Arguments about *CE in general* usually fall short, since the diverse features of individual technologies might not be addressed by overarching arguments that necessarily homogenize different technologies. Hence, placeholder T ought to be specified when evaluating the arguments.

The central thesis of the argument map is the claim that CE should be researched into immediately. This claim is backed by three theses: Readiness for deployment is desirable; side effects of research and development are negligible; and there is no alternative to immediate research.

First of all, one major driver behind researching into a new technology is the prospect to deploy it. The desirability of deployment of any CE technology is the necessary condition for its research. In other words: If the deployment was morally wrong or not wanted for other reasons, then why research at all? Consequently, the majority of the moral arguments about CE address the desirability of CE deployment.

Secondly, research on CE should only be carried out, if the side effects of this research are deemed to be negligible. Lastly, it is assumed that in order to have a certain technology ready 'in time' research ought to be carried out immediately.

On CE deployment

A number of arguments support the possibility of deploying CE^{13} . One of them is the lesser evil argument, also known as the arm-the-future argument (Betz 2012a). The lesser evil argument assumes that there might be a future point in time where we would have to choose between dangerous, unabated climate change on the one hand, and the deployment of CE on the other. In this case, CE would clearly be the lesser evil. The deployment of CE then would indeed be desirable (if not unavoidable). Because of this prospect, one should start researching into CE technologies now, in

¹² Unless indicated otherwise, the following chapter is a direct synopsis of Betz and Cacean 2012 and all quotes are from there.

¹³ Note, that there is no literature arguing in for immediate CE deployment. The possible need for deployment and the possible benefits of CE are highlighted in current literature. Still, scientists argue very carefully, only suggesting that CE might eventually be part of a policy mix, if certain criteria are met.

order to have them ready in time and grant future decision makers an additional option to their climate portfolio (arm the future).

This line of reasoning, which is among the most prominent in favor of CE, can be attacked in various ways: For one, it is not obvious that CE will really be the lesser evil. Taking into account the many unclear side effects, it is debatable what is seen as 'less' or 'evil' (Betz 2012a). Additionally, as Gardiner (2010) argues, there are choices that may be tarnished in each case. Intentional climate manipulation might be one of those choices.

The lesser evil argument is akin to the emergency argument. In a climate emergency, where global climate impacts happen fast and intense, CE might serve as a back-up plan, an insurance that serves as a shield against this kind of catastrophe (Betz 2012a, Hamilton 2013, Keith 2013, Uther 2013). Lately, the emergency framing of climate change has been challenged. Invoking an emergency situation in order to justify the deployment of a risk technology might lead to problematic political and social implications (Horton 2015, Sillmann et al. 2015). First of all, it is not clear what would count as an emergency situation and who has the power to define it. If an emergency state was to be declared, power might concentrate in the hands of a few, with the associated risk of abuse (Horton 2015). The legal status and political background of an emergency situation might threaten deliberative, democratic structures. Additionally, as Gardiner (2010) has prominently pointed out, there is no moral obligation to prepare for an emergency situation that we still have the capacity to avoid (or at least reduce).

The rebuttal of the emergency framing also gives rise to the argument central for the thesis at hand: the buying time argument. Instead of preparing for an emergency scenario that we could arguably still avoid, CE could be used as a stopgap measure to buy time until climate politics show effect. This argument will be developed in detail in the following chapters.

Another argument in favor of CE deployment relates to the two-degree target. As said before, political and social efforts may not be enough to stabilize temperature at this level. Hence, the deployment of CE seems necessary. This argument relies on assumptions about the future development of international climate politics, which might be pessimistic. Assuming the necessity of CE because of insufficient mitigation efforts could also turn out to be a self-fulfilling prophecy and enhance a trade-off with mitigation.

A third line of reasoning pro potential CE deployment calls to the monetary benefits of CE in comparison to fast and far reaching mitigation. CE is deemed to be quicker and easier than drastic mitigation; hence deploying it would be economically rational Those kinds of arguments are easily rebutted, for ease and effectiveness are by no means the only issues when considering CE deployment. Negative side effects of CE might seriously outweigh the benefits of quick and easy deployment.

The largest group of arguments against CE deployment consequently relates to potential side effects of deployment. Critics of CE fear unseen and uncontrollable effects on the environment (Robock 2008). Hence, they oppose the deployment and consequently the research of CE technologies. For advocates of research precisely this uncertainty is reason to deepen the research.

Additionally, the deployment of CE might lead to great injustice. Especially with SAI, studies indicate that the ecological impacts will be distributed unequally, potentially harming the world's vulnerable more than the already benefitted. Especially native peoples that contribute the least to global warming might be affected substantially by artificial climate modification. The question of consent becomes essential in such cases (Whyte 2012).

There is also a discrepancy between local and global effect of CE. While globally, a CE technology might turn out to be beneficial, local side effects such as dangers associated with underground storage of CO_2 , or the change in regional precipitation patterns might prove to be harmful for the affected areas. This inequality is morally relevant, if it cannot be compensated for. This again might give reason not to deploy CE.

Taking the numerous side effects into account, CE has been criticized on a general note. CE is seen as a new form of human hubris that aims at controlling nature regardless the consequences. Such arguments claim that CE shows the failure of a people that cannot live on a given environment without destroying it ("fouling the nest") (Fleming 2010, Gardiner 2012). Lastly, religious considerations speak against the deployment of CE, for it is seen as a violation of a God-given order.

On CE research and development

The most prominent argument against research is the so-called moral hazard argument (which can also applies to the possible deployment of CE). In short, it states that the prospect of CE deployment will diminish the efforts to mitigate. From

the moment that CE deployment becomes a real option, mitigation will seem unattractive. In comparison to 'easy and effective' CE, mitigation seems expensive, time consuming and tedious. But since mitigation efforts are arguably the priority in tackling climate change, CE should not be offered as a cheap solution (Corner and Pidgeon 2010, Hale 2012, Lin 2013, Morrow 2014).

Advocates of CE meet this concern by pointing out that CE and mitigation are no alternatives, but rather equal parts of a comprehensive climate policy portfolio. Many arguments in favor of CE assume that any CE deployment will be accompanied by serious mitigation efforts¹⁴. Under this perspective, the assessment of CE depends on the supporting climate strategies like mitigation and adaptation. This is also the reason, why the investigation at hand evaluates the BT-argument in light of comprehensive deployment scenarios, and not with respect to CE alone.

However, as stated before, there are many uncertainties tied to CE deployment. Advocates of CE research argue that those uncertainties can only be resolved through further research. In order to enable a robust political decision about future climate actions, decision makers must be provided with all relevant information. In researching different options for tackling climate change, current and future decision makers would then have more choices of action (arm-the-future argument, see above). A global 'moratorium' on climate engineering research could help enable an informed, consensual decision. On the other hand, there might be irreducible uncertainties associated with CE deployment, which even further research cannot eliminate (Section 6.2.1).

2.4 Challenging the canonical distinction

So far, CE has been discussed as an umbrella concept, with only distinguishing between the two broad categories of CDR and SRM (Section 2.1). But as stated in the previous section, arguments about *CE in general* are usually shortsighted, and may not be applied to individual technologies. Instead, I propose that arguments about the deployment of CE need to be technology-specific and must account for an at least broadly sketched climate portfolio. This is due to two reasons:

1. The terminology about CE might be insufficient and biased.

¹⁴ For an overview on this literature see Baatz (2016).

2. Some arguments do not apply to some (groups of) technologies.

1. Terminology insufficient. CE technologies have been introduced as another strategy to tackle climate change as opposed to mitigation or adaptation. But, as mentioned in Section 1.1, at least CDR technologies are related to mitigation strategies, as they too aim at reducing the CO_2 concentration in the atmosphere. Over-simplifying the group of CE and CDR technologies in particular, might hinder progress in ethical as well as legal discussion. The definition of a technology

influences the legal framework, under which its deployment might be regulated. While mitigation technologies might be regulated domestically, CDR technologies might fall under strict international agreements about nature conservation (Reynolds 2014, Reynolds 2015, Armeni and Redgwell 2015). Hence, it is crucial to define what mitigation strategies amount to as opposed to CDR technologies.

The objective of mitigation is to reduce, limit and stabilize climate change at a certain level. The IPCC defines Mitigation:

"Mitigation is a human intervention to reduce the sources or enhance the sinks of greenhouse gases."

(IPCC 2014a, p. 4)

The above quote mentions two aspects of mitigation: *reduce the sources* and *enhance the sinks* of GHGs. Both parts do not rule out that CDR measures can be seen as part of mitigation. Especially BECCS aims at capturing CO_2 before it enters the atmosphere, and hence can be said to reduce the source of emission. Additionally, biomass acts as a natural sink for CO_2 .

So, it seems crude to frame mitigation and CE (including CDR) as two clearly distinct strategies. They overlap – moreover, depending on the definition, are equal. Another, more promising categorization comes from Clare Heyward (Heyward 2013). She urges to abandon the dichotomy between Mitigation and CE strategies, since the attributes for this distinction are insufficient, too broad and too vague. Heyward especially finds the often-cited Royal Society's definition of CE as "deliberate, large-scale manipulation of the planetary environment in order to counteract anthropogenic climate change" (Royal Society 2009, p. 77) unhelpful when discussing the special ethical issues associated with climate responses.

Her proposal is to distinguish the strategies according to the *point* at which they occur in the "process between emitting GHGs and the loss of human wellbeing" (Heyward 2013, p. 25).

Aim	Avoiding climate change			Avoiding	Responding to
				"dangerous"	dangerous
				climate	climate change
				change	
	Avoiding a given level of		Avoiding	Ensuring that	Providing
	atmospheric GHG		global average	rising	redress for
	concentration.		temperature	temperatures	injuries to core
			increases.	do not impact	interests.
				upon core	
				interests.	
Strategy	Mitigation	CDR	SRM	Adaptation	Rectification
	Reducing	Drawing	Increasing	Improved	Financial
	GHG	GHGs out of	Albedo.	irrigation,	compensation,
	emissions.	the		flood	symbolic
		atmosphere.		defences,	reparation.
				protection	
				against	
				disease.	

Table 3. Categorization of CE technologies according to Heyward 2013.

In this categorization, mitigation and CDR are distinctly separated, as the former aims at reducing CO_2 output *in the first place*, and the latter at minimizing the present CO_2 concentration *ex post*.

Another set of CE definitions has been developed independently, but in close resemblance to Heyward's proposal. The authors (Boucher et al. 2014) argue that an unclear definition of CE technologies might obscure and even prevent political decision-making. They propose a different terminology, in which CDR measures no longer count as mitigation measures, but rather constitute a separate form of climate interventions, and SRM technologies might even fall under adaptation measures. Most notably, they present a separate category of domestic CO₂ removal, which enables the distinction between local and global CDR.

Proposed Name	Short definition	Examples
and Acronym		
Anthropogenic	Initiatives and measures	Improved energy efficiency,
emissions	to reduce or prevent	reduction in production and/or
reductions (AER)	anthropogenic emissions	consumption of goods and services,
	of warming agents into	introduction of renewable energies,
	the atmosphere.	nuclear energy, fossil fuel energy
		with CCS, reducing emissions from
		deforestation and forest degradation,
		emission reductions of BC and ozone
		precursors.
Territorial or	Removal of CO ₂ and	Reforestation, biochar and other
domestic removal	long-lived	means of increasing storage of C in
of atmospheric	greenhouse gases from the	soils, small-scale afforestation,
CO ₂ and other	atmosphere operating	BECCS, CO ₂ air capture and storage
long-lived	within national	in territorial (geological) reservoirs,
greenhouse gases	jurisdictions and little	enhanced weathering (without input
(D-GGR)	consequences outside.	of by-products into rivers or the
		oceans).
Trans-territorial	Removal of atmospheric	Large-scale afforestation, ocean
or trans-boundary	CO_2 and long-lived	alkalinity, enhanced weathering (with
removal of	greenhouse gases from the	input of by-products into rivers or the
atmospheric CO_2	atmosphere operating or	oceans), iron fertilization, injection
and other long-	having consequences	of CO_2 into the ocean.
lived greenhouse	partly or fully across or	
gases (1-GGR)	beyond national	
	jurisdictions.	
Regional to	Intentional modification	Injection of stratospheric aerosols.
planetary	of the Earth's energy	marine cloud brightening, cirrus
targeted climate	fluxes in order to offset	suppression, desert brightening on a
or environmental	climate change at the	large scale, ocean heat mixing,
modification	regional to global scale.	modification to Arctic sea ice.
(TCM)		
Climate	Initiatives and measures	Relocating urban or rural settlements,
change	to reduce the vulnerability	building dykes, air conditioning,
adaptation	of natural and human	agricultural crop choices, reflective
measures	systems to the effects of	crops, whitening of human
including local	climate change. Local risk	settlements on a small scale,
targeted	management.	irrigation.
climate or		
environmental		
modification		
(CCAM).		

Table 4. Incomplete Table according to Boucher et al. 2014. Only the definition and examples are reprinted, while the original table of Boucher at al. also includes: Mapping onto existing terminology, Scale of Action, Scale of Impact, Impact on the global Commons, Trans-Boundary or Trans-National Side Effect and Permanence of the Effect.

Following this terminology, there is a clear distinction between measures that remove CO_2 from the atmosphere and measures that reduce its initial output. This allows to frame negative emission technologies like BECCS as CDR measures (D-GGR in terms of Boucher et al. 2014) as opposed to mitigation strategies (AER). While I will not exactly share the nomenclature of the presented terminology, for my purpose it is important to note that BECCS is not a mitigation approach, but generally counts as a CE technology.

The results so far suggest that a general discussion of CE technologies falls short, since it is not at all clear, what 'CE' technologies are. While there are evaluation patterns that apply to CE in general or to some technology groups, there might be nevertheless single features that are obscured by a generalization. This finding correlates with the decision to use the placeholder T in the argument map introduced above. Betz and Cacean (2012) suggest that the moral arguments about CE can only be fully evaluated, when the placeholder T is substituted with a concrete technology. This goes even beyond a redefined terminology, but instead asks for a technology-specific evaluation of the arguments.

2. Technology-specific arguments. General arguments about CE might obscure certain aspects of a technology or of its envisaged deployment. This is true not only of the technologies themselves, but also of the portfolios in which they are embedded. Each technology should be evaluated in a comprehensive climate portfolio, where both the extent of its planned deployment as well as the additional climate options like mitigation is taken into account. To illustrate this, I will briefly discuss some arguments which are not applicable to certain technologies, or, even if they are technology-specific, do not apply to all possible deployment scenarios.

Emergency-Argument

Some authors see CE as a kind of emergency insurance, in case climate change is faster and more severe than expected (Section 2.3). CE could then help to stabilize temperature and to avert the worst effects. The insurance metaphor is therefore closely linked to the emergency-argument. While we should put all our efforts into preventing a climate catastrophe, the damage could be so immense that we should be prepared for its occurrence.

An insurance measure should quickly and effectively balance out the catastrophe against which it is made, and compensate for its damage. Hence, the focus of authors adopting this emergency-metaphor lies on the controllability and economic rationality of CE measures (Luokkanen et al. 2013). This might be the case especially for SAI, since it is deemed to be easy to apply, cheap in its maintenance costs and rather effective in cooling the earth.

The emergency framing loses its impact when used for CDR measures. Due to the physical inertia of the atmosphere, effectiveness of CDRs is only visible after years or even decades. While during an abrupt temperature rise with potentially irreversible effects, SAI could be used quickly and effectively to stabilize temperature artificially, CDR can only stabilize temperature in the long term. Emergency considerations do not apply to CDR technologies.

Termination problem

If SAI is used as a substitute for fast and far reaching mitigation, CO_2 concentration will continue to rise. Only the effect of high CO_2 concentration, namely the temperature rise, will be artificially influenced and stabilized. An abrupt termination of SAI (for example, due to social and political upheavals) would lead to an accelerated heating enforced by the large volume of atmospheric CO_2 (Royal Society 2009, Robock 2008). This might be disastrous for the climate and the biosphere. Additionally, this situation could put future generations in a dilemmatic situation: Either they would have to decide to continue to operate SAI, with possibly fatal side effects, or they would have to accept accelerated climate change (Ott 2012, see Section 2.2).

The termination problem is obviously coined for a specific case: SAI deployment without adequate mitigation. It is easy to see that SAI does not lead to a termination problem when it is accompanied with drastic mitigation cuts – for example in a buying time scenario.

Furthermore, the termination problem does not apply to any types of CDR measures – for they do address the root cause of climate change by reducing the CO_2 content in the atmosphere. Even if CDR measures were used as an alternative to and a substitute for CO_2 emission reduction, and were to be discontinued, there would be no rapid temperature rise as in the illustrated SAI case.

Dual use

A very special problem is connected to SRM technologies. Due to their influencing the radiative balance, they can manipulate local weather patterns with potential negative side effects. For example, local floods or droughts can be triggered by SAI. This makes SAI and other SRM technologies potential weapons of warfare (Corner and Pidgeon 2010, Goodell 2010). In fact, the idea of weather manipulation as an instrument of warfare goes back a least to the middle of the last century and much of the research on SAI and other SRM technologies is based on military research of similar schemes (Fleming 2010). Weather modification for the purpose of warfare is prohibited by international convention, and the 'rearmament' of these technologies could already be read as a violation of the convention (Corner and Pidgeon 2010). Obviously, however, this specific risk is only present for SRM or SAI. It is difficult to conceive a weapon of mass destruction being derived from long-term CDR measures. That being said, some specific forms of CDR, like ocean fertilization, might very well be used as a form of large-scale pollution with hostile intentions.

Some arguments, on the other hand, apply to all CE technologies, albeit specified to the technology at hand. Consideration of procedural and distributional justice, for example, should be addressed for all CE technologies. But the examination of those issues too is case-different. And while for one technology a fair distribution of its side effects might be found, other technologies might not allow for an equitable solution. Here, too, technology- and portfolio-specific arguments ought to be made.

Taking all this into consideration, I will focus on two specific climate engineering technologies: sulfate aerosol injection (SAI) and bioenergy with carbon capture and storage (BECCS). Also, according to the results of the previous sections, in evaluating the BT-argument, I will concentrate on two different deployment scenarios – or policy mixes – for each technology. Accompanying climate options like mitigation influence the intensity and time frame of the technology-deployment and hence shape its evaluation. Especially, I will make a case-discrimination with respect to mitigation efforts. Before those deployment scenarios will be discussed, the physical background of the technologies is delineated. Sections 2.5 and 2.6 draw substantially from McNutt et al. (2015a) and Rickels et al. (2011). If not indicated otherwise, the depiction of the technologies is a direct synopsis of McNutt et al. (2015a); this also goes for the indication of further literature.

2.5 Sulfate aerosol injection

Scientists have pursued the idea to intentionally manipulate the weather at least since the middle of the 19th century (Fleming 2010). The 'promethean dream' to control the climate has a long history that is partially connected to the military use of 'weather control' (Hamilton 2013). It was the observation of temperature declining after the Mount Pinatubo eruption in 1991 that gave the idea of climate intervention as a way to tackle climate change broader support and eventually led to further research on sulfate aerosol injection (SAI) (Crutzen 2006). Small particles of sulfur in the stratosphere are thought to reflect some amount of incoming sunlight, thereby reducing the radiative forcing and thus stabilizing the ambient air temperature at a certain degree. SAI currently is the most prominently discussed SRM-technology, for it is deemed to be cheap and effective. Many scientists think of it as a 'Plan B'; an insurance if all other efforts to combat climate change fail (see Section 2.3)

Many studies attest SAI the ability to stabilize temperature (McNutt et al. 2015a, Rickels et al. 2011, Royal Society 2009, Wigley 2006, Robock et al. 2009). However, temperature is by no means the only problem of climate change. First of all, since SAI does not change the carbon dioxide concentration in the atmosphere, it cannot address problems caused by it, e.g. the acidification of the oceans (Robock 2008). It will most likely affect global precipitation patterns, thus potentially enhancing droughts or floods in countries already affected by climate change. It is also not clear who will regulate the development, let alone the deployment. Last but not least, precisely because of its ability to serve as a Plan B and a last resort for catastrophic climate change, the danger of lessening mitigation efforts might turn out to be a problem (Section 2.3). For a better understanding of the issues surrounding SAI, the scientific background will now be sketched.

Atmospheric-chemical background¹⁵

Sulfate aerosol injection (SAI) relies on the role that aerosols play in atmospheric chemistry. Aerosoles are particles with tiny diameter that reside in the atmosphere. Depending on the height, they remain in the atmosphere for days or up to years. They

¹⁵ The following depiction is a summary of McNutt et al. 2015a, p 66ff, if not indicated otherwise. 39

interact with the earth's energy balance by either reflecting sunlight or by absorbing it. Both reflecting and absorbing particles have a cooling effect on the earth's climate. Aerosols may occur naturally (dust, soil, bacteria) or may be of anthropogenic origin. Despite their suggested benefit on the earth's energy balance, artificial sulfur aerosols also have health and environmental impacts.

Sulfur aerosols are formed when (natural or anthropogenic) gases containing sulfur are transported into the higher altitudes, namely the stratosphere. They may also be injected directly, e.g. by volcanic eruption or artificially by means of balloon or aircraft. There, sulfur atoms react with oxygen (e.g. from water vapor) to form larger particles (so-called nucleation). Those newly formed aerosols either proceed to ascend and evaporate or they are being transported to the polar regions.



Figure 3. SAI scheme injected by balloons (source: wikicommons).

Benefits and idealized deployment scenarios

Sulfate aerosols can reduce the radiative forcing due to their ability to reflect insolation. Assuming a doubling of the current CO_2 concentration over the next decades, the resulting radiative forcing induced by CO_2 would be about 4 W/m².

Some models assume this amount as the baseline scenario for their setup (Kravitz et al. 2012, Rasch et al. 2008, Robock et al. 2008). Here, a large research gap is evident and models diverge quite greatly depending on their setup. Some models have found, that under the assumption of a doubling of atmospheric CO_2 , an injection of 5 MtSO₂/yr would reduce temperature about 0.3° C to 0.4° C (Robock et al. 2008). Another study that used more advanced microphysical assumptions calculated a decline in temperature that is twice as big for similar injection rates (Jones et al. 2010). Yet other authors who include a more nuanced view on the chemical processes, suggest that in order to reach a 'sulfur burden' reaching a radiative forcing of 4 W/m², sulfur injection of 10 Mt/yr is needed (Kravitz et al. 2012, Rasch et al. 2008). The differences in the result indicate that the effectiveness of the technology depends on several as of yet unclear factors, including the means of injection, diffusion and declining of the particles, atmospheric physical and chemical reactions of the aerosols.

The envisioned medium of SAI is either a gas-mix of water vapor and sulfate or crude SO₂ particles, which influence the lifespan of the aerosols. The most costefficient way to inject SAI into the stratosphere is presumably by means of aircraft, but the technical details of emitting those particles at 25 km height are not to be ignored (Rickels et al. 2011). Generally, injection into the stratosphere (20-25km above sea level) is assumed; the site of injection changes for different model studies. While injection at the tropical stratosphere is proposed by some authors, others suggest a broad global injection at an even higher latitude. Stratospheric sulfate aerosols have different local and global effects. Since they both scatter and absorb sunlight, they are responsible for net global cooling, but may result in local heating. The system of aerosol formation is characterized by small changes having a large effect. Thus, recent research suggests that adequate models are not yet in reach. The same holds for the technical methods of sulfur injection.

It is important to note the status of the scenario assumptions, such as those given above. Some studies assume full-blown SAI as an alternative to mitigation efforts (e.g. Tjiputra et al. 2016), others design their studies against varying RCPs (e.g. Kravitz et al. 2015) with varying deployment rates, others adapt a doubling of the CO_2 concentration (see above). Such assumptions are naught but baselines, against which the effectiveness of certain SAI approaches is measured. It does not mean that the authors of the studies necessarily believe their assumptions to happen or to be the most realistic prognoses about future development in climate politics. Especially do those authors by no means propose using solely SAI to counteract global anthropogenic CO_2 emissions. While one could presuppose certain self-enhancing processes in technology development, which could make the deployment of a newly researched technology inevitable (see Section 2.3), the status of the scenario assumptions does not suggest as much. Another type of SAI scenario employs the idea of using SAI in order to reduce the warming rate, rather than stabilizing temperature. This scenario is discussed in Chapter 4 and serves as a case study for the further investigation, because, unlike the temperature-stabilizing scenarios, the moderate SAI scenario resonates best with the buying time idea.

Effects on the environment

SAI is associated with many environmental effects. I will briefly discuss some of them exemplarily:

Ozone Depletion. SAI deployment offsetting a doubling in CO_2 concentration is estimated to delay the recovery of the ozone layer. However, this effect is thought to be fairly small, significantly smaller than the ozone depletion measured between 1980 and 2000 (McNutt et al. 2015, p. 86f).

Photosynthesis. Since the atmospheric CO_2 concentration would not be reduced by SAI, but in fact would continue to rise due to anthropogenic emissions (in the aforementioned scenarios), certain plants might profit. Excess CO_2 might serve as a 'fertilizer' to plants, increasing their productivity. Moreover, the light diffusing quality of sulfate aerosols would result in a light spectrum more beneficial to most land plants (Rickels et al. 2011).

Precipitation. A decline in global precipitation is seen as one essential side effect of SAI deployment. Due to a higher saturation of the upper atmosphere with water vapor, less water might reach the earth's surface. As a result, rainfall as well as soil moisture might decline in some regions. However, these impacts might be smaller than the impacts experienced under unabated climate change. (McNutt et al. 2015, p. 95f).

Acid Rain. Since sulfur will eventually be removed from the atmosphere via rainfall, the acidity of the soil and the ocean might be affected ('acid rain'). The contribution

of SAI to acid rain, however, as compared to current man-made pollution is thought to be negligible (Kravitz 2009).

Ocean Acidification. The ocean's ability to store CO_2 is largely responsible for its acidification. Acidification means that the pH level of the ocean decreases, thereby influencing the ecosystem and its inhabitants (Raven et al. 2005). While ocean acidification is a result of continued anthropogenic carbon emission, and not of the use of SAI itself, it is still associated with this technology. Since ocean acidification might pose a serious ecologic problem, a technology that does not address this problem, might not represent a viable option.

Public Health. When the aerosols are washed out of the atmosphere, e.g. by rainfall, humans and animals alike might be exposed to them. In how far those particles inflict the respiratory system is still unclear. While some authors suggest that a certain impact is to be expected (Effiong and Neitzel 2016), others deem those impacts to be way below the recommended maximum exposure (Jones et al. 2016).

2.6 Bioenergy with carbon capture and storage

Carbon capture and storage (CCS) is a technology that aims at minimizing CO_2 emissions from large emitters, such as fossil fuel power plants, by preventing CO_2 exhaust to reach the atmosphere. If carbon capture and storage is coupled with biomass production and power generation, this technology is referred to as bioenergy with carbon capture and storage (BECCS). BECCS is dealt with as a promising clean ('green') energy technology which would supply energy demands as well as realize net negative CO_2 emission (McNutt et al. 2015b).

The classification of this technology is not clear-cut. While BECCS has been considered a mitigation technology (Keith 2000, Lin 2009, Royal Society 2009), other authors include BECCS into the climate engineering canon, being a CDR technology (Harrison and Hester 2014, Zhanga et al. 2015). BECCS can be seen as a hybrid between mitigation and climate engineering technologies: Biomass on the one hand can serve as a renewable energy, encouraging the detachment from fossil fuels. CCS, on the other hand, is a potential risk technology that, applied at large scale, can be framed as a CE technology. Especially under the impression of the Paris

Agreement, BECCS – seen as a technical solution to the 'emission gap' (see Section 4.2) – has been framed as a CE approach:

"BECCS which stands for bioenergy with carbon capture and storage, is one example of a carbon dioxide removal (CDR) technology. (...) Like all geoengineering techniques, BECCS is controversial, and its feasibility and potential scale are still uncertain." (Shepherd 2016)

I discuss BECCS as a climate engineering technology for two reasons: Firstly, according to the definition of Heyward (2013) and Boucher (2014) (Section 2.4), any measure to technically remove CO_2 from the atmosphere may count as (a version of) a CDR approach. And secondly, the scale, the aim and the technical uncertainties associated with BECCS raise similar moral issues and concerns as is the case with other CE technologies.

In essence, the actual nomenclature is secondary. What strikes me as important, though, is the acknowledgment of large-scale BECCS as a risky technology with potential planetary impacts and side effects. A deepened discussion of the potential scale of deployment will follow in Sections 4.2, 5.4 and 6.1. The framing of BECCS as a CE technology can highlight other issues and moral problems than would be present when the 'lens' of mitigation is applied. The following paragraphs sketch the scientific background of BECCS and draw substantially from McNutt et al. (2015b).

Scientific background, benefits and idealized deployment scenarios

Bioenergy with carbon capture and storage (BECCS) is an approach that combines renewable energy from biomass with technical (i.e. chemical) CO_2 removal. While During its maturation, biomass assimilates (stores) CO_2 from the ambient air via photosynthesis. During the conversion of biomass for energy use (such as heat, electricity, or fuel), CO_2 is released. The resulting CO_2 exhaust, present as a stream of highly dense CO_2 , can be captured at point source (via chemical or technical absorption) and then be stored in underground caverns. So, in theory, BECCS is a carbon negative energy source.



Figure 4. Schematic BECCS cycle (source: Canadell and Schulze 2014).

Energy Potential BECCS could contribute substantially to the worlds energy supply. Theoretically, the estimated maximum energy potential lies between 50 to 675 EJ/yr. To put this number into perspective, the energy demand of Germany in 2016 was at 13.4 EJ (BMWE 2017). By end of the 21st century, the theoretical maximum contribution of BECCS could reach up to 400 EJ/yr, but is highly limited by the availability of arable land. This makes biomass plantations and food production direct competitors for land. Studies reporting high numbers of BECCS energy potential assume a radical shift in dietary choices away from meat which would free up arable land for energy production (Edmonds et al. 2013). However, there is little indication that changes in food choices will occur in the near term future. On the contrary, global consumption of meat and diary products is continuously rising (Henchion 2014). Given these limits, BECCS might only play an auxiliary role in global energy security.

Mitigation Potential Since the practical effectiveness of BECCS is severely restricted, the cost-optimal level would reside at a level of 12 to 14 Gt CO_2/yr . Beyond this amount, the cost of each Gt CO_2 captured by BECCS would rise steeply due to land limitations. This is but a fraction of the estimated future global fossil fuel emission of 31Gt CO_2/yr . If BECCS were to be used as the single factor to ensure RCP2.6 in light of currently projected emission trajectories, the estimated land-use of food-production and biomass-production combined would rise to up to 2.1 billion

hectares in 2100. To put this number into perspective, today's global cropland amounts to 1.6 billion hectare.

The effectiveness of BECCS is also limited by the potential to process biomass as well as the readiness of effective storing mechanisms. One study suggests a carbon loss of about 50% during the BECCS carbon flow (Smith and Torn 2013). Moreover, the energy input required for capturing, transporting and storing might counterbalance the negative emission potential of BECCS.

The mitigation potential is limited by other factors as well. As has been noted recently, BECCS might create a feedback on the carbon cycle. When carbon dioxide is removed form the atmosphere, the ocean might release CO_2 in order to balance the artificial carbon loss (Chen and Tavoni 2013). Also, soil carbon emissions during and after the plantation of biomass may also reduce the net negative emissions of BECCS (Pourhashem et al. 2016).

Effects on the environment

Transport and storage safety. The gravest side effects of BECCS arguably occur during transport and storage (McNutt et al. 2015b). There are two kinds of harms associated with BECCS: Infliction on the local population near CCS sites due to exposure to CO₂ during the injection, or long-term CO₂ pollution by storage leakage (Medvecky, Lacey and Ashworth 2013). Leak pipelines might cause highly concentrated carbon dioxide pollution. On the other hand, if storage sites are not safe, for example due to seismic activity, leakage may contribute to severe CO₂ pollution of land and sea habitats. A devastating, though natural example of what a CO₂ pollution looks like, gives the 1986 Lake Nyon disaster. Lake Nyon, located in the Northwest region of Cameroon, houses a magma chamber filled with carbon dioxide. In August 1986, the lake released a huge cloud of about 300 000 tons of CO₂ which caused at least 1700 people and 3500 animals to suffocate. The eruption was probably caused by underground seismic activity (e.g. Kling 1987).

Having said that, recent studies suggest that the probability of leakage is fairly low (Wennersten et al. 2015). If so, future decision makers could still decide upon carbon storage.

Land competition and food security. Since BECCS relies on the availability of arable land, the transformation of woodland or grassland into biomass plantations might be

considered a side effect of BECCS. Primary forests or native grasslands have a high biodiversity and contribute to natural carbon sequestration. It is morally relevant which land is used for bioenergy production (Section 5.4). The ethical aspect of land-use is especially important if biomass were to be produced in the global South, where arable land is scarce and the available cropland is essential for self-sufficient, small-scale agriculture. Moreover, if animal rights and species conservation are considered, the conservation of wildlife habitat would influence the acceptability of biofuel plantations.

Moreover, food prices may rise due to the increased demand of cropland. A recent study suggests, however that this effect is reversed, as soon as BECCS influences the carbon price. Subsequently, BECCS might even temper the rise in food prices (Muratori et al. 2016). Additionally, as stated above, if food choices were to change, some pressure on cropland could be reduced. Whether BECCS could be deployed beneficially, is influenced by land availability, food security and other choices regarding mitigation.

2.7 Results

This chapter has served to introduce the moral controversy on climate engineering and to depict two specific technologies that are of focus in the following investigation. Most notably, the moral obligation to engage in fast and far reaching mitigation has been established.

(Mitigation Obligation) The current generation has the moral obligation to drastically reduce their CO_2 emissions within the next few decades.

The mitigation obligation influences the reconstruction of the BT-argument, as will be shown in the course of Chapter 3. Nevertheless, it is an independent obligation that ought to be met regardless of the possibility of CE deployment.

The terminology as well as assessment patterns for CE *in toto* is debatable. The terminology under which a technology is framed has consequences down to its legal status. Regarding the moral controversy, some arguments of the ongoing debate do not necessarily apply to individual technologies or mask their singular features. Especially important seems the distinction between mitigation and CDR techniques

which is not always easy to draw. Taking those difficulties about classification into account, this thesis will argue technology-specific with the focus on two technologies: SAI and BECCS. Building onto the general physical background of those technologies, which has been given above, two specific deployment scenarios will be depicted in Chapter 4, enabling the scenario-specific evaluation of the BT-argument in Chapter 5.

Chapter 3 The Buying Time Argument

Are we running out of time? The effects of global climate change are palpable, yet CO_2 emissions continue to rise globally. In order to launch an adequate mitigation program and to cope with further climatic changes, we arguably need more time. CE technologies are framed as an option to buy time, since they might be able to postpone some climate effects or to artificially remove excess CO_2 from the atmosphere. This possibility gives rise to both approval and critique of CE. Advocates of future CE deployment believe it to be an adequate (if not plain necessary¹⁶) option to avoid harmful climatic effects. For critics, on the other hand, the vision of buying time via CE is just a way of delaying a real solution by means of a quick technical fix.

This chapter is dedicated to the construction and the discussion of the BT-argument. First, the current discourse on the buying time idea is recapitulated by presenting a selected corpus of literature (Section 3.1). From those quotes, several implicit assumptions of the buying time argument are distilled and discussed. The metaphor of 'buying time' is translated into the more tangible terms of pressure and reduction of pressure. That way, the pretentious use of time variables is avoided and the idea of the argument can be captured more directly.¹⁷

The buying time argument is translated into a deductively valid premiseconclusion-structure (Section 3.2). The central principle (buying time principle, BT-principle) is designed so as to capture the different implicit assumptions identified in the before section. The BT-principle works with two placeholders – climate goal G and policy option O. In section 3.3 I identify four climate goals that are associated with the buying time idea and discuss their desirability as secondary climate goals.

¹⁶ Informal discussion in the geoengineering google-group demonstrates that some researchers believe CE to be the only option to avoid dangerous climate impacts, such as the melting of the polar ice caps. Latest call for actively deploying CE: Veli Albert Kallio, 6. Juni 2017 15:58 (https://groups.google.com/forum/#!topic/geoengineering/vcNIf7EgIC4)

¹⁷ I continue to call this specific argument the buying time argument, even though the wording no longer appears in the premises themselves. Resemblance and connection to the current discourse on the buying time argument seems important to me in order to enable further discussion.

3.1 Current state of research

The idea to use CE deployment in order to 'buy time' in face of hesitant mitigation efforts seems rather intuitive. If, however, the exact meaning is to be evaluated, 'buying time' turns out to be a metaphor; a buzzword that promises a prudent, minimally invasive use of CE technologies. But what does it mean to 'buy time'? From whom? How much? And at what price?

It seems that those questions beg the definition of certain time concepts, like time span, or earlier-than relations. But trying to formulate such concepts, ironically, does not quite grasp the original idea of the buying time argument. It might be misleading for the understanding of the buying time framing to try pinpointing the exact amount and definition of time related concepts in the use of CE.¹⁸

- 1. There will be a point in time t_1 , when the CO_2 level in the atmosphere (CO_2^+) will lead to a climate situation Sit⁺.
- 2. Policy mix P might stabilize the atmospheric CO_2 concentration below CO_2^+ by t_2 .
- 3. t_1 might lie before t_2 .
- 4. Policy mix P might not be able to prevent climate situation sit⁺ before t₁
- 5. It is mandatory to prevent a climate situation sit⁺ and policy mix P is the preferred way to do so.
- 6. If Policy mix P might not be able to prevent climate situation sit⁺ at t₁, but it is mandatory to prevent a climate situation sit⁺ and policy mix P is the preferred way to do so, then it might be necessary to engage in different policy mix Q to prevent climate situation sit⁺ before t₁, given that option is morally allowed.
- 7. Policy mix Q is morally allowed only if: a. there are no moral restrains against it, b. the side-effects of this option do not overweight the benefit, c. there is no better alternative and d. Policy mix Q will end and be replaced by Policy mix P eventually.
- 8. Between t_1 and t_2 , the side effects of CE technology T are acceptable compared to the benefit.
- 9. There are no moral constraints against CE technology T.
- 10. Between t_1 and t_2 , there is no better alternative to CE technology T.
- 11. Policy mix Q will end and be replaced with Policy mix P eventually.
- 12. (CL) It might be necessary to deploy CE technology T between t_1 and t_2 .

The problem with those time related premises lies in the evaluation of the placeholders t_1 and t_2 and the subsequent determination of the time span of CE deployment. Should CE be deployed in t_1 or before t_1 ? If so, how much before t_1 ? Also, what does it mean to assess the side effects of a CE technology only between t_1 and t_2 (premise 8, 10)? The argument is not deductively valid, because, among other things, the conclusion asks for temporally limited CE deployment between t_1 and t_2 , but the principle in premise 6 does not include a time limit, but rather asks for deployment before t_1 .

While it would certainly be possible to create a deductively valid reconstruction of the BT-argument in a similar vein as the argument above, I felt that further pursuing the reconstruction actually did not further specify the buying time idea, but instead led to side stages about what time concepts are needed to define a 'prevent-relation' (premise 4), essentially creating a different argument. At this point, I refrained from further pursuing the idea of defining time-specific premises for the BT-argument. Rather, I focused on the goals that are thought to be realized with the help of temporal CE deployment. This then led me to the formulation of the pressure terminology, which I deem to better represent the idea behind the BT-argument. The fuzzy notion of time-concepts is captured via the finitude thesis, that asks for limited, rather short and controllable CE deployment (Section 3.2, Section 5.).

¹⁸ An earlier version of my BT-argument reconstruction was this:

Rather, since 'buying time' is a metaphor, it makes sense to decipher it into more tangible terms. In order to achieve this, I will analyze a selected corpus of articles using the buying time metaphor. It will turn out that the buying time framing can be translated into a 'reducing pressure' framing.

While I wish to substantiate the BT-argument for two specific technologies, the quoted literature refers to different (groups of) CE technologies. So, in advancing the premises of the BT argument, I continue to refer to CE in general, and will cut the BT-argument for the specific technologies in Chapter 5.

Averting dangerous climate impacts

The most prominent benefit of a future CE deployment is the prospect of avoiding dangerous climate impacts. In a climate emergency situation, CE might be able to help avert catastrophic climate effects, such as the reaching of tipping points (see Section 1.1). Having said this, the emergency framing of CE deployment has been criticized as of late (see Section 2.3). Recent literature argues to abandon the emergency framing altogether (Sillmann et al. 2015, Horton 2015). Instead, a peak shaving deployment to buy more time is outlined.

"The most common alternative [to the emergency framing], known as 'peak shaving', would entail application of SAI or some other type of solar geoengineering to reduce the worst effects of peak GHG emissions while implementing an ambitious program of mitigation and adaptation. One variant of peak shaving would involve a gradual rampup of solar geoengineering until a modest plateau is reached; as emissions declined, so too would injections of stratospheric aerosols (...). Another potential approach would utilize SRM on a regional scale, for example in the Arctic, in order to address specific, localized damages from climate change."

(Horton 2015, p. 4f)

With the help of a CE technology, the worst impacts of temperature rise could arguably be averted. CE technologies might

"...be used as a stop-gap measure to buy time for a societal transformation to a carbonfree economy, shaving off the worst effects of climate change along the way" (Schäfer et al. 2014, p. 5). The peak-shaving framing assumes that the worst effects of the GHG-peak would indeed be catastrophic. Especially climate tipping points might have grave impacts on natural and human systems and ought to be avoided (see Section 2.2) – if need be with climate engineering:

"A popular framing is that sunlight reflection could "buy time" for decarbonizing the economy and allowing greenhouse gas concentrations to stabilize and then come down." (Lenton and Vaugham 2013, p. 2)

"Some geoengineering measures appear to offer humanity the ability to shave the peaks off CO_2 driven emissions and avoid tipping points. Wigley (2006) argues that sulphate particle injection might be used to "buy time" in reducing CO_2 emissions. This potential to shave peaks would go some way to addressing the concern voiced by some that temperature rises over the next century may exceed irreversible "tipping elements" in the climate system leading to drastic changes with potentially catastrophic impacts on human and natural systems."

(Rayner et al. 2013, p. 8)

"An optimal outcome would seem to be that mitigation can be accomplished relatively rapidly and relatively easily, such that geoengineering might need to be implemented for only several decades in order to shave off the peak change in climate and increase the likelihood that the most adverse and irreversible environmental and societal consequences will not be triggered."

(MacCracken 2009, p. 11)

This 'peak-shaving quality' is found in many CE technologies, especially SAI; they are thought to be used temporarily in order to avert some serious effects. Those technologies then might represent a quick, easy-to-control tool that could be ramped up and down relatively simple.

While the emergency framing assumes deployment only in case of devastating climate impacts, the buying time framing advocates a *preventive* deployment, as to avoid those impacts in the first place. Huttunen and Hildén (2013), who have also collected and analyzed publications about CE, have identified several frames for CE deployment. They especially distinguish the 'technological fix' and the 'last resort' frames for CE, and indicate that a buying time deployment would be an alternative to a possible emergency deployment:

"In the 'Technological fix' subframe, represented by 27 articles, geoengineering is presented as a possible third option beside mitigation and adaptation. It could be used to buy time for mitigation or used side by side with mitigation. None of the analyzed papers suggests that geoengineering should completely replace emission reductions. The key difference to the 'Last resort' subframe is the argument that geoengineering could be used before any emergency situation arises, if justified by costs estimates." (Huttunen and Hildén 2013, p. 13)

Those quotes are paradigmatic for the discourse on CE as a means to buy time. The idea of a peak shaving deployment that those authors present, can be illustrated like this:



Figure 5. Simplified peak shaving/ buying time deployment. The orange line indicates a temporary, modest deployment of CE that declines parallel to CO_2 emissions.

This simple idea and *a fortiori* the above figure are, however, spurious¹⁹. They assume that the worst climate impacts coincide with the peak of GHG emission: "reduce the worst *effects of peak GHG emissions*" (Horton 201, p. 5, emphasis added [FN]). But climate impacts result from the CO₂ concentration in the atmosphere, not the act of emitting. Since the concentration of CO₂ will not necessarily go down as soon as we stop emitting, dangerous climate impacts may occur even after we have stopped emitting CO₂ (see Figure 6 below).

¹⁹ I mean not to devalue the work of the cited authors in general, but rather wish to highlight the potential misconception underlying an advocacy of the BT-argument.



Figure 6. CO_2 concentration in the atmosphere will remain near stable or even continue to grow, even if CO2 emissions are cut drastically within this century (source: Van Vuuren 2011).

The notion to use CE until our CO_2 emissions goes down is misleading – CE as a peak-shaving measure would have to be used until CO_2 *concentration* goes down, which, as indicated, might considerably lag behind our mitigation efforts.

It could be argued that this is a harmless misunderstanding and that one similarly could advance a peak-shaving argument that relates to peak concentration. However, I think that this is the first misconception about the BT-argument, which makes it so appealing. If the stopgap quality is associated with CO_2 emissions, the time-span of CE deployment would arguably not surpass a few decades for all RCPs except RCP8.5. But if the argument was transparently related to the CO_2 concentration, the anticipated duration of CE deployment would rise significantly. In order to balance out high CO_2 concentration, CE would have to be used well beyond of the end of the century for all RCPs except RCP2.6 – a fact that highly influences the moral, political and technical evaluation of the technology.

But of course, the act of emitting carbon dioxide does influence the atmospheric concentration. Insufficient cuts in CO_2 emission would inflate the time span even for a moderate buying time deployment (MacCracken 2009). In order to make the BT-argument as strong as possible (which is the aim of this investigation), the relation between CO_2 emissions, CO_2 concentration, mitigation efforts and temperature rise must be made as transparent as possible. I will attempt to do so by specifying the deployment scenarios with regard to different emission pathways (Chapter 4).

But even this first plunge into the topic indicates that framing CE deployment as a means to buy time or to shave the peak, would still amount to a deployment for

several decades, possibly even beyond the end of the century – nothing that can be controlled or implemented easily.

Managing transition, Transition anyway

When searching for buying time strategies in face of climate change, Nico Bauer (2005) has presented carbon capture and storage (CCS) as a possibility to buy time. His work is related to the thesis at hand, as it has a similar research question ("*Is CCS an option to buy time?*") and deals with a similar technology. In the introductory section of his work, Bauer defines his research question.

"Is CCS an option to buy time in order to shift the climate induced transition to a renewable based energy system? The question assumes that the shift towards renewables is inevitable in the long run, if fossil fuel scarcity induces this transition. The pressure from the climate system leads to a premature transition and CCS could defer this prematurity towards its *natural* timing. \Box

(Bauer 2005, p. 4, emphasis in original)

Bauer enfolds the metaphor of buying time by highlighting the discrepancy in time between a "premature ²⁰ and a "natural transition of society. Nevertheless, Bauer assumes that there will be a transition of society away from fossil fuel – latest when fossil fuel reservoirs are exhausted. There is no alternative to transitioning towards a low-, and eventually no-carbon society. But how will this transition happen? Will it take place in a manner that is manageable and bearable for our society or will it happen abruptly and at high cost? CCS and *mutatis mutandis* some CE technologies like BECCS might help to manage this transition.

From an early stage on, several authors have stressed that a gradual mitigation program has many advantages (Wigley et al. 1996, Wigley 2006, Nordhaus 2008), and CE might be able to help phase in ambitious mitigation. The Royal Society formulates the buying time argument similarly:

"The length of time required to phase out fossil fuels, and to modify the various global human systems contributing to climate change, may be longer than the time available to

²⁰ The word 'premature' suggests some sort of oblivion. Climate change seems to have stricken quite without warning, and society is now faced with the impossibility to change at short notice. This, however, is not the case. The danger of a rise in temperature has long been known, and transition now is by no means premature, but rather overdue.

avoid serious adverse impacts. In this case, SRM might have the potential to temporarily stabilize the global temperature and its associated effects, while providing time to reduce GHG emissions."

(Royal Society 2011, p. 19)

CE technologies (or SRM specifically) might help to phase in efficient mitigation by reducing the pressure of immediate action. In this sense, CE technologies are a way of managing the transition of society towards decarbonization; CE is an auxiliary tool in climate politics. The BT-framing presupposes that decarbonization is indeed a political goal, moreover, a globally shared political goal

Reducing pressure of mitigation and/or adaptation

The inevitable transition towards carbon-free production and consumption should arguably come at the lowest possible social and economic costs. This is part of the rationale that David Keith gives in his 2013 book (Keith 2013). Keith argues prominently for a prudent SRM deployment, partially because of its cost-effectiveness. In a similar vein, Wigley (2006) has offered a calculation of a policy mix consisting of mitigation and CE that proves to be less costly than rapid mitigation alone:

"A combined mitigation/ geoengineering approach to climate stabilization has a number of advantages over either alternative used separately. A relatively modest geoengineering investment (...) could reduce the economic and technological burden on mitigation substantially, by deferring the need for immediate or near-future cuts in CO_2 emissions. More ambitious geoengineering, when combined with mitigation, could even lead to the stabilization of global mean temperature at near present levels and reduce future sea-level rise to a rate much less than that observed over the 20th century (...)" (Wigley 2006, p. 454)²¹

As important as mitigation might be, an additional CE deployment could "provide additional time to address the economic and technological challenges faced by a mitigation-only approach" (Wigley 2006, p. 452). Since some CE technologies could arguably be launched slowly and as easily be tapered off again, so the

²¹ Note that Wigley assumes permanent low SRM. His results can still be used to illustrate how economic pressure of mitigation can be reduced via additional CE deployment.

argument goes, it could be a form of adjustment to correct and support the progress of Mitigation.

"Even absent any climate emergency, CE can be used as part of a low-cost intertemporal climate response by deploying it incrementally to shave the near-term peak of global heating that would occur this century even with aggressive emission cuts. Such an incremental deployment, phased in then out over a century or so, could reduce nearterm climate disruption and associated risks, thereby buying time and allowing emission cuts and adaptation measures to be made in an orderly program of technology deployment and capital turnover, at much lower cost and disruption than under more rapid deployment."

(Parson and Ernst 2013, p. 318)

It is the core assumption of the BT-argument that transition will happen anyway. How this transformation will come about, is left tacit. But it is mutually seen as a costly process, and CE deployment could help to minimizing the costs of transition.

"Due to the high costs arising from the necessary transition to a carbon neutral economy, all options that could contribute to reducing the costs of this transition should be explored."

(Zürn and Schäfer 2013, p. 6)

The buying time metaphor is used to describe a management of an inevitable decarbonization of society while reducing its costs.

"Catastrophic climate change would likely unfold over a number of centuries, but avoiding it will require a technological revolution, and geoengineering might help to "buy time" to develop and diffuse these new technologies" (Barrett 2008, p. 50)

In summary, this is what I deem to lie at the heart of the buying time idea: Shifting away from fossil fuels and towards a net-zero emission economy is challenging. CE technologies might render this transition easier, less costly or less abrupt, thereby taking some pressure off immediate choices of action.

One central assumption of the BT-argument is the prospect of society developing towards a low- or even no-carbon economy (see above). If production and consumption start being carbon neutral, CO_2 emissions will eventually decline also. Then, the argument goes, CE deployment might no longer be needed. So, the assumption of the inevitable transition of society implies that CE deployment be only temporal. The whole idea of the BT-argument is that deployment of CE will be finite and purpose-bound:

"... geoengineering might need to be implemented for only several decades ..." (MacCracken 2009, p. 11)

"... at least among policy makers, nobody believes that geoengineering offers anything but a relatively short stopgap to buy time for other action." (Bunzl 2009, p. 2)

These quotes state explicitly that a BT deployment of CE would have to be limited in time. The authors seem to be quite optimistic about the "relatively short" deployment period, which, as indicated, might inflate when drastic emissions cuts are further delayed (see above). It is nevertheless salient to the BT-argument to assume that, at some point, CE deployment will no longer be needed. This is indeed the very meaning of a stopgap: being limited in time.

Summary

Four key elements of the buying time framing have been derived from the selected quotes of the ongoing discourse:

Averting dangerous climate impacts Managing transition, Transition anyway Reducing pressure of mitigation and/or adaptation Temporal deployment only

From those assumptions, two relate to potential climate goals that are to be realized with the help of CE: *Averting dangerous climate impacts, Reducing pressure of mitigation and/or adaptation.* One defines the envisaged mode of CE deployment:

Temporal deployment only, which is backed by assumptions about future political and social development: *Transition anyway.*

The idea of the BT argument can be captured, when the desired outcome of CE deployment and its relation to other climate policy goals is at the center of the argument reconstruction. Two possible climate goals associated with a stopgap deployment of CE will be adopted for the further investigation:

- 1. Avoiding dangerous climate impacts
- 2. Reducing pressure

Having those goals established, a comprehensive argument can now be formulated which includes prudent requirements for a goal-orientated buying time deployment.

3.2 Establishing the buying time argument

In essence, the BT argument is a consequentialist argument: A certain means should realize a certain end. For the BT argument, this consequentialist principle reads: A certain climate policy mix should be adopted in order to reach a pre-defined climate goal. However, there are some restrictions to the use of CE that originate from the buying time framing, as has been illustrated in the section above. The special *buying time requirements* shape the *buying time principle*, which underlies the *buying time argument*. If a climate policy mix incorporates the BT-requirements, it can be said to be a *buying time deployment*.

For a comprehensive climate portfolio, placeholder O is introduced, while placeholder G denotes the desired climate goal. Both the placeholders and the requirements shape the form of the buying time principle. It can be formulated as follows:

Buying Time Principle (BT-Principle)

If: i. climate goal G is desirable, ii. option O leads to climate goal G and is beneficial in so doing so, iii. option O only reckons with finite CE deployment, iv. CE in O does not lead to less mitigation compared to mitigation in O without CE, v. there is no option O' which is maximally finite and which leads to G and which is better than O, vi. there are no general moral constraints on option O; then: option O should be adopted.

The BT-argument, which incorporates the BT-principle, can be established: 59

Buying Time Argument (BT-Argument)

- BT1. If: i. climate goal G is desirable, ii. option O leads to climate goal G and is beneficial in so doing so, iii. option O only reckons with finite CE deployment, iv. CE in O does not lead to less mitigation compared to mitigation in O without CE, v. there is no option O' which is maximally finite and which leads to G and which is better than O, vi. there are no general moral constraints on option O; then: option O should be adopted.
- BT2. Climate goal G is desirable (desirability thesis).
- BT3. Option O leads to desirable climate goal G (effectiveness thesis).
- BT4. Option O only allows for finite CE deployment (finitude thesis).
- BT5. CE in O does not lead to less mitigation compared to mitigation in O without CE (no-impediment thesis).
- BT6. There is no option O' which is finite and which leads to G and which is better than O (no-better-option thesis).
- BT7. There are no general moral constraints on option O (morality thesis).
- BT8. THUS: Option O should be adopted.

The BT-principle consists in six requirements (i. - vi.), which seem essential to me when reconstructing the buying time idea. I will refer to them as *theses*, since they can be stated as complete, truth-apt sentences. Each thesis states a necessary condition; together they are sufficient for the BT-principle to be true. The characteristics of the buying time argument are captured quite adequately in this formulation of the argument, as will become clear in the following discussion of the BT-requirements.

i. Climate goal G is desirable (desirability thesis).

The desirability thesis states that climate goal G ought to be politically desirable. If the climate goal is not desirable, there is no point in implementing a process that realizes this goal. Instead, a different goal and hence a different policy mix would then be the better choice. Section 3.3 will further discuss this thesis along with the respective climate goals.

ii. Option O leads to climate goal G and is beneficial in so doing so (effectiveness thesis).

Once the desirability of climate goal G has been established, the effectiveness of climate option O must be guaranteed. Whether option O really leads to climate goal

G can only be decided via scientific research. CE research currently consists in computer-based model simulations, which aim at validating claims about efficiency and effectiveness, as well as taking some side effects into account. The kind of premises that one would expect in support of the effectiveness thesis are predictions about the future, obtained by computer-generated simulations.

The validity and scope of those predictions are limited, though. For one, model simulations are only as good as their scenario assumptions. If implausible or incomplete assumptions enter the scenario, the results might also be dubitable (Dieckhoff and Leuschner 2016). Additionally, scenarios generated by simulations are only *possible* states of the world. The probability of those scenarios can only be estimated (Betz 2015). The question is, what degree of probability would suffice for a CE deployment to be called effective? Would we want a 99% probability or would we settle for 75%? And how could those numbers possibly be obtained?

Here it is important to highlight the status such scenario assumptions have in the current discussion. Scientists engaging in CE research do not necessarily believe their assumptions to be the most realistic prognosis about future development in climate politics. In particular, those authors by no means propose SAI or BECCS as an alternative to mitigation efforts, even if they might calculate the effectiveness of those technologies for a business-as-usual scenario.

Lastly, there might still be insurmountable uncertainties associated with CE deployment (Betz and Cacean 2012). There might be side effects that we simply cannot anticipate through model studies, and which might turn out to be disastrous. Those insurmountable uncertainties could be a reason not to engage in CE at all (see also discussion of the irreducible uncertainties argument in Section 6.2.1).

The effectiveness thesis asks for option O to be beneficial as well. Option O is beneficial if the benefits of this option should outweigh its negative side effects. This simple request might prove to be the hardest as well as the most important to evaluate. The problem evidently lies in the formulation and evaluation of a costbenefit relation in case of CE, accompanied with the immediate critique as to whether such a narrow economic concept is the best measure to decide upon the desired outcome. Associated with it are many uncertainties: What parameters should be included in the research? What weighing should be attributed to the parameters? How to deal with uncertainties? How to deal with issues of justice?
In the course of this work, an evaluation of this requirement can only be touched upon. I do not attempt to come to a final conclusion. Rather, I wish to discuss the *conditions for the possibility* for a beneficial deployment (see also Section 5.1 and 5.4).

iii. Option O only reckons with finite CE deployment (finitude thesis).

The finitude thesis, which only allows for temporally limited CE deployment, is especially important to the BT-argument. Of course, this makes the BT-principle rather strong. Incorporating the finitude requirement into the BT-principle can be justified through the following considerations:

Many authors of the buying time discourse adhere to the temporal deployment only assumption (Section 3.1). Words like 'postponing' and 'stopgap measure' are used frequently when relating to the buying time framing. The sheer meaning of the words implies some form of finitude – a stopgap is by definition only a bridge; the second best choice until a real solution is obtained. The meaning of the word 'postponing' also implies some form of temporal finitude. While it is in theory possible to infinitely postpone something, postponing, in practice, usually refers to some reasonable timeframe.

Secondly, I would argue that any CE deployment which is not finite in concept is no buying time deployment, but instead amounts to a *substitute* for mitigation efforts. Successful decarbonization of society implies, that mitigation efforts show an effect. Under this assumption, BT-deployment would be used as an auxiliary measure to reduce pressure and buy time. Conversely, if the process of decarbonization is assumed to fail, CE deployment would have to be used continuously as a substitute for insufficient emission cuts. So, the finitude requirement only then is superfluous, if CE is anticipated to be a substitute for emission cuts.

This leads to the third reason as to why the finitude requirement has been included in the BT-principle. The idea of CE being only a temporary stopgap measure is backed by the idea that the transition of society will happen anyway. Eventually, a low- or no-carbon society is thought to be inevitable, making artificial counterbalances to the carbon circle (like BECCS) or the radiative forcing (like SAI) redundant. In this picture, CE is not structurally needed for the transition *per se*. It is only needed to manage the manner of transition. Once the necessary change towards a low- or no-carbon economy has been achieved, CE is deemed to be superfluous. The transition anyway hypothesis that many authors silently adopt (see Section 3.1) is the main reason as to why CE can and should be finite. This line of thought will be deepened in Chapter 6 where the danger of CE undermining a transition of society is discussed (Section 6.2.2).

So, I argue that only finite deployment schemes count as buying time scenarios. However, the concept of finitude must be specified somewhat. While a deployment of 800 years (MacMartin 2014) is in theory finite, this would in practice surpass any reasonable policy making. Finitude as I wish to understand it not only refers to a pre-determined period of time, but also implicitly makes assumptions about controllability. If the time span in which any CE measure is deployed makes unrealistic assumptions about centuries to come, I will argue that this timespan is not finite in a practical sense.

Specifying the concept of finitude might also capture the ambiguity that arises when the terms 'short' and 'finite' are used synonymously: I understand finitude as a practically controllably timeframe that does not critically surpass the end of this century – a timeframe that correlates to the RCP's prognostic realm (van Vuuren 2011, IPCC 2013), and in this sense is relatively short.²²

iv. CE in O does not lead to less mitigation compared to mitigation in O without CE (no-impediment thesis).

The next requirement essentially demands that there is no or at least a controllable relation between mitigation and CE deployment. It requests that mitigation efforts remain constant, whether CE will be deployed or not. Underlying the no-impediment requirement is the demand for CE deployment not to be used as a substitute for mitigation efforts. This requirement can be backed by the following considerations:

In Section 2.2, I argued that there is a strong moral obligation to pursue mitigation efforts even in face of possible CE deployment. I have established this obligation as superior to CE deployment, because CE and mitigation essentially do not address

²² Of course, this limitation can be criticized. Even potential infinite deployment of global technology has proven to be sustained by man: Global telephone and internet technologies are examples of technologies that are both global and potentially infinite. However, those technologies are not necessarily risk technologies. For CE technologies, the deployment span should be finite, short, and controllable due to of the potential hazardous side effects of those technologies.

the same issues. Particularly if mitigation is understood in a broader sense as part of a sustainable transformation of society away from a fossil fuel-based economy, CE measures cannot substitute for that. In light of the mitigation obligation, CE and mitigation cannot serve as substitutes. Additionally, there are enhanced risks in CE deployment, and CE deployment might eventually lead to a dilemma for future generations.

The no-impediment thesis serves two purposes: On the one hand, it stresses the importance of mitigation efforts; on the other hand, it connects the discussion to a broader context of transitioning society. Especially in connection with the transition anyway assumption, which has been identified in section 3.1 and which also serves to back the finitude thesis, the no-impediment thesis gains force for the social and political debate. This will be further discussed in Chapter 6.

v. There is no option O' which is maximally finite and which leads to G and which is better than O (no-better-option thesis)

A full evaluation of this thesis goes far beyond the scope of this work, as potential harms and benefits would have to be compared comprehensively between different climate portfolios including CE. Defining a better-than relation may be equally challenging as finding a reasonable cost-benefit relation. Rather, for the purpose of this work, the no-better-option thesis has the function of a 'safeguard' that helps to identify an inadmissible narrowing of options by presenting CE-deployment as the only or best option. In particular, this means that, if ambitious mitigation were to be seen as a realistic and viable approach, which did not inappropriately burden social and economic structures, this option should be preferred to a portfolio including CE. If two portfolios equally realize a desired climate goal G, the portfolio without CE should be adopted. If there is no immediate benefit gained from CE, it should not be part of a climate policy mix, due to its inherent uncertainty being a risk technology.

While this may seem obvious, it might be hard to achieve. Framing CE as a viable option might diminish the willingness to engage in equally viable, but potentially more difficult options (this is a version of the moral hazard argument, see Section 2.3). The no-better-option thesis urges us to be very careful in ruling out other options, and not to project a CE-portfolio prematurely.

vi. There are no general moral constraints on option O (morality thesis).

Finally, the morality thesis asks for option O to be generally morally acceptable. There are at least two drawbacks to putting the morality thesis separately at the end of the argument: Defining a beneficial deployment (effectiveness thesis) rests on normative assumptions (e.g. about fairness) and thus could also be dealt with in the context of the morality thesis. Secondly, if there is indication that a CE deployment scheme might be in general morally unacceptable, this might rule out the scenario to start with. After all, if CE deployment is inherently morally wrong, why bother exploring any other requirements like the effectiveness- or the finitude thesis? Meanwhile, the work at hand has a different research question: Are there possible BT-deployments absent any general moral constraints? So, for now, it is assumed that CE is not inherently morally wrong. Of course, the general moral status of CE

that CE is not inherently morally wrong. Of course, the general moral status of CE *is* of importance, but will be discussed only after the compliance with the other central BT-requirements is assessed.

The order of the BT-requirements determines the order in which the requirements are evaluated in the remainder of this thesis: At first, the 'hard facts' about the deployment scenario are given and the stage for any policy option O is set: Is the goal desirable, and does the portfolio realize it in a (theoretically) beneficial manner? After this, the central BT-requirements are evaluated: Is option O finite? Does it impair mitigation efforts? And could there be an alternative? Finally, the general moral status of climate option O is assessed.

3.3 Instantiations and desirability of climate goal G

Only if placeholders G and O are instantiated do the premises become truth-apt and can the argument be evaluated for its validity. In this section, climate goal G is further subdivided into four climate goals and the desirability of each goal is established.

Two broad climate goals at which CE deployment is aimed have already been introduced in Section 3.1: 1) avoiding dangerous climate impacts, and 2) reducing pressure. Those two goals can now be further specified:

Ad 1. Avoiding dangerous climate impacts

In Section 2.2, the general obligation to avoid irreversible and dire climate impacts has been established in the context of the mitigation obligation. In particular, climate tipping points ought to be avoided. Reducing global warming to 2°C above pre-industrial level generates a 66% chance of realizing this goal and avoiding temperature-dependent tipping points (IPCC 2013).

Additionally, there are some tipping points that are affected by the rate of climate change, i.e. that are rate-dependent. A policy option that would aim at reducing the rate of temperature change would also address rate-dependent climate tipping points. So, the broad goal to avoid dangerous climate impacts can be divided into two sub-goals:

Climate Goal 1	Avoiding temperature-dependent tipping points
Climate Goal 2	Avoiding rate-dependent tipping points.

Ad 2. Reducing pressure

The notion of 'reducing pressure' is what I deem to lie at the heart of the buying time framing. According to the authors using this framing, the time gained via a buying time deployment would be put to use as to manage mitigation and adaptation in an orderly fashion: More clear-headedly, less cost-intensively – in short, with less pressure.²³

The goal of reducing pressure is actually a compound goal. It focuses on stabilizing the global mean temperature at a certain degree, while at the same time asking for reduction in its social and economic costs. In the following, costs for mitigation are separated from costs for adaptation. This generates two additional climate goals:

- Climate Goal 3 Reducing adaptation costs and stabilizing temperature at 2°C above pre-industrial level, compared to adaptation costs without CE.
- Climate Goal 4 Reducing mitigation costs and stabilizing temperature at 2°C above pre-industrial level, compared to mitigation costs without CE.

²³ Economic considerations may especially shape the understanding of the pressure terminology, e.g. when phasing out economic branches is compared to phasing in new ones.

For the sake of readability, I will use the short version 'reducing mitigation pressure' and 'reducing adaptation pressure' for the remainder of this work, while having in mind that those goals are actually compound²⁴.

The starting point of each argument is the stipulation that those goals are desirable. The first requirement of the BT-argument asks for any climate policy goal to be desirable (desirability thesis). Is this the case for each of the four goals?

Climate Goal 1: Avoiding temperature-dependent tipping points.

Mitigation efforts essentially aim at realizing climate goal 1. Hence, the desirability of climate goal 1 has been established in Section 2.2 in the context of the mitigation obligation. A version of the BT-argument, which assumes any option O in order to achieve climate goal 1, amounts to the so-called two-degree argument that has already been introduced into the debate (Betz and Cacean 2012). It assumes that the two-degree target can be realized only with the aid of CE (see also Section 5.2.2). Regardless of whether this argument turns out to be true, the anticipated climate goal is indeed desirable.

Climate Goal 2: Avoiding rate-dependent tipping points.

Given that tipping points might have irreversible and potentially negative consequences for both current and future generations, it is morally imperative to avoid them. The same holds consequently for rate-dependent tipping points. Hence, both 'avoiding tipping point' goals are *prima facie* morally desirable. Therefore, if at all possible, the current generation should aim at avoiding each sort of tipping points and the two respective climate goals can be said to be indeed desirable.

Climate Goal 3: Reducing adaptation pressure

The (moral) obligation to minimize or reduce costs, respectively, lies at the heart of the buying time argument. This obligation can be backed in two different ways:

²⁴ The goal of reducing pressure is not only compound, but also comparative: State of the world 1 should be transformed (reduced) into state of the world 2 by means of policy option O. What 'reduction' would translate to in terms of policy pathways and what amount of reduction would be acceptable *all things considered* cannot be decided in the scope of this work. In this sense, some fuzziness of the BT-argument remains. Still, I believe the reducing-pressure terminology to be more accessible in terms of research and further policy choices.

Either with reference to economic rationality or via considerations about appropriateness when dealing with moral obligations.

Economic rationality in a broad sense is a strategy that aims at reducing costs while maximizing utility, e.g. achieving a desired goal (e.g. Anand 1995). At first sight, this is by no means a problematic approach. In fact, this kind of thinking is involved in most of our day-to-day actions. If there are two ways in which a result can be obtained, it seems rational to choose the less cost-intensive way²⁵. Problems only arise when minimizing costs leads to negative or even immoral behavior: If, for example, a third person or the environment is negatively affected. Other people's rights (and moral obligation towards the environment and animals) constitute the boundary at which cost-minimization should end. For the deployment of CE this means, if the negative side effects of the deployment impose risks on other people or the environment, this strategy is not morally acceptable, even if it might minimize mitigation or adaptation costs.

Appropriateness on the other hand refers to the amount or degree of 'burden' or 'sacrifice' that can be expected from moral agents when carrying out their moral obligation. Minimizing adaptation costs (and mitigation cost, respectively) can be justified by appropriateness-consideration, if it is assumed that the envisaged adaptation pathway is 'too hard' on the moral agents.

Regarding adaptation, it might very well be the case that the pace of climate change generates severe problems for societies and citizens seeking to adapt. Some adaptation measures might take considerable time to be realized (IPCC 2014c). If it were possible to lessen the rate of climate change, adaptation measures could be installed in a timely manner and possibly at lower cost.

That being said, there is a discrepancy between local burdens and global impacts of climate change. While adaptation costs are borne by single countries, climate change and CE technologies act on a global level with possible global side effects. While for some countries it would be desirable to reduce adaptation costs via SAI, other countries might be harmed by the side effects of such deployment. Thus, climate goals must be agreed upon internationally and in accordance with a mutually beneficial weighting of side effects. If those challenges are met, reduction of adaptation costs could be a desirable climate goal.

²⁵ Cost is a broad concept that also may include time, material, and also opportunity costs.

Climate Goal 4: Reducing mitigation pressure

The same line of reasoning applies with respect to the reduction of mitigation cost. By both economic rationality as well as appropriateness considerations, reduction of mitigation costs can be justified (e.g. Hampicke 2000).

Mitigation costs can be defined as the costs that result from engaging in carbonneutral behavior at both an economic as well as individual level. The simple assumption here is that an extended time frame in which those measures ought to be realized leads to less pressure and hence less costs. Riahi et al. (2015) suggest that this assumption might really turn out to be overly simplistic. They show that, on the contrary, postponed mitigation efforts even increase overall mitigation costs. If a certain emission budget is to be satisfied, diverting from the optimal emission curve - in the sense of there being a temporal emission overshoot - asks for future emissions to decrease rapidly, which is even more costly. Only if BECCS or other negative emission technologies are anticipated in such an overshoot scenario can mitigation costs be reduced – but, as Riahi et al. urgently argue (Riahi et al. 2015), this 'bet' on future negative emissions is highly speculative and comes at the cost of possibly not complying with the two-degree target. Whether under these circumstances the reduction of mitigation costs is still desirable depends on a careful weighting of the different options and the consequences associated with realizing or not realizing certain political objectives.

With regard to appropriateness, Baatz and Ott (2016) have argued recently that certain mitigation strategies are inappropriate if they push citizens beyond a sufficiency threshold. This will probably not be the case in certain energy-shift scenarios for Germany in the next couple of decades. There are, for example, energy-shift scenarios for Germany that would come at reasonable costs and would resonate with the mitigation goals of the German Government. So, at least for Germany, reduction of mitigation costs might turn out to be superfluous in light of appropriateness considerations.

Another thing worth highlighting is that the obligation to mitigate is untouched by the desirability to minimize the costs of mitigation strategies. Minimizing mitigation costs should not be confused with minimizing mitigation efforts. Specifically, the reduction of mitigation costs ought not to be achieved by reducing mitigation altogether. This is why the mitigation obligation holds independently of the BT-argument. The no-impediment thesis in turn is influenced by the mitigation obligation, forbidding inadmissible substitution between mitigation and CE. The evaluation of the no-impediment thesis turns out to be quite complex, though, especially in case of reducing mitigation pressure via BECCS (see Section 4.5.3 and 4.5.4.)

The discussion in the previous paragraphs suggest that the desirability of the two 'reducing pressure' goals is highly dependent on the way they are brought about. They are susceptible to weighting with other goals or ethical obligations. In that sense, they might be called *secondary* climate goals (see 2.2). In particular, the reduction of mitigation costs can only be justified if it does not impede other, primary climate goals, such as the two-degree target.

Nonetheless, ultimately decisive for those weighting issues is the actual, non-ideal situation decision makers are confronted with. So, while in theory those secondary climate goals should be renounced as soon as they are found to interfere with far-reaching mitigation efforts, in reality, some sort of CE might turn out to be the only option for a 2° , or even a 1.5° world.²⁶

3.4 Results

In the above chapter, the central assumptions of the buying time framing have been sifted out so as to formulate the buying time principle underlying the BT-argument:

Buying Time Principle (BT-Principle)

If: i. Climate goal G is desirable, ii. option O leads to climate goal G and is beneficial in so doing so, iii. option O only reckons with finite CE deployment, iv. CE in O does not lead to less mitigation compared to mitigation in O without CE, v. there is no option O' which is maximally finite and which leads to G and which is better than O, vi. there are no general moral constraints on option O; then: option O should be adopted.

According to this principle, a buying time deployment of CE needs to be *desirable*, *effective*, *beneficial*, *finite*, *not in conflict with mitigation*, *the best available option* and *morally allowed*.

²⁶ This limitation, however, is no drawback to my claim. My aim in this thesis is the thorough analysis of one specific argument in favor of CE in light of certain, to date plausible, assumptions. This is not diminished by the fact that some assumptions might turn out to no longer hold true.

The outstanding feature, which sets this argument apart from other consequentialist arguments about effective and beneficial CE-deployment, is arguably the incorporation of the finitude thesis and the no-impediment thesis. The assumptions backing the finitude thesis are:

- (*Transition Anyway*) Successful decarbonization of society is anticipated. Eventually, a low- or no-carbon society is thought to be reality, making artificial counter-balances to the CO₂ circle superfluous.
- (*No Substitute*) If decarbonization is successful in the end, CE deployment will cease, since it will no longer be needed to counteract insufficient emission cuts. This means that CE of any kind would only be deployed additionally to a comprehensive mitigation pathway, not instead of one. A non-finite CE deployment would amount to a *substitute* for mitigation efforts which is not acceptable within the BT-framing.
- (*Controllability*) Finitude can only be justified as a concept, if it incorporates some form of practical controllability. Deployment scenarios that point to well beyond the end of this century, may be finite in theory, but are not controllable in a practical sense.

The assumptions that shape the formulation of the no-impediment thesis are:

- (*Mitigation Obligation*) The current generation has the moral obligation to drastically reduce their CO₂ emissions within the next few decades. This obligation holds even in face of possible CE deployment.
- (*No Substitute*) If decarbonization is successful in the end, CE deployment will cease, since it will no longer be needed to counteract insufficient emission cuts. This means that CE of any kind would only be deployed additionally to a comprehensive mitigation pathway, not instead of one. A non-finite CE deployment would amount to a *substitute* for mitigation efforts, which is not acceptable within the BT-framing.

The BT-principle and -argument respectively, include two placeholders: climate goal G and policy option O. Four climate goals have been identified:

Climate Goal G

Climate Goal 1	Avoiding temperature-dependent tipping points.
Climate Goal 2	Avoiding rate-dependent tipping points.
Climate Goal 3	Reducing adaptation pressure.
Climate Goal 4	Reducing mitigation pressure.

In Chapter 4, I will introduce four different deployment scenarios that serve as instantiations for policy option O, enabling in-depth evaluation of the BT-argument in Chapter 5.

Chapter 4 Deployment Scenarios

The buying time argument scheme is characterized by the use of two placeholders O and G. Only if the placeholders are replaced with a concrete option and goal the argument's plausibility can be evaluated. Instantiations for climate goal G have been discussed in the previous chapter. The guiding question of the following two chapters is, whether there are climate policy options O that make the argument plausible. In other words: Is the proposed buying time idea compatible with concrete deployment scenarios?

To answer this question, two deployment scenarios will be depicted in this chapter. There are deployment scenarios that resonate with the buying time idea, and may have, in fact, been created specifically for that purpose. Two such scenarios have come to my attention: 1. Reduced Warming Rate – buying time for adaptation, and 2. Flexible Emission Budget – reducing mitigation pressure.

The first scenario is a SAI deployment that has been proposed by Keith and MacMartin (2015) (Section 4.1). The authors introduce a SAI scenario set-up designed to reduce the pace of the global warming rate for different emission trajectories (RCPs). The extraordinary feature of this scenario-bundle is the finitude of SAI deployment, which is present in at least two sub-cases. The authors do not argue that this deployment scenario should be adopted or that it is the best scenario conceivable. Rather, they demonstrate how a well-defined scenario can be shaped on base of guiding principles, and present a case study for the sake of advancing the discussion.

The second scenario incorporates BECCS deployment in light of current mitigation trends (van Vuuren et al. 2013). The authors demonstrate, how BECCS might be used to artificially extend the remaining emission budget in order to realize the two-degree target. Again, the authors of this scenario do not argue for its actual deployment, but rather use it to illustrate what a certain climate policy assuming BECCS would amount to.

4.1 Reduced warming rate – buying time for adaptation

One scenario for a sulfate aerosol injection scheme (SAI, see also section 2.5) has been introduced by David Keith and MacMartin and their colleagues (MacMartin et al. 2014, Keith and MacMartin 2015). In their proposal, SAI is used in order to reduce the rate of global temperature change for a given emission trajectory, rather than stop the temperature increase all together. This scenario is an excellent example for the BT-argument, for it is designed specifically in accordance with the buying time approach:

"At a minimum, introducing [the proposed SAI scenario] would increase the *amount of time* to both learn and adapt; by reducing the needed rate for adaptation, it could *reduce costs of adaptation* [...]."

(MacMartin at al. 2014, p. 2, emphasis added [FN])

Additionally, the authors incorporate the idea to argue case-specific: "One cannot meaningfully evaluate solar geoengineering without a scenario for its implementation." (Keith and MacMartin 2015, p. 201). This resonates perfectly with the assumption of the work at hand, which asks for a case specific analysis of the empirical premises of the BT argument.

Lastly, the authors assume (at least for some RCPs) the finitude of the SAI deployment. This scenario is, taken all things into consideration, the best candidate for a plausible instantiation of the BT argument that has come to my attention so far. Consequently, the SAI scenario of Keith and his colleagues will function as one case study for the further evaluation of the BT argument.²⁷

Temperature increase in the next century, even under optimistic mitigation assumptions, will have severe influence on life on earth. It is not only the absolute change in global mean temperature, which catalyzes dangerous climate events, but also the *rate* of warming. The frequency and strength of some climate events are determined by the warming rate and a quick temperature rise will make adaptation to those climate threats even more difficult (MacMartin et al. 2014).

SAI could limit the rate of global temperature rise. MacMartin, Keith and their colleagues have proposed a modest, temporal deployment of SAI (the authors use the

²⁷ Keith and MacMartin 2015 is a revised version of MacMartin et al. 2014.

acronym SRM synonymously). This way, the authors argue that SAI could buy time for adaptation measures.

"In particular, using SRM to limit the rate of temperature change would provide more time for both ecosystems and human systems to adapt to climate changes." (MacMartin et al. 2014, p. 11)

The scenario is designed to meet specific criteria: It has to be temporal, moderate and responsive. These are the normative principles that guide the scenario setup.

The technology that the authors envisage is a sulfate aerosol solar radiation management (SAI). The physics behind this technology has been discussed in Section 2.5, and the authors specify the physical set-up only in certain details.

They assume the radiative forcing of one million ton of sulfur (MtS) to be 0.6 - 0.8 Wm⁻². For a given RCP, the scenario targets at cutting the rate of change of the respective radiative forcing in half. This suggests that the choice of a certain RCP is decisive for the scenario.

"The choice of emissions trajectory has a profound influence over the magnitude and duration of geoengineering required to limit the rate of change of global mean temperature."

(MacMartin et al. 2014, p. 5)

For the main part of their investigation and for illustrative purpose, the authors assume RCP4.5 and calculate the amount of injected sulfur accordingly. In the RCP4.5 pathway, injection would start with 0.035 MtS per year and would increase steadily in the first decade by the same amount. After 10 years, the yearly injection would amount to 0.35MtS. The rate of injection would be adjusted in order to steady the rate of temperature change (picture below, second panel).



Figure 7. The top panel shows the total radiative forcing for RCP4.5, and a radiative forcing profile in which the rate of change is halved. The second panel indicates the suggested SAI (SRM) profile. Effect on global mean temperature is shown in the third panel, indicating the time benefit, and the corresponding injection rates per decade are shown in the last panel (source: Keith and MacMartin 2015).

As the figure above shows, the point in time, at which temperature reaches a certain degree above pre-industrial level, is delayed for several years, if SAI is used to limit the rate of temperature change.

Using a different approach, the authors determine the amount of SAI in accordance with a steady warming rate of 0.1°C per decade (instead of cutting warming rate in half, as in Keith and MacMartin (2015)). For RCP4.5, the length of required SAI deployment would have to be 160 years, if SAI is used to limit temperature change to 0.1°C per decade.



Figure 8. The top panel (a) shows SAI (SRM) being used for either maintaining constant global mean temperature ('0°C per decade) starting in 2020 or limiting the rate of change to 0.1°C per decade for an RCP4.5 pathway. Panel (b) shows the corresponding solar reduction potential of SRM as well as the length of time over which SAI (SRM) would have to be deployed (source: MacMartin et al. 2014, online version in color).

For other RCPs, the below picture indicates the required timeframe and intensity of SAI schemes for varying temperature goals. If SAI is used to constrain temperature rise to 0.0 °C, that is to maintain current temperature, for all RCPs but RCP2.6, the finitude of the CE scheme is not given.



Figure 9. The panels show the duration (a) and amount (b) of SAI that would be required to constrain the rate of temperature change for different RCP pathways. For RCP8.5, the amount and duration of SAI is significantly larger than for other pathways. If SAI (SRM) is used to maintain a constant temperature (zero rate of change), the maximum solar reduction is finite for each RCP, but duration is only finite for RCP2.6 (indicated by 'x' for the other RCPs). (MacMartin et al. 2014, online version in color.)

Whether this scenario is optimal or even possible, is still not decided (as the authors state themselves). Instead, it presents itself as one first step towards a comprehensive analysis of purpose-bound and technology specific deployment scenarios. The authors use this case not to argue for its implementation, but rather to make discussion of the matter more transparent.

It is important to repeat that SAI in this scenario would not be used to tackle absolute temperature change: "(...) it [temporary SRM deployment] would not address problems such as temperature-dependent tipping point thresholds (...)" (MacMartin et al. 2014, p. 2). SAI would only account for reducing the *rate of change* for a given temperature rise, not its magnitude. In this setup, the main driver to limit global temperature rise would remain to be mitigation. Accordingly, it is still and only mitigation which could possibly lead to the stabilization of global mean temperature at 2°C above pre-industrial level (given the assumptions of the respective RCP).

This scenario does not frame SAI as a substitute for mitigation efforts, since both SAI and mitigation are concerned with different aspects of global climate change. While the former addresses the rate of change, the latter influences its quantity. The absolute temperature change is still determined by mitigation efforts, and SAI would be used supplementary to reduce the rate at which this change is to come about. SAI in this scenario does not reduce the dire need for drastic emission cuts – it rather points out what additional help SAI could generate. This is a great advantage of the scenario, since it does not initially run afoul of being a "cheap techno-fix" which supposedly makes mitigation obsolete. Moreover, the no-impediment thesis becomes at least plausible in this scenario, as shall be discussed in Chapter 5.

4.2 Flexible Emission Budget – minimizing pressure for mitigation

Another way in which CE could buy time relates to the emission budget. The budget approach to the two-degree target is a common method when evaluating necessary emission strategies (e.g. IPCC 2013, Kriegler et al. 2015). This approach rests on the observation that net cumulative emissions of anthropogenic GHG are near linear to the change of global mean temperature. Even if the specific emission trajectory alters – CO_2 emission reaches its peak early and decreases slowly or it peaks late and falls steeply – if the net cumulative emissions are roughly the same at the end of the century, absolute temperature change will be similar for those trajectories. The budget approach usually embraces the period until the end of the century, at which time a net-zero economy is assumed. CO_2 concentration of about 450ppm is assumed to realize the two-degree target with a probability of more than 66% (Rogelj 2016).



Figure 10. Four CO_2 emission pathways with identical cumulative carbon emissions over the twenty-first century (a) and their corresponding temperature projections (b). The grey area indicates the uncertainty range (source: Rogelj et al. 2016).

Rogelj et al. (2016) have analyzed the varying methods, by which different carbon budgets were obtained. They conclude that if differentiated climate feedbacks are taken into account, the remaining carbon budget would be no more than 590-1,240GtCO₂ between 2015 and 2100. The question is, of course, how the remaining emissions are to be allocated and whether an overshoot can be balanced out via future negative emissions.

Van Vuuren et al. (2013) show how drastically emissions would have to be reduced in order to realize near-zero emissions in 2100. For example, assuming a 2% reduction rate per year, emissions in 2020 must be similar to those in 2000 and must lie 10% under the anticipated baseline scenario (Fig. 1, Panel a) which assumes a business-as-usual economy. If a limited mitigation potential is assumed, reduction must happen even more quickly (Fig. 1, Panel b).



Figure 11. Illustrative emission pathways based on carbon budget of 1600 GtCO₂ and constant emission reduction rate. Panel (a) shows emission pathways going to zero emission with the respective reduction rate, panel (b) shows results if the emission reduction potential is limited, panel (c) shows the results if negative emissions via BECCS are assumed. Panel (d) summarizes the emission window for 2020, if emissions in 2100 are zero with a maximum reduction rate of 2-3% per year (source: Van Vuuren et al. 2013).

The authors now demonstrate how the pressure of immediate mitigation could be lessened when assuming future negative emissions obtained with bioenergy with carbon capture and storage (BECCS).



Figure 12. Emission trajectory of scenarios with and without BECCS leading to stabilization at 450 and 480ppm CO₂-eq (source: Van Vuuren et al. 2013).

It is obvious from the above figure, how BECCS can postpone the point of installing drastic emission reduction, thereby allowing an emission overshoot. The red graph, which indicates stabilization at 450 ppm by 2100 under the assumption of RCP2.6, demands a sharp decrease in carbon emission by 2020 latest. The blue graph, however which assumes BECCS and negative emissions in the future, delays this point by about 10 years. Also, the curve decreases much steeper than the red one, hence implicating a more powerful CO_2 -reduction-potential of BECCS as compared to mitigation alone. This indeed could lessen the pressure of immediate mitigation and hence might be a means to buy time.

On the other hand, if a more ambitious climate target is aimed at, BECCS might still be able to allow a temporary overshoot, but would demand emission reduction to be as immediate as the RCP2.6 suggests (black graph).

It is worthwhile in this context to evaluate the intended national determined contribution (INDC), as have Riahi et al. (2015) done²⁸. The authors distinguish two mitigation scenarios that have short-term targets for 2030: HST and LST. While LST (low short-term target) aims at emitting only 53 Gt of CO₂-eq. by 2030, HST (high short-term target) settles at a mitigation effort of reaching 61 Gt of CO₂-eq. In the HST pathway, already 70% of the two-degree emission budget is spent by 2030, creating an "emission gap" as compared to an optimal two-degree pathway.

This of course puts a lot of pressure on the remaining decades to reduce emissions substantially. By 2050, the identified emission gap would have to be closed in order to reach stabilization at 450ppm or 550ppm resp. This "leaves only 20 years (2030 to 2050) for a fundamental transformation of the global energy-economic system." (Riahi et al. 2015, p. 13).

Depending on the assumptions about energy demand, energy transformation, technical feasibility and social and political behavior, this 20-year transformation is seen as either impossible or feasible. Notably the availability of negative emission technologies (NETs) influences the judgment.

The authors now reach the important conclusion that

²⁸ More recently, the Paris pledges have been analyzed. At the time of writing, the latest INDCs also would not be compatible with a stabilization of temperature at below 2°, and relying on negative emissions seems insufficient (Kartha and Dooley 2016).

"... in particular the HST pathway, narrows policy choices and increases the risks that some of the currently optional technologies, such as the large-scale deployment of biomass or CCS, will become "a must" by 2030 in order to achieve low stabilization targets."

(Riahi et al. 2015, p. 20)

In other words: The way short-term emission pathways are designed today highly influences the options available to the future.²⁹ This highlights the interdependency of CE technologies and current emission strategies. If, for example, negative emission technologies like BECCS are anticipated, an emission overshoot is still consistent with the two-degree target. On the other hand, when high short-term emissions are agreed upon, BECCS seems to become inevitable. This self-enhancing process is morally relevant, and will be discussed in Section 5.4.

4.3 Results and complete argument matrix

This chapter has introduced two different scenarios that serve as the blue print against which the BT argument is evaluated. The first one is a well-defined SAI deployment scenario, which has been sketched out for different RCPs. The second scenario evaluates, how BECCS can be used to close the emission gap of current emission pledges. Both scenarios serve the same purpose for the upcoming chapters: Replacing placeholder O in the BT argument, thus enabling the evaluation of the premises.

The SAI scenarios show how time span and intensity of the designed deployments would differ for each RCP. I will adopt a case-discrimination accordingly, with one case adopting a mitigation pathway related to RCP2.6, in which remaining within the two-degree emission budget is assumed to be more likely, and one case assuming an emission overshoot that exceeds the two-degree budget. The same case-discrimination will be applied to the BECCS cases.

However, even in an RCP2.6 scenario, there might be an emission overshoot, which could possibly be leveled out by future negative emissions (Section 4.2). That being

²⁹ From a policy perspective initial low pledges might be prudent: Because reduction targets under the Paris Agreement will increase over time, states will only pledge low INDC in the first period even if their energy policies are already more advanced.

said, the two-degree budget serves as the quantitative threshold defining emission efforts, while the RCPs indicate the likelihood of remaining within this budget. This generates four climate policy options O:

Climate Policy Option O

Climate Policy Option a	SAI + CO_2 emissions remaining within the two-
	degree budget ($CO_2 < 2^\circ$ budget).
Climate Policy Option A	$SAI\ +\ CO_2$ emissions exceeding the two-degree
	budget (CO ₂ > 2° budget).
Climate Policy Option b	$BECCS \!$
	degree budget ($CO_2 < 2^{\circ}$ budget).
Climate Policy Option B	$BECCS$ + CO_2 emissions exceeding the two-degree
	budget (CO ₂ > 2° budget).

The lower case letters a and b indicate emissions remaining within the two-degree budget; the upper case letters A and B signify an overshoot scenario.

An argument matrix, which plots O and G against each other, yields sixteen version of the BT-argument:

Climate Goal G		Preventing climate tipping		Reducing pressure		
		points				
			Avoiding	Avoiding	Reducing	Reducing
			temperature	rate-	adaptation	mitigation
			- dependent	dependent	pressure	pressure.
			tipping	tipping		
			points	points.		
/			1	2	3	4
	Deliev Ontion O					
<u> </u>	$CO_{2} < 2^{\circ}$, 	91	3)	23	34
	budget	a	aı	a2	as	uT
-	$CO_2 > 2^\circ$	Α	A1	A2	A3	A4
SA	budget					
	$CO_2 < 2^\circ$	b	b1	b2	b3	b4
	budget					
S	$CO_2 > 2^\circ$	B	B1	B2	B3	B4
EC	budget					
Bl						

Table 5. Argument matrix. The rows depict climate goals G, the columns depict portfolio options O. Each entry represents a version of the BT-argument

As introduced in Chapter 2, a climate policy portfolio consists of at least three options: mitigation, adaptation and CE. For now, I will assume that there is no relation between these options³⁰. Instead, I will incorporate two CE technologies, SAI and BECCS, and differentiate two sub-cases regarding mitigation efforts, as the above table illustrates. Those sub-cases are not determined by the compliance with the mitigation obligation. Rather, they are purely quantitative: Are emission cuts anticipated that would remain within the two-degree budget of CO₂ emissions?

By including the amount of mitigation as a background assumption of climate policy option O, the moral status of mitigation efforts is not decisive when evaluating the premises of the BT-argument. There is a clear strategic benefit in doing so: The premises of the buying time argument can be evaluated in light of the 'empirical' assumption about the climate portfolio, without having to decide about normative premises including a moral principle that asks for CE to be accompanied by fast and far reaching mitigation. This enables a rather analytical evaluation of the BTargument.

This analysis, however, generates far-reaching results. Most notably, the sub-cases assuming an overshoot render at least one premises of the BT-argument implausible or at least undecided. I believe to be able to demonstrate that the BT argument only then has *plausible* premises, if emissions are assumed to remain within the two-degree budget. This is indeed a great result, and a clear benefit as opposed to continuously emphasizing the moral obligation to engage in drastic mitigation efforts in face of CE – which already is obvious and still does not lead to more ambitious mitigation.

The mitigation obligation is of course the moral epicenter of this work. It enters the BT-argument in two ways: As the no-impediment thesis, and later, as a general evaluation pattern in Chapter 6. Especially does the mitigation obligation hold independently of any assumed CE deployment. I deem this result to be a benefit of the argument reconstruction at hand.

³⁰ This assumption will be further scrutinized in Chapter 6. 85

Chapter 5 Evaluation of the Arguments

In Chapter 4, I have introduced two technology deployment scenarios, which I believe to resonate best with the buying time idea, thereby introducing four instantiations of placeholder O. In Chapter 3, different instantiations for climate goal G have been introduced, which combined result in a sixteen-field matrix for the BT-argument. The aim of this chapter is now to evaluate for each combination of G and O, whether the respective argument generates plausible³¹ premises. Not all versions must or can be discussed in detail, however. If only one premise turns out to be implausible, the whole argument can be rejected (Section 5.2). Surprisingly, only two arguments can be said to be plausible (Section 5.1). Moderate SAI plus sufficient mitigation could help avoid rate-dependent tipping points as well as help reduce adaptation pressure (argument a2 and a3).

The remaining versions are neither obviously plausible nor obviously implausible. In most cases, I cannot decide upon the plausibility of a given scenario due to a lack of knowledge. Especially when scientific facts are at stake, which surpass the information of the two exemplary deployment scenarios introduced in Chapter 4, evaluation of the arguments must remain incomplete. I will label those arguments accordingly as being beyond the scope of this work, while at the same time urging for further research (Section 5.3).

On the other hand, there are deployment-scenarios that seem in itself rational, but are at odds with the BT argument in its present form. Most notably, the prominently discussed case of BECCS as a means to buy time in face of less than sufficient mitigation turns out to obscure the dubitable status of at least two buying time requirements – the finitude of the deployment and the impact on mitigation efforts. In this situation, there are two possibilities: Holding onto the strong version of the BT-argument and rejecting the scenario, or reformulation the BT-requirements so that they incorporate the desired scenario. In this chapter, the first path is traced (Section 5.4), while a reformulation of the BT-argument is sought in Chapter 6.

The discussion here serves two purposes: 1. on the methodological level, it shows how the buying time argument can be evaluated and 2. on the practical level, it might

³¹ Plausibility, as well as likelihood is understood in a broad, every-day meaning. Plausibility in a argumentation-theoretical discourse is an elaborate concept (e.g. Walton 2001), referring to which is not necessary for the current purpose.

give some insight on how to assess different deployment scenarios. This kind of argument-evaluation is a first step towards a comprehensive case study, which is necessary for assessing the moral status of any CE deployment. While I am aware that this can only be a first tentative judgment, this work serves as an example of how argument analysis can indeed be politically relevant.

5.1 Plausible instantiations

There are two combinations of G and O that yield arguments with plausible premises. The green areas signify those arguments.

Climate Goal G		Preventing climate tipping		Reducing pressure		
		points				
			Avoiding	Avoiding	Reducing	Reducing
			temperature	rate-	adaptation	mitigation
			- dependent	dependent	pressure	pressure.
		tipping	tipping			
			points	points.		
	Policy Option O		1	2	3	4
	$CO_2 < 2^\circ$	a	al	a2	a3	a4
	budget					
Г	$CO_2 > 2^\circ$	Α	A1	A2	A3	A4
SA	budget					
	$CO_2 < 2^\circ$	b	b1	b2	b3	b4
	budget					
	$CO > 2^{\circ}$	D	D1	D2	D2	D.4
ğ	$CO_2 > 2^\circ$	в	ы	B2	ВЭ	B4
EC	buaget					
E E E		1	1			

Table 6. Plausible Instantiations of the BT-argument.

The two arguments a2 and a3 will be discussed in turn, starting with the strongest and most interesting case of SAI-deployment in order to reduce pressure for adaptation (argument a3, Section 5.1.2). The second argument a2 is very similar to a3, and will be discussed briefly. Both argument a2 and a3 assume mitigation efforts to remain within the two-degree budget, while the use of SAI addresses the rate of temperature change.

In order to evaluate the plausibility of the arguments, each premise needs to be scrutinized and its plausibility needs to be shown. This will be done for those two arguments a2 and a3, with the exception of the morality thesis. As said before, the research question of the thesis at hand is, whether a BT-deployment is possible absent any general moral constraints about CE. This means that for the time being, it is assumed that neither SAI nor BECCS are inherently morally wrong. Whether this

is true, will be subject of Chapter 6, when general moral constraints on plausible BTinstantiations are discussed.

5.1.2 Argument a3

G	Climate Goal 3	Reducing adaptation pressure.
0	Climate Policy Option a	SAI + CO_2 emissions remaining within the two-
		degree budget ($CO_2 < 2^\circ$ budget)

The first deployment scenario introduced in this chapter has clearly been designed to meet the climate goal identified as the Reducing Adaptation Costs Goal (Climate Goal 3). The climate portfolio in this scenario aims at stabilizing temperature at 2 °C, while on the other hand, lessening the rate of change in order to buy more time for adaptation.

The envisage CE technique is a special version of SAI. The calculations are made – among others – for an RCP2.6 pathway, hence, for this version, mitigation in option O will be assumed to enable emissions to meet the two-degree target.

It is essential to this scenario that SAI is not used to stabilize temperature, but only to lower the *rate* of temperature change. The stabilization of temperature in accordance with the specific RCP is achieved solely through cuts in CO₂ emissions.

Of the six BT-requirements, only the following theses need to be evaluated in the upcoming section: effectiveness thesis, finitude thesis, no-impediment thesis, and nobetter-option thesis. The desirability thesis has already been established for each climate goal G (Section 3.3), and the morality thesis will be focus of Chapter 6. As said before, the question of the work at hand is, whether a BT-deployment is possible absent any general moral constraints about CE. This means that, for the time being, SAI and BECCS are framed as not being inherently morally wrong.

(Effectiveness Thesis) Option a leads to desirable climate goal G3 and is beneficial in so doing so.

According to the scenario set-up, SAI will be successful in reducing the warming rate (MacMartin et al. 2014). The authors argue that SAI will thus be able to reduce adaptation pressure, by generation more time for learning and installing different adaptation measures. Furthermore, if the RCP2.6 mitigation pathway is assumed, the stabilization of temperature at 2°C above pre-industrial level is likely to be guaranteed. Therefore, indeed, SAI deployment can be called effective.

The second part of the effectiveness premise asks for a beneficial deployment. The weighting of the benefits and the negative side effects in order to guarantess some form of beneficial deployment, is a matter of great dispute. It would be beyond the scope of this work to fully decide for the deployment scenarios at hand, if the benefits outweigh the costs. So, rather, some central and most discussed side effects of SAI deployment are isolated and presented here.

Whether the scenario at hand eventually is beneficial or not, depends on model results I cannot provide. The question here is, if MacMartin's and Keith's deployment scenario has the theoretical *conditions for the possibility* of being beneficial. I will focus on three possible negative side effects associated with SAI: precipitation, methods of injection and affiliation to the fossil fuel sector and procedural justice. The guiding question is: Are there circumstances that would render this scenario beneficial?

Precipitation. Keith and MacMartin have specified some climate damages resulting from SAI, including the deviation of precipitation from pre-industrial level (Keith and MacMartin 2015). The authors suggest that the level of SAI, which maximizes climate benefits (namely the temperature stabilization), is much higher than the level that maximizes the sum of harms and benefits. In consequence, the level of beneficial SAI would always be less than the level of most effective SAI. Given that, is it possible that the moderate SAI approach could count as beneficial? What percentage of change in precipitation could still be 'acceptable'?

Acceptable climate impacts generated by SAI deployment³² could, for example, be defined in accordance with basic human rights, such as access to food and clear water, and not infringing the sufficiency threshold of the affected (Shue 2010).

If deviation from natural precipitation patterns violates human rights or makes selfsufficient life in certain areas of the world difficult if not impossible, it surely would not be beneficial. Questions in this regard are, among others: What groups of people are affected by SAI impacts? In what climate and precipitation range can their lifestyles exist? Are there forms of compensation conceivable?

This very short discussion means to illustrate the theoretical possibility to shape a beneficial SAI deployment, if those questions can be answered satisfactorily.

³² This supposes, that there are acceptable impacts of SAI, and that not simply all effects of SAI would count as inacceptable. This correlates with the notion of SAI not being inherently morally wrong.

Methods of Injection. The authors propose the use of aircraft to inject the sulfate (Keith and MacMartin 2015). Obviously, airplanes must be fueled. Hence, the use of SAI would, at least to some extend, require the use of fossil fuels and thus add to continued CO_2 emission. This could only be acceptable, if the other benefits (such as prevention of climate tipping points) outweigh this challenge.

However, it could be argued that the use of fossil fuel for aircraft is a contingent factor, and sustainable aviation fuel is underway. If this was a real possibility, not only would SAI injection cease to add to carbon emission, but would also lose its connection to the fossil fuel sector.

Procedural justice. While it is long and by argued that the use of SAI would be simple and cost-effective, even allowing for single states to undergo the deployment, it is still a technique that requires central control. Since it is a large-scale manipulation of a common good – the atmosphere – its deployment has to be subdued to common decision making. A global moratorium on SAI use might be hard to achieve, but at least it could be argued that common global agreements are possible and are in fact, presently installed in many different areas. Hence, global agreement on SAI deployment can be possible and is necessary for beneficial deployment.

So, in conclusion there could be situations in which a BT-deployment of SAI could be more beneficial than harmful. However, this might be only the case for low-scale SAI deployment, or far less or far shorter deployment than might be desirable from a purely temperature-based assessment. As the authors of this scenario have stated: "Whatever the weighting of these benefits and cost functions, the ratio of benefits to costs will be largest for very small amounts of SRM." (Keith and MacMartin 2015, p. 204). They continue to observe: "(One) must conclude that the optimal amount of SRM will always be less than the amount of SRM that maximizes benefits." (Keith and MacMartin 2015, p. 204). This very important conclusion should be present for any possible framing of SAI deployment. It implies that beneficial SAI deployment might be possible, but only within clear-cut boundaries. Those boundaries might arguably be given in a RCP2.6 world, where emission are cut to remain within the two-degree budget and SAI would have to be deployed for a short period of time only.

(Finitude Thesis) Option a only allows for finite SAI deployment.

In their 2014 paper, MacMartin at al. determine the deployment time of SAI for different RCPs. In an RCP2.6 world, for example, SAI deployment would only be necessary for 40 years. However, the time frame increases sharply: For the RCP4.5, the duration of deployment would be 160 years, culmination to roughly 800 years for RCP8.5.



Figure 13. Comparison of the solar reductions required to maintain a 0.1°C per decade rate of temperature change for different emission pathways (source: MacMartin et al. 2014). See also Figure 9, Section 4.1.

In Section 3.2 I have argued that finitude is a practical concept, which makes assumptions about controllability. Any time frame that clearly surpasses the end of the century, may in theory be finite, but cannot be handled practically. Additionally, it presents a huge risk transfer into the future.

The authors also calculate the termination time for the RCPs, depending on the allowed rate of temperature change (see Section 4.1, Figure 9). Only if 0°C change is allowed (i.e. if SAI is used to stabilize temperature at pre-industrial level indefinitely), the use of SAI in all RCPs but RCP2.6 is infinite. Under all other circumstances, both the intensity and the duration of the deployment is projected to be finite. But again, it must be stressed that some expected durations of SAI surpass any realistic timeframe on which policy decisions could rely on.

Assuming CO_2 emissions to remain within the two-degree budget, and assuming that this might most likely be the case in RCP2.6, a duration of about 40 years is indeed a manageable, finite timeframe.

(No-impediment Thesis) SAI in option a does not lead to less mitigation compared to mitigation in option a without SAI.

As illustrated before, MacMartin and his colleagues contrast the SAI deployment with different RCPs. The RCPs and with them, mitigation efforts, are the background assumptions that enter their model set-up. In this sense, they assume mitigation efforts staying constant for each RCP. Thus, the scenario design aligns with the requirement that SAI deployment should not compromise mitigation efforts. Since the envisaged SAI deployment would only address the *rate* of temperature change and not the temperature change itself, it would not act as a substitute for mitigation. If, moreover, a mitigation pathway like the RCP2.6 is assumed, the two-degree target might be reached with mitigation efforts alone, while SAI would address the secondary climate goal of reducing adaptation pressure.

However, there are some drawbacks to the evaluation of the no-impediment thesis. It might be difficult to attribute the increase in temperature and the change in the warming rate to SAI and mitigation separately. Since the effectiveness of a SAI scheme and of mitigation partly overlap, there might be no clear role-attribution to either of them. This problem is even more manifest in the BECCS-cases.

Additionally, the likelihood of the scenario can be challenged. Even if the design of the scenario makes the no-impediment thesis true, there is no guarantee that this will be the case if implemented in "real life" climate policy. This "reality gap" will be focus of discussion in Chapter 6. Here, it needs to be stressed, that at least the assumptions of this deployment scenario presuppose constant, sufficient and eventually successful mitigation efforts.

(No-better-option Thesis) There is no option O' which is finite and which leads to climate goal G3 and which is better than option a.

As introduced above, the no-better-option thesis in my BT-argument reconstruction works as a 'safeguard' to make sure that no option not including SAI deployment, is ruled out prematurely. If two portfolios equally realize a desired climate goal G, the portfolio without CE should be adopted. So, in context of this version of the BT- argument (Argument a3), it needs to be asked if there is a conceivable portfolio without SAI that is equally powerful in reducing the rate of temperature change and thus in reducing pressure for adaptation.

The temperature change can be influenced by ambitious mitigation strategies. But SAI is specifically designed to influence the rate of change *on basis* of those mitigation strategies. It is an additional climate strategy that serves a secondary purpose. Hence, a portfolio without SAI might be able to determine the temperature change, but would have no possibility to influence the rate of temperature change in this given setup³³.

So, regarding the specific climate goal to reduce adaptation pressure and assuming that even drastic mitigation primarily influence the absolute temperature change, SAI presents itself indeed as the best option for this purpose.

5.1.2 Argument a2

G	Climate Goal 2	Avoiding rate-dependent tipping points.
0	Climate Policy Option a	SAI + CO ₂ emissions remaining within the two-
		degree budget ($CO_2 < 2^\circ$ budget)

In this version of the buying time argument, the same deployment scenario is taken into consideration as above: It is climate option *a*, (MacMartin et al. 2014, Keith and MacMartin 2015) that introduces time limited and purpose bound SAI in order to reduce the warming rate. They calculate their deployment scheme in light of different RCPs, and as in the previous chapter, I will assume a scenario which ensures the two-degree target via mitigation efforts. For argument a2, the climate goal is to avoid rate-dependent tipping point, such as the shutdown of the thermohaline circulation (Stocker and Schmittner 1997). Since the in-depth discussion of this specific policy mix has been given above, I will not discuss each premise individually, but rather discuss them together.

At first glance, all BT-requirements can be said to be met – the scenario was designed exactly for this purpose. SAI is effective, in that it would reduce the warming rate and hence prevent rate-dependent tipping points, it might also be beneficial in so doing so. Mitigation efforts would ensure the realization of the two-

³³ Of course, if temperature would rise less sharply, the rate of change would also be influenced. But for a given projected temperature rise, only SAI could arguably influence the rate of change. There might be other CE strategies to also serve the purpose of influencing the changing rate. However, the question here is, whether there is a portfolio without any CE measure that realizes the same goal.

degree target. Also is SAI finite and the no-impediment thesis is true for the same reason as it was true in the previous argument: Since mitigation efforts are assumed to stay constantly high in the respective RCP, they will not be affected by SAI. Lastly, the no-better-option thesis must be said to be plausible, too, since no climate strategy without SAI might address the rate of global temperature change.

To conclude, this version of the BT-argument which aims at avoiding rate-dependent tipping points via SAI and which assumes a high level of mitigation in order to realize the two-degree target, is plausible. However, the discussion indicates quite clearly that the underlying emission pathway is the decisive element for the evaluation. Even when assuming SAI, emissions must be reduced massively, in order to reach the two-degree goal.

5.2 Implausible instantiations

The above section (Section 5.1) has introduced two instantiations of the BTargument with respect to one specific SAI deployment scenario. The premises resulting from this instantiations are *prima facie* plausible. In this section, I will discuss six versions of the BT-argument that I deem to be implausible, represented by the red areas (see table below).

Climate Goal G		Preventing climate tipping		Reducing pressure		
		points				
			Avoiding	Avoiding	Reducing	Reducing
			temperature	rate-	adaptation	mitigation
			- dependent	dependent	pressure	pressure.
		tipping	tipping			
		points	points.			
Policy Option O		1	2	3	4	
	$CO_2 < 2^\circ$	a	al	a2	a3	a4
-	budget					
	$CO_2 > 2^\circ$	Α	A1	A2	A3	A4
SA	budget					
	$CO_2 < 2^\circ$	b	b1	b2	b3	b4
S	budget					
	$CO_2 > 2^\circ$	B	B1	B2	B3	B4
CC	budget					
BE	o anger					

Table 7. Plausible and implausible instantiations of the BT-argument.

In case of identifying implausible arguments, it suffices to point out only one implausible premise. I will discuss the implausible argument instantiations in groups of what climate goal they are supposed to reach (arguments a1 and b1) and what policy option they assume (arguments A1 – A4), respectively. One tentative conclusion of the table above is that an SAI-portfolio assuming CO_2 emissions exceeding the two-degree budget results in an implausible instantiation of the BTA (arguments A1 – A4). If this is true, this result might influence the decision making-process in case a buying time deployment via SAI is advocated in light of possible insufficient mitigation.

5.2.1 Arguments a1 and b1

Policy option *a* has already been discussed as a plausible instantiation of two BT-versions (arguments a2 and a3, see Section 5.1). If the goal differs, however, the SAI deployment scenario leads to implausible premises of the respective argument.

If climate goal 1 is stipulated, the BT-argument essentially amounts to the twodegree target: It aims at preventing temperature-dependent tipping points. This is to be achieved via stabilization of global temperature rise to maximally 2°C above preindustrial level (see Section 1.1, Section 2.1, Section 2.2, Section 3.3 and Section 4.1).

Climate portfolio *a* might be able to realize the two-degree target, since it assumes emissions to remain within the two-degree budget. This way, mitigation in option O guarantees the realization of climate goal 1. As said before, the SAI deployment in this set-up does not address the amount of temperature change, but the rate at which it happens. In argument a1, SAI deployment is superfluous for reaching the respective climate goal

If the two-degree target would be reached with mitigation alone (which is the background assumption of this argument), then there is indeed a better option than mitigation plus SAI, and that is: mitigation alone. If mitigation alone would be effective in realizing the two-degree target, then SAI would be redundant. Hence a portfolio without SAI is a better option; the no-better option thesis would become implausible.

The same reasoning applies to argument b1 and policy mix b, respectively, where BECCS is assumed additionally to already sufficient mitigation. That way, BECCS would also be superflous, and the better option as compared to BECCS combined with mitigation would be: mitigation alone. In both arguments a2 and b1, the twodegree target can be reached with mitigation alone and CE deployment is not needed. The no-better-option thesis becomes implausible in both cases.

5.2.2 Arguments A1, A2, A3 and A4

As indicated in Section 3.3, each of the four climate goals includes the stabilization of temperature rise at 2°C above pre-industrial level. A mitigation trajectory which leads to a global temperature rise of more than 2°C does not realize the two-degree target. Hence, climate option A which assumes CO_2 emissions to exceed the two-degree budget and which assumes SAI deployment not to address absolute temperature change, is not effective in reaching any of the relevant climate goals.

Having said this, SAI deployment could still be used to achieve the two-degree target under insufficient mitigation efforts, if it is used to address temperature change directly. A lack of mitigation efforts could possibly be balanced by more SAI, if SAI is used as a substitute. The respective argument to support such SAI deployment is the so-called two-degree target argument that has been already been introduced by Betz and Cacean (2012) (see also Section 3.3). Such a deployment scheme, however, would deeply differ from the scenario introduced above in Section 4.1. In a temperature-stabilizing scenario, SAI would act as a substitute to mitigation. The no-impediment thesis becomes at least questionable in such a set-up.

Additionally, if mitigation efforts will not be able to stabilize temperature mediumterm, SAI deployment might have to be continued infinitely, or at least for a very long timespan (Section 4.1, also MacMartin et al. 2014). This violates one essential buying time requirement: the finitude thesis. In summary, a scenario that assumes SAI deployment in light of insufficient mitigation makes at least two premises of the corresponding argument implausible: the finitude thesis and the no-impediment thesis.

The below table shows the implausible versions of the BT-argument and indicates the premises, that become implausible or at least questionable in the specific set-up.
Climate Goal G			Preventing clima	te tipping	Reducing pressure	
			points			
			Avoiding	Avoiding	Reducing	Reducing
			temperature	rate-	adaptation	mitigation
			- dependent	dependent	pressure	pressure.
			tipping	tipping		
			points	points.		
/			1	2	3	4
Policy Option O						
	$\begin{array}{c} \text{CO}_2 < 2^{\circ} \\ \text{budget} \end{array} \mathbf{a}$		No-better-			
			option			
			thesis	21) T	NT.
	$CO_2 > 2^\circ$	A	NO-	NO-	NO-	NO-
	budget		thesis	thesis	thesis	thesis
			finitude	finitude	finitude	finitude
SAI			thesis	thesis	thesis	thesis
	$CO_2 < 2^{\circ}$ b		b1			
	budget					
S	$CO_2 > 2^\circ$ B					
EC	budget					
B						

Table 8. Implausible instantiations of the BT-argument. The implausible premises of the respective argument are shown in the fields.

5.3 Lack of knowledge

In most cases, I simply do not have the expertise to judge the plausibility of a given scenario, especially when scientific facts are at stake that surpass the information of the two exemplary deployment scenarios introduced in Chapter 4.

Climate Goal G			Preventing clima	te tipping	Reducing pressure	
		points				
			Avoiding	Avoiding	Reducing	Reducing
			temperature	rate-	adaptation	mitigation
			- dependent	dependent	pressure	pressure.
		tipping	tipping			
			points	points.		
	Policy Option O		1	2	3	4
	$CO_2 < 2^\circ$	a	a1	a2	a3	a4
	budget					
П	$CO_2 > 2^\circ$	Α	A1	A2	A3	A4
SA	budget					
	$CO_2 < 2^\circ$	b	b1	b2	b3	b4
	budget					
S	$CO_2 > 2^\circ$	В	B1	B2	B3	B4
CC	budget	-			20	
BE						

Table 9. Plausible, implausible and undecided arguments. Light blue indicates a lack of knowledge about the scenario, dark blue indicates two cases that are unclear with regard to the underlying BT-principle.

There are six arguments that I cannot fully evaluate in this work due to a lack of knowledge. Four of them are arguments assuming a BECCS-scenario (b2, B2, b3 and B3). Those arguments mean to realize the related climate goals 2 and 3: Reducing adaptation pressure and avoiding rate-dependent tipping points. Those arguments share the same uncertainty: It is unclear whether BECCS – a technology which primarily deals with the concentration of CO_2 in the atmosphere and only secondarily with the pace this concentration comes about – can address any rate-dependent climate goals. It is beyond the scope of this thesis to judge the many ongoing model-runs for BECCS.

Two other arguments containing uncertain premises are a4 and b4. They each assume a mitigation pathway, which realizes the two-degree target independently of additional CE deployment. The respective climate goal is a reduction in mitigation pressure (climate goal 4). While argument B4, which assumes BECCS in face of insufficient mitigation as a means to reduce further mitigation pressure, will be the focus of discussion in the next section (Section 5.4), the notion of reducing mitigation pressure for already successful mitigation might make no sense. Hence, arguments a4 and b4 are included in the group of indeterminable arguments.

The first two arguments I will deal with are arguments b2 and B2:

Argument b2:

G	Climate Goal 2	Avoiding rate-dependent tipping points.
0	Climate Policy Option b	BECCS + CO_2 emissions remaining within the two-
		degree budget ($CO_2 < 2^\circ$ budget).

Argument B3

G	Climate Goal 2	Avoiding rate-dependent tipping points.
0	Climate Policy Option B	BECCS + CO_2 emissions exceeding the two-degree
		budget ($CO_2 > 2^\circ$ budget)

Whether BECCS is able to tackle the rate of temperature change is unclear. It might have some effect on the warming rate by manipulation the amount of CO_2 in the atmosphere. This is not a direct adjustment of the warming rate as exemplified by the SAI scenario. It cannot be ruled out, however that BECCS might be able to realize climate goal 2. If, in addition, an emission overshoot is anticipated (argument B2), BECCS would arguably be used to both stabilize the absolute temperature change and influence the pace of warming. The interdependency of those two aspects and the role BECS can play here, is opaque to me. The effectiveness thesis cannot be evaluated for both arguments in the given context.

Argument b3:

G	Climate Goal 3	Reducing Adaptation pressure
0	Climate Policy Option b	BECCS + CO_2 emissions remaining within the two-
		degree budget ($CO_2 < 2^\circ$ budget).

Argument B3

G	Climate Goal 3	Reducing Adaptation pressure
0	Climate Policy Option B	BECCS + CO_2 emissions exceeding the two-degree
		budget ($CO_2 > 2^\circ$ budget)

For similar reasons, arguments b3 and B3 cannot be further evaluated. Since climate goal 3 is related to climate goal 2 - a reduction in the rate of climate change may allow for a reduction of adaptation pressure – the same uncertainties arise. The scenarios introduced in Chapter 4 do not give any insights on whether BECCS might be able to buy time for adaptation.³⁴

It is, again, not impossible to conceive a BECCS scenario, in which the point in time, at which adaptation measures would have to be installed, is postponed by the use of BECCS. Especially in addition to ambitious mitigation efforts (argument b3), deployment of BECCS could enhance an already effective mitigation pathway to give adaptation measures more time to be implemented. Justifying those arguments, however, is not possible in the scope of this work.

The next two arguments both relate to climate goal 4: Reduction of mitigation costs.

Argument a4:

G	Climate Goal 4	Reducing Mitigation pressure
0	Climate Policy Option a	SAI + CO_2 emissions remaining within the two-
		degree budget ($CO_2 < 2^\circ$ budget)

³⁴ One may argue that time for adaptation *follows* from more time for mitigation (BECCS as a means to buy time for mitigation will be discussed in Section 5.2). At the same time that mitigation pressure is lessened, time for adaptation measures is also generated. However, this claim also cannot be backed in the current investigation and is left for further research.

Argument b4

G	Climate Goal 4	Reducing Mitigation pressure
0	Climate Policy Option b	BECCS + CO_2 emissions remaining within the two-
		degree budget ($CO_2 < 2^\circ$ budget).

In case of argument a4, it needs to be pointed out that SAI deployment as introduced by MacMartin and Keith (MacMartin et al. 2014, Keith and MacMartin 2015) and his colleagues is calculated against the backdrop of a certain RCP trajectory. SAI in those scenarios works additionally to mitigation efforts and, at least theoretically, does not affect them (see Section 5.1). Since SAI is used only to reduce the rate of change, it does not relate primarily to mitigation costs. Whether there is a connection between mitigation costs and the lessening of the rate of temperature change, must be validated by further research. A preliminary assessment indicates that the effectiveness thesis is at least dubitable in case of argument a4.

Of course, this result is obtained within the limitation to the SAI-scenario introduced above. Other SAI scenarios are imaginable that very well could aim at reducing mitigation costs. A very common perception of SAI deployment is as a limited substitute for mitigation in order to stabilize temperature, when mitigation alone would not be successful in doing so. In such a set up, however, the no-impediment thesis would be in jeopardy.

Whether SAI as part of a policy mix, as introduced in section 4.1, will have any effect on mitigation costs, is at least debatable. Further research is needed for each of the unclear arguments, if they are to be established as plausible instances of the buying time argument.

5.4 A special case: Argument B4

There are deployment-scenarios that seem in itself rational, but are at odds with the BT-argument in its present form. Especially the prominently discussed case of BECCS as a means to buy time in face of less than sufficient mitigation (Section 4.2), turns out to obscure the dubitable status of at least two buying time requirements: The finitude of the deployment and the impact on mitigation efforts. In this situation, there are two possibilities: Relying on the strong version of the BT-argument and rejecting the BECCS scenario, or reformulation the BT-requirements

in a way that they incorporate the scenario. In this section, I will trace the first option and discuss climate policy option B as an implausible instance of the BT-argument.

Argument B4 might be the most important one when discussing the buying time framing. Argument B1 is closely related to B4 and will be discussed simultaneously. Both arguments assume that mitigation efforts will not be able to stabilize temperature at 2°C above pre-industrial level. The underlying idea of those argument versions then is that in light of insufficient mitigation, BECCS could close the emission gap and thus help reaching the two-degree target, even if emissions temporarily exceed the two-degree budget. It is this the scenario that has been introduced in Section 4.2. Additionally, if future negative emissions are anticipated, the timeframe for drastic emission cuts would broaden, thus buying time for further mitigation approaches.

Argument B1:

G	Climate Goal 1	Avoiding temperature-dependent tipping points	
0	Climate Policy Option B	BECCS + CO_2 emissions exceeding the two-degree	
		budget ($CO_2 > 2^\circ$ budget)	

Argument B4:

G	Climate Goal 4	Reducing Mitigation pressure
0	Climate Policy Option B	BECCS + CO_2 emissions exceeding the two-degree
		budget ($CO_2 > 2^\circ$ budget)

Since both arguments assume insufficient mitigation, they will be discussed together. The BT-theses will be specified for argument B4, the validity of the according thesis in B1 follows from them³⁵.

As the discussion will reveal, a BECCS deployment to artificially enhance the emission budget violates at least two BT-requirements, the finitude- and the noimpediment requirement, and it may very well also violate the morality requirement. That being said, this scenario may turn out to be actually necessary when tackling climate change. We may find ourselves in a situation, in which the scenario we propose violates our own normative standards. This quite sobering thought will be further strengthened in Chapter 6. On the other hand, this fact might suggest that the

³⁵ Since climate goal 4 includes climate goal 1, if a premise including goal 4 is valid, the same holds *mutatis mutandis* for a premise including goal 1.

BT-argument in its current form is too strong. A possible weak version of the BTargument will also be discussed in Chapter 6.

I will discuss all BT-theses for argument B4 in detail. The main focus lies on the effectiveness thesis (including the beneficial requirement); the finitude- and the nobetter-option thesis will be discussed together. Especially the no-impediment thesis will pave the way for discussion in Chapter $6.^{36}$

(Effectiveness Thesis) Option B leads to desirable climate goal 4 and is beneficial in so doing so.

The idea of the emission budget is that temperature can be stabilized at a certain degree regardless of the concrete emission trajectory. The two-degree budget has been advised to be around at 950 GtCO₂ highest, if CO₂ concentration is to be stabilized at 480ppm (IPCC 2014). This means, an emission pathway assuming a short-term overshoot can still be consistent with the two-degree target, if it is balanced out with net negative emissions by the end of the century.

While a substantial transformation in society would have to take place within the next decades or even years (see Section 4.2), with the assumption of future BECCS, the need for immediate and drastic emission cuts might be delayed. So, BECCS, as envisaged by the scenario introduced in 4.2, might indeed be effective in reducing mitigation pressure, because it might postpone the point in time, at which emissions have to go down, while at the same time minimizing the necessary amount of reduction.

However, this assessment of BECCS is influenced by possibly utopic assumptions about the effectiveness and readiness of the technology. As has been suggested recently by Kartha and Dooley (2016), it might be risky to rely on future negative emissions with BECCS, because this option might turn out to involve unacceptable risks, or prove to be either not feasible or ineffective. The effectiveness of BECCS is severely limited by land availability and the readiness of capturing schemes. Betting on negative emission may turn out to be a dangerous self-fulfilling prophecy (Fuss et al. 2014).

The effectiveness thesis asks for BECCS to be beneficial as well. The point to be made here, again, cannot be to definitely decide whether the buying time deployment

³⁶ The following discussion is partially adopted form Muraca and Neuber (2017).

of BECCS is beneficial or not. This would require more in-depth research than can be provided here. Rather, the question is, whether the proposed BECCS deployment may provide the *conditions for the possibility* of beneficial deployment.

As with the SAI scenario, I will focus here on three exemplary side effects of BECCS: Land-use conflicts, transport and storage safety and procedural justice. The guiding question is: Are there circumstances that could render a BECCS scenario beneficial?

Land-use conflicts. The potential of BECCS is restrained by many factors, such as land availability, freshwater for irrigation, food demand, and future climatic and social developments. Near-term, this potential is massively limited by the availability of arable land. The conversion of agricultural land into biomass plantations is determined by food demands, dietary choices and might include heavy fertilization (Schäfer et al. 2015; McNutt et al. 2015b).

If BECCS were to be used in a way consistent with RCP2.6, i.e. in order to reach the 2-degree-target, the estimated land-use of food-production and biomass-production combined would amount to 2.1 billion hectares in 2100. To put this number into perspective, today's cropland amount to 1.6 billion hectare (McNutt et al. 2015).

The conversion of land into biomass plantations ought to suffice certain ethical criteria. One set of ethical principles is proposed by the Nuffield principles on Biofuel (Weale et al. 2011):

i. Biofuel development should not be at the expense of people's essential rights (including access to sufficient food and water, health rights, work rights and land entitlements).

ii. Biofuels should be environmentally sustainable.

iii. Biofuels should contribute to a net reduction of total greenhouse gas emissions and not exacerbate global climate change.

iv. Biofuels should develop in accordance with trade principles that are fair and recognize the rights of people to just reward (including labor rights and intellectual property rights).

v. Costs and benefits of biofuels should be distributed in an equitable way.

The first two principles refer to land-use problems. This is especially important if biomass were to be produced in the global South, where arable land is scarce and the available cropland is essential for self-sufficient, small-scale agriculture. Moreover, if animal rights and species conservation are considered, the conversion of wildlife habitat would not be an option either.

Land-use conflicts may be prevented partially, if food-production where to use less land. This might either happen via increased productivity or via a change in food choices. Increasing the productivity of food-production might be possible in some regions of the world, but mostly via drastic measures (such as the employment of chemical fertilizers).

Highly specialized and fertilized monocultures of biomass might arguably not be compatible with the Nuffield principles on Biofuel. Moreover, the agricultural infrastructure needs to be augmented and needs constant care. The stream of maintenance might outweigh the benefit of BECCS, regarding both the increased cost and the energy needs.

Transport and storage safety. The gravest side effects of BECCS arguably occur during transport and storage (McNutt et al. 2015b). There are two kinds of harms associated with BECCS: Infliction of local population near CCS sites due to exposure to CO₂ during the injection, or long-term CO₂ pollution by storage leakage (Medvecky, Lacey and Ashworth 2013). Leak pipelines might cause highly concentrated carbon dioxide pollution. On the other hand, if storage sites are not safe, for example due to seismic activity, leakage may contribute to severe CO₂ pollution of land and sea habitats. The safety of those storing sites depends on further technological observation, like monitoring and early warning systems that would enable to evacuate the affected in time. But since both seismic and volcanic activity is still not sufficiently predictable, storage of CO₂ can never be fully safe. This is true for both geological caverns as well as deep-sea storage. While the leakage of the former might affect land-animals and humans, the latter might affect sea-animals and even the ecosystem ocean as a whole. Having said that, recent studies suggest that the probability of leakage is fairly low (Wennersten, Sun and Li 2015). If so, future decision makers could still decide upon carbon storage.

Procedural justice: Finally, BECCS power plants require a huge amount of upfront investments. Controlling BECCS should suffice criteria of procedural justice, which maximizes the right of co-determination of locals. Decisions about CO_2 storing sites may prove to face similar difficulties as the search for nuclear repositories. At least, decisions about the site of growing, processing and storing should aim at being consensual, taking the special threat to locals into account.

In summary, BECCS is prone to issues of both procedural and distributional justice. Benefits and side effects of BECCS should be distributed fairly, and the governance of BECCS needs to be democratic and transparent. This does not suggest that beneficial BECCS deployment is impossible. Rather, the discussion above might indicate further research questions for beneficial BECCS deployment.

Recent examples of scholars exploring the possibility of beneficial (an morally sound) CCS deployment are Boucher and Gough (2012) and Medveky et al. (2014). Boucher and Gough (2012) introduce guidelines, by which the deployment of CCS technologies, and *mutatis mutandis* BECCS, can be assessed. Those guidelines rest, among others, on central ethical principles about procedural and distribution justice as well as human rights. Medveky et al. (2014) highlight central moral issues with BECCS, and categorize them, especially with regard to transport and storage safety. Research in this direction could help to further strengthen the case for beneficial BECCS deployment.

(Finitude Thesis) Option B only allows for finite BECCS deployment;

(No-better-option Thesis) There is no option O' which is finite and which leads to climate goal 4 and which is better than option B.

In light of insufficient mitigation, there might be no other way than to deploy BECCS (or other negative emission technologies) in order to reach the two-degree target. This complies with the results of a recent meta-study by Fuss et al. (2014). Fuss et al. shows, how many models consistent with the two-degree target assume future use of BECC. The authors think this to be an alarming result, since both the effectiveness or even the readiness of the technology is not determined yet, hence relying on BECCS as a potential way to reduce immediate mitigation pressure might be risky (also Kartha and Dooley 2016). The no-better-option thesis might turn out to be true, but only because BECCS may present itself as a self-fulfilling prophecy.

With regard to finitude, there is no indication in the scenario introduced above (Section 4.2) (van Vuuren 2013 and 2016) that BECCS will end before 2100. In accordance with my understanding of finitude, which also implies practical controllability, BECCS in policy option B is not finite. This alone should be reason to reject the possibility of BECCS as being a BT-deployment, since it violates at least one premise of the argument: the finitude thesis.

(No-Impediment Thesis) BECCS in option B does not lead to less mitigation compared to mitigation in option B without BECCS.

The evaluation of this thesis is more difficult than the evaluation of the corresponding thesis in an SAI scenario. Because the design of the SAI scenarios was especially framed as to not interfere with mitigation efforts, the BECCS scenario at hand enables a lessening in mitigation efforts, or rather: It alleviates the negative side effects of insufficient mitigation efforts. It is only intuitive to assume BECCS and mitigation to be at least partially substitutes.

The use of BECCS is designed to *delay* drastic mitigation strategies, hence it can be said to it indeed compromise mitigation efforts. To put it differently: The BECCS scenario at hand is designed to work under compromised mitigation efforts. On the other hand, once these studies are published, there might be a feedback loop that even diminishes existing pledges and seduce policy makers to rely even more heavily on future BECCS. Already most RCP2.6 projections assume future negative emission technologies, mostly BECCS (Fuss et al. 2014). So, by the same token, by which the no-better-option thesis becomes plausible, the no-impediment thesis becomes implausible.

The difficulties in assessing the no-impediment thesis are connected to the ability of BECCS to serve as a *substitute* for mitigation. As stated initially (Section 2.2), small-scale BECCS might be count as a perfect substitute for mitigation. What this could mean for the evaluation of the arguments at hand (B1 and B4) shall be further discussed in Chapter 6.

5.5 Results

In this chapter, four categories of instantiation of the BT-argument have been identified.

Plausible Versions (green)

I have identified two plausible versions of the BT-argument, arguments a2 and a3. In those cases, SAI is deemed to be finite, short, beneficial and effective, while not interfering with mitigation efforts. This form of moderate and strictly purpose-bound SAI may realize the BT-requirements.

This result is somewhat surprising, because SAI especially has been criticized frequently on moral grounds. Nevertheless, a beneficial buying time deployment of

SAI might be conceivable, within the given boundaries. It is to be noted that the plausibility of this version prominently depends on the effectiveness of mitigation efforts. Only if emissions remain within the two-degree budget, a plausible SAI deployment is possible. Thus, in addition to the moral obligation to mitigate, it can be shown that CE deployment in a BT-framing is only *plausible* with sufficient mitigation.

Implausible Versions (red)

Five additional SAI-instantiations and one BECCS-instantiation have been identified as being implausible. When relating to Goal 1, both SAI and BECCS turn out to be superfluous in the respective policy option a and b (argument a1 and b1). All other SAI-scenarios, which assume emissions to exceed the two-degree budget, become implausible (arguments A1, A2, A3 and A4) – they violate both the finitude- as well as the no impediment-requirement. If that is true, this result might influence the decision making-process, as soon as a 'buying time deployment' of SAI in light of insufficient mitigation is advocated.

Undecided Versions (light blue)

For several scenarios, I cannot provide further scientific back up. Those BT-versions must remain undecided in the work at hand. Out of six undecided scenarios, five belong to BECCS scenarios (arguments b2, B2, b3, B3 and b4). This might indicate that there is much more uncertainty regarding BECCS than there is regarding SAI³⁷, and highlights the need for comprehensive research on BECCS, before it is mutually assumed as a potential future negative emission technology. Model-runs could be performed under further constraints, such as food security, property rights over land, and nature conservation targets (biodiversity). Without such morally relevant constraints BECCS modeling might remain abstract.

³⁷ Of course, it might be an argumentative bypass to follow from my lack of knowledge that there is greater uncertainty with BECCS than with SAI. In this sense, all research is only possible in the personal boundaries of knowledge and might be biased. However, a comprehensive scenario analysis for BECCS, like the SAI-scenario by MacMartin et al. (2014) and Keith and MacMartin (2015), has of yet not come to my attention.

The special case of BECCS (dark blue)

A BECCS deployment to artificially enhance the emission budget turns out to violate at least two BT-requirements: the finitude- and the no-impediment requirement, and it may very well also violate the morality requirement. Exactly this case, though, is the *raison d'être* for the contemplation and research of BECCS deployment scenarios. The above discussion has adhered to the strong BT-requirements, leading to a rejection of BECCS in light of insufficient mitigation efforts. Chapter 6 will trace another route by asking whether the BT-requirements might be too strong for the special case of BECCS.

To sum up, there are possible deployment scenarios that resonate with the buying time framing. Especially the two SAI cases (argument a2 and a3) might count as reasonable instantiations of the BT-argument. However, those scenarios only work if (and only if) parallel mitigation efforts take place that themselves will be able to reach the two-degree target. As a result, beneficial CE deployment will have to be quite limited, purpose bound and finite. Climate option *a* seems to fulfill those criteria. In this portfolio, SAI deployment is an additional strategy to take some pressure off or to delay the point in time by which certain goals have to be achieved. And – going back to the quotes from the ongoing discourse (section 3.1) – this is all there is to the buying time idea.

BECCS deployment, on the other hand, poses quite a riddle for the BT-argument, even if it seems to be a perfect instantiation at first sight. Especially the noimpediment thesis might turn out to be too strict for a BECCS.

			Preventing climate tipping points		Reducing pressure	
			Avoiding temperature dependent	Avoiding rate-dependent	Reducing adaptation pressure	Reducing mitigation pressure.
			tipping points	tipping points.		
			1	2	3	4
	$CO_2 < 2^{\circ}$ budget	a	No-better-option thesis. This climate goal amounts to the two- degree target. If CO_2 emissions remain within the two-degree budget, SAI deployment is superfluous in reaching the two-degree target.	SAI might be able to prevent rate depended tipping points, while mitigation efforts serve to realize the two-degree target.	SAI might be able to reduce the pressure of adaptation, if it lessens the rate of temperature change, which could generate more time for ecosystems and human systems to adapt.	Since mitigation is assumed to be ambitious in this case, it is unclear, in how far a lessening of the rate of temperature change might influence mitigation costs.
SAI	$CO_2 > 2^\circ$ budget	Α	No-impediment thesis. If carbon emissions exceed the two-degree mitigation, impeding mitigation efforts. <i>Finitude thesis</i> . Because SAI would function as a substitut assumption of exceeding carbon emissions,	e budget, SAI would be used in order to stabilize temperature. It would then be used as a substitute for e for mitigation efforts, it would have to be used continuously in order to stabilize temperature. Under the SAI would not be finite.		
	$CO_2 < 2^\circ$ budget (b	No-better-option thesis. This climate goal amounts to the two- degree target. If CO_2 emissions remain within the two-degree budget, BECCS deployment is superfluous in reaching the two-degree target.	Whether BECCCS can influence the rate of temperature change, cannot be decided within the scope of this work.	If additional BECCS can influence the rate of change, it might also reduce adaptation pressure.	While emission would remain within the two-degree budget, BECCS could be used to further reduce the time pressure for mitigation. Research for BECCS together with sufficient mitigation has not been reviewed for this research thesis.
BECCS	$CO_2 > 2^\circ$ budget	В	<i>The special case of BECCS.</i> BECCS in light of insufficient mitigation might be used to stabilize temperature change. It might however violate several BT-requirements. A weak version of the BT-argument could incorporate this case.	Especially in light of insufficient mitigation, it cannot be decided here, whether BECCCS can influence the rate of temperature change.	Especially in light of insufficient mitigation, it is unclear, whether BECCS can reduce adaptation pressure.	The special case of BECCS. BECCS in light of insufficient mitigation might be used to reduce mitigation pressure as it enhances the emission budget. It might however violate several BT-requirements. A weak version of the BT-argument could incorporate this case.

 Table 10 (same as Table 2). Full table of instantiation of the BT-argument.

Chapter 6 Deepened Analysis of the Buying Time Argument

According to the analysis in Chapter 5, there are two possible deployment scenarios that resonate with the buying time framing: A moderate, time limited SAI deployment in order to either reduce adaptation pressure (argument a3) or to reduce rate-induced climate impacts (argument a2). However, those scenarios are plausible only if parallel mitigation efforts take place that themselves will be able to reach the two-degree target. As a result, beneficial SAI deployment will have to be quite limited, purpose-bound and finite, as the authors of the scenario acknowledge (MacMartin et al. 2014). In this scenario, SAI deployment is viewed as an additional strategy to take some pressure off or to delay the point in time, by which certain climate goals, like adaptation measures, have to be realized. And – going back to the quotes from the ongoing discourse (Section 3.2) – this is all there is to the buying time idea.

BECCS deployment, on the other hand, poses quite a challenge for the BT-argument. The hope that BECCS may be used to reach the two-degree target even in face of poor mitigation efforts (arguments B1 and B4), is essentially the *raison d'être* for the contemplation and research of BECCS deployment scenarios. However, at least two BT-requirements are not plausible in those instantiations: the no-impediment thesis and the finitude thesis. As argued in Section 3.2, it is exactly those requirements that constitute the core of the BT-argument. This imposes a dilemmatic choice: Either the strong BT-requirements are maintained, leading to a rejection of the desired BECCS scenario(s), or the BT-requirements are watered down, establishing a 'weak' buying time argument, with the danger of allowing undesired scenarios to count as BT-scenarios as well.

The first horn of the dilemma – the rejection of the BECCS scenario(s) on grounds of the strong BT-requirements – has been traced in the previous section (Section 5.4). The next section will try to reformulate the BT-argument so as to capture the BECCS scenario (Section 6.1). Especially the possibility of acceptable substitution between BECCS and mitigation will be further examined.

Finally and lastly, the research thesis at hand scrutinizes general moral constraints on CE deployment. This will be discussed by means of the two plausible SAI-

instantiations (argument a2 and a3). The question is: If every of the strong BT-requirement is plausibly met, could CE deployment still be morally wrong? I will focus on two general moral objections to CE deployment: the hubris- and the techno-fix argument (Section 6.2) and show how a moderate, finite SAI deployment might not be susceptible to those general moral concerns.

6.1 Weak buying time argument

The hope that BECCS may be used to realize the two-degree target even in face of poor mitigation efforts is eventually the *raison d'être* for the contemplation and research of BECCS deployment scenarios. As discussion in Section 5.4 has shown, however, if the strong BT-requirements are applied, such scenarios must be rejected. It might be argued, then, that in such cases the BT-argument is too strong and that the BT-argument should be reformulated in order to capture the BECCS scenarios. This would yield two respective versions: a strong BT-argument and a weak BT-argument. In what follows, a weak version is established. Especially two BT-theses become implausible in arguments B1 and B4: the finitude thesis and the no-impediment thesis. In order to establish a different version of those theses it is worthwhile to reconsider the assumptions, which shaped their formulation in the first place (Section 3.2). They can be called supporting theses, as they support the BT-argument directly.

The assumptions backing the finitude thesis are:

- (*Transition Anyway*) Successful decarbonization of society is anticipated. Eventually, a low- or no-carbon society is thought to be reality, making artificial counter-balances to the CO₂ circle superfluous.
- (*No Substitute*) If decarbonization is successful in the end, CE deployment will cease, since it will no longer be needed to counteract insufficient emission cuts. This means that CE of any kind would only be deployed additionally to a comprehensive mitigation pathway, not instead of one. A non-finite CE deployment would amount to a *substitute* for mitigation efforts, which is not acceptable within the BT-framing.

• (*Controllability*) Finitude implies some form of practical controllability. Deployment scenarios that point to well beyond the end of this century, may be finite in theory, but are not controllable in a practical sense.

The assumptions that led to the formulation of the no-impediment thesis are:

- (*Mitigation Obligation*) The current generation has the moral obligation to drastically reduce their CO₂ emissions within the next few decades. This obligation holds even in face of possible CE deployment.
- (*No Substitute*) If decarbonization is successful in the end, CE deployment will cease, since it will no longer be needed to counteract insufficient emission cuts. This means that CE of any kind would only be deployed additionally to a comprehensive mitigation pathway, not instead of one. A non-finite CE deployment would amount to a *substitute* for mitigation efforts, which is not acceptable within the BT-framing.

Rejecting the strong version of the BT-argument and instead adopting a weak version would imply that at least one of the supporting theses is implausible. It seems to me that especially the no-substitute thesis may not hold in case of BECCS: BECCS might quite well function as an acceptable substitute for mitigation, as will be discussed below.

A weak BT-principle could read like this:

Buying Time*-Principle (BT*-Principle)

If: i. Climate goal G is desirable, ii. option O leads to climate goal G and is beneficial in so doing so, iii. CE in O is a perfect substitute for mitigation in O, iv. there are no general moral constraints on option O; then: option O should be adopted.

Most notably, the no-impediment thesis was replaced by the requirement to function as a perfect substitute (requirement iii):

iii. CE in O is a perfect substitute for mitigation in O (substitute thesis).

Also, the finitude thesis and the no-better-option thesis have been removed. Both theses seem to be at odds with the notion of CE serving as a perfect substitute for mitigation. The no-better-option thesis implies that any option reaching goal G without CE should be preferred to an option including CE (Section 3.2). But if CE in portfolio O is used as a perfect substitute for some amount of mitigation, option O is as desirable as the possible option O' without CE. Similarly, the finitude thesis can be removed, if BECCS is a perfect substitute that can be carried out indefinitely. So, both theses have been declined for the sake of the substitute thesis.

For the specification of what 'perfect substitute' amounts to, I will recall the definition given in Section 2.2, by Baatz and Ott (2016):

"Strategy Y is a perfect substitute for mitigation if it avoids all negative climatic effects resulting from GHG emissions, without thereby creating other harms or risks." (Baatz and Ott 2016, p. 99f.)

Does the BECCS overshoot-scenario (climate option B) resonate with the weak BTargument? Is the substitute thesis plausible?

The BECCS scenario introduced in Section 4.2 (van Vuuren et al. 2013) was designed to close the emission gap current mitigation efforts seem to generate. It assumes the reduction potential of BECCS to be -6 Gt CO_2/yr and -10 Gt CO_2/yr , respectively (van Vuuren et al. 2013, p. 18). This is well below the maximum reduction potential, which other authors assume to be as high as 18 Gt CO_2/yr (McNutt et al. 2015b). As mentioned before, the potential of BECCS is limited by the availability of arable land and by infrastructure including fresh water access and fertilization. In a recent study, Smith and Torn (2013) calculated a BECCS scenario with a specific type of plant (switchgrass) with focus on water and fertilizer-requirement. In order to remove 3.7 Gt CO_2/yr from the atmosphere (which is much less than the necessary 6Gt CO_2/yr in the emission-gap scenario!), the magnitude of all three factors would be exorbitant: 200 million hectare of land which is 20 times the area currently used in the US for bioethanol production, 20% of the global production of fertilizer (20 Tg/yr of nitrogen) and 4% of global renewable water resources (4,000 km3/yr) (Smith and Torn 2013, McNutt et al. 2015b).

Even a fraction of the assumed mitigation potential in van Vuuren's scenario might turn out to be strenuous on the environment as well as competitive regarding landavailability and infrastructure. The side effects of such a scenario make it questionable, whether BECCS is effective and beneficial and could possibly count as a perfect substitute for mitigation.

This does not rule out other, less strenuous BECCS scenarios. Those might be considerably downsized, and might have a significantly lower mitigation potential. So, in theory, beneficial BECCS is conceivable, that could also serve as a perfect substitute for mitigation. Yet, the mitigation potential of such deployment scheme might not be enough to close the emission gap, hence, it might not prove to be effective in a very basic sense. The BECCS-deployment in question, which is assumed to close the emission gap (option B, arguments B1 and B4), is due to its magnitude no perfect substitute. Even a weak version of the buying time argument does not render arguments B1 and B4 plausible.

Additionally, I wish to argue that the rejection of the finitude thesis poses a serious problem for any anticipated deployment of CE as a means to buy time. As said before, the transition anyway thesis lies at the heart of the buying time argument: Only if the prospect of a decarbonized society is taken seriously, CE-deployment can be framed as being an auxiliary measure, a temporary stopgap to buy time. Climate goal 4 (reducing mitigation pressure) itself can only be formulated, if mitigation as part of decarbonizing the economy is seen as the primary political goal, of which the reduction of pressure is the secondary goal. If the finitude thesis is not taken into consideration, this supposed weak BT-framing might jeopardize a fundamental transition of society (see also Muraca and Neuber 2017).

The weak version of the BT-argument allows for (some amount of) substitution between CE and mitigation. A rampant substitution, though, might negatively influence the pursuit of decarbonization. A trade-off between mitigation efforts and CE measures is at least a possibility (Baatz 2016, see also chapter 2.3). As soon as CE measures are framed as viable substitute for emission cuts, mitigation efforts might decline (Corner and Pidgeon 2010, Betz and Cacean 2012, Hale 2012, Baatz 2016). If furthermore, CE measures are not required to be temporally limited, a fundamental transition of society might not be undertaken – indefinite and acceptable substitution via CE could enable current modes of production and consumption to persist. How true is the buying time framing under those assumptions?

"Would [CE technologies] be used as a stop-gap measure to buy time for a societal

transformation to a carbon-free economy, shaving off the worst effects of climate change along the way? Or as a substitute for this transformation, allowing for business as usual to continue?"

(Schäfer et al. 2014, p. 243)

How these questions are answered, seems to be a question of faith. Bunzl (2009) for example argues very strongly against a moral hazard, i.e. the danger of diminishing mitigation efforts in light of CE.

"Moral hazard only arises for geoengineering if you think that research or, if it came to it, implementation would undermine other actions and lead to more, not less greenhouse gas output. That seems far-fetched since, at least among policy makers, nobody believes that geoengineering offers anything but a relatively short stopgap to buy time for other action."

(Bunzl 2009, p. 2)

The above quotes show, again, how dearly connected the idea of decarbonizing society is to the idea of a finite CE-deployment. It is the finitude of CE deployment that correlates to the transition of society – a transition that is deemed to be both necessary as well as inevitable by most authors. CE deployment beyond any practically controllable time frame (like the BECCS-scenario in option B) might equally push a decarbonization of society into a distant future.

"The transition to a low-carbon society must not be delayed any longer. The climate crisis is already here and considerable CO_2 emission reductions must be made immediately. Developing and evaluating geoengineering will waste valuable time: this will not mean buying time, but rather spending or losing it. This is why the countries of the world cannot afford geoengineering. The discourse critical of geoengineering claims that geoengineering opportunities could result in the avoidance of CO_2 emission reductions and delay the necessary mitigation of climate change. Geoengineering is a way of postponing the unavoidable structural change of contemporary society."

(Anshelm and Hansson 2014, p. 142)

The moral relevance of not imposing a dilemma on future generations has been established in light of the mitigation obligation (Section 2.2). Postponing the needed

decarbonization of our society might constitute an equally inadmissible risk transfer to future generations.

Another problem related to intergenerational justice is connected to carbon storage in the BECCS scenario. Confronted with deep uncertainty, current decision-makers cannot oversee all possible incidents and side effects related to carbon storage. In this sense, BECCS passes down the buck to future generations. It defuses the current problem of CO_2 emissions into a future problem of artificial carbon sinks.

In this light, BECCS will only buy time in a very diffuse and speculative sense. In order to be more than a lip service, I urge that the BT-argument should include requirements about both finitude and controllability.

This summarizes the considerations about an alternative, weak version of the BTargument. A weak BT-argument would not incorporate any finitude requirement and would allow for (some form of) substitution. I have argued that the special case of BECCS as a means to close the emission gap, would not count as permissible substitution, due to its magnitude – it would not be plausible even under a weak BTargument. Furthermore, I have argued that the finitude thesis is dearly connected to one crucial assumption of the BT-framing: the transition anyway thesis. Refraining from the finitude requirement might influence potential policy choices as to delay a needed decarbonization of society. While a non-finite BECCS deployment might still be morally admissible, if not plain necessary, it is no 'buying time' deployment as of my understanding. The weak version of the buying time argument is not an alternative, but no buying time argument at all.

6.2 General moral constraints

Two plausible instantiations of the (strong) BT-argument have been identified: argument a2 and a3. Both arguments assume drastic mitigation ensuring CO_2 emissions to remain within the two-degree budget with additional SAI deployment to limit the rate of temperature change. The evaluation of the arguments, however, was not completed as to include the last thesis – the morality thesis. This shall be done in the next section. I will discuss two general moral arguments against CE-deployment specified for the SAI scenarios at hand: the hubris argument and the techno-fix argument.

6.2.1 Hubris argument

One general moral argument against the deployment of CE has been brought forward from the viewpoint of virtue ethics. The hubris or playing-god argument states that we should not engage in any CE scheme, for the scope of the endeavor is just beyond our human understanding. Trying to control the climate system, even for a limited period of time, would be a sign of hubris, in that it ignores the role we humans play on this planet. The hubris-argument has been one of the earliest virtue ethical arguments against the deployment of CE (Jamieson 1996, Ott 2010, Gardiner 2010, Owen 2014).

A common perception of the hubris framing is that man should not aim at interfering with a given natural order. Order in nature is more than arbitrary; it is in itself valuable and hence refers to – if not a divine being – a metaphysical system. Such a version of the hubris argument lacks an essential deductive step from any sort of metaphysical order to the fact that we as humans need to respect it (Levine 2014). While the accusation of hubris comes easy, only a handful of philosophers have advanced this argument to the full. A very recent exception is the work of Meyer and Uhle (2014), in which they try to justify the virtue ethical heuristic for the evaluation of CE. Their definition of hubris reads like this:

"Persons (or groups of persons) show hubris, if they act with a reprehensible overestimation of their abilities."

(Meyer and Uhle 2015, p. 5)

The authors suggest that this concept might serve as a *heuristic* for the topic of CE, in that is highlights certain aspects of technology deployment that otherwise would be hidden. Hubris as defined by Meyer and Uhle (2015) does not rely on any metaphysical order. Rather, it is connected to *epistemic virtues* of what we can possibly know. From this perspective, hubris ought to be avoided because it presents a failure and a vice: The failure lies in the misjudgment of the *probability* of CE research results, the vice lies in the recklessness of advocating controllability of CE despite the epistemic failure.

Misjudgment about probabilities is in itself not reprehensible, but rather inevitable and quite common among human beings. Scientists know rather well that the scope of their models is limited, that the climate system is too complex to fully understand and that there are still numerous unknown factors that may lead to unintended sideeffects when pursuing intentional climate manipulation.

But as Levine (2014) has stated recently, those concerns could be best described as a consequentialist argument. Such an argument would not rely on a virtue ethical heuristic, but rather would include precautionary thinking when irreducible uncertainties are at stake.

Thusly understood, the argument reconstruction that has been proposed by Betz and Cacean (2012, p. 120) could serve to capture the hubris-framing in a consequentialist manner:

Irreducible Uncertainties

- 1. There are major irreducible uncertainties regarding the effectiveness and side effects of CE deployment.
- 2. Irreducible uncertainties cannot be reduced through further $R\&D^{38}$.
- If uncertainties regarding the effectiveness and side effects cannot be reduced, neither can effectiveness be guaranteed nor can catastrophic side effects be excluded.
- THUS: It is not true that: Further R&D into the CE technology T may (a) ensure its effectiveness and (b) exclude catastrophic side effects of its deployment.

This argument can be used to object the effectiveness thesis of the BT-argument. It shows, that the hubris argument can be re-formulated as a prudent consequentialist argument under uncertainty that asks us to obtain from any consequences that might have harmful consequences and that we cannot decide upon. Many scholars would either endorse this argument or criticize it on grounds of the empirical premises (1) and (2). Evaluation of those two premises then would constitute the main focus of future debate.

Having said this, I wish to highlight a different aspect of the argument, which goes beyond the evaluation of its empirical premises and relates to the hubris heuristic proposed by Meyer and Uhle (2014). While the trouble with uncertainties surrounding CE can be captured via the consequentialist argument above, within the

³⁸ R&D stands for research and development.

hubris framing, the misjudgment of the boundaries of our knowledge is not simply an epistemic failure, but a vice – it is reckless!

The recklessness lies in the fact that *against better judgment* the knowledge gap is ignored (Meyer and Uhle 2014, p. 6). The virtue ethical framing of the uncertainty argument urges us to be more moderate, even humble about our knowledge of the natural system and our ability to control it – and the lack thereof. It urges us to weigh our lack of knowledge in those delicate systems higher than the amount of what we actually do know – because it may be dangerous, if we don't. Possible dangerous side effect of CE deployment might arise because of a specific mind-set, which overestimates what we know and underestimates what we don't. So in essence, the hubris framing suggests a certain judgment about our gap of knowledge. This judgment, the internal assessment of the data we have, is determined by (epistemic) virtues. So, the virtue ethical argument of hubris asks us to judge our knowledge in a specific way. It hence is not an alternative to the consequentialist 'irreducible uncertainty, but it asks us to *weigh* this uncertainty in a specific way. There is no need to appeal to a metaphysical preorder to make the hubris argument work.

While there are rational decision principles for uncertain settings (like the precautionary principle), the hubris framing as a virtue ethical argument urges us to make those precautionary considerations a priority when weighting the arguments pro and con CE deployment. In this sense, it serves as a meta-argumentative guideline about the status of arguments, in that it weighs the argument about irreducible uncertainty stronger than other arguments, like arguments about effectiveness.

A hubristic SAI deployment could be avoided, if the reckless ignorance of uncertainties is ruled out. MacMartin and Keith (2014), the authors of the SAI-deployment scenario adopted in arguments a2 and a3, seem to think that they have done so. They include a learning curve and ask SAI deployment to be responsive, so that new information can lead to a modification or even the decline of SAI deployment³⁹. While the hubris-argument has moral force in evaluating SAI deployment, it might arguably be met by a careful and modest deployment-scheme –

³⁹ If SAI deployment was modified, because observation has suggested inacceptable impacts, the responsibility for those impacts of past deployment still remains. If liability is not determined, even a learning curve might not guarantee fair and beneficial SAI deployment.

the hubris argument is not a general moral constraint to plausible BT-deployment scenarios.

6.2.2 Techno-fix argument

An alternative framing of the buying time idea refers to CE as the 'second-best solution' for climate change. It states that if there is a problem, but our first choice is not at our disposal (yet), we need to adopt another option, given that second solution has no moral restrictions. In our daily lives, we act on base of such a principle of the second-best choice frequently. In political contexts and collective decision-making, the same principle can and must apply. A special version of this principle arises in cases, where social issues cannot be settled by social means alone and a technical solution to those problems seems apt.

Weinberg (1991) has concerned himself with those 'technological fixes' to social problems. His question was: "To what extend can social problems be circumvented by reducing them to technological problems?" (Weinberg 1991, p. 42).

Weinberg deems the reframing or redefining of some social problems as technical ones to be very promising. Social problems tend to be very complex and solving them might require changes at the individual behavioral level of members of society, which is hard to achieve. On the other hand, technological fixes are clear cut, easy and seemingly effective solutions to some social problems. They can be applied independent of individuals. As an example for a beneficial technological fix, Weinberg mentions the introduction of the seat belt against car accident deaths.

He is, however, very clear about the limitation of those technological fixes. Social problems need to be solved by social means eventually, and technological fixes are always a second best choice.

"Technology will never replace social engineering. But technology has provided and will continue to provide to the social engineer broader options, to make intractable social problems less intractable; perhaps, most of all, technology will buy time – that precious commodity that converts violent social revolution into acceptable social evolution."

(Weinberg 1991, p. 48)

For Weinberg one of the most beneficial features of a technological fix is that it *may buy more time for social transition*. The time generating quality of technological 121

fixes lies at the heart of the BT-argument in favor of CE, as has been demonstrated thoroughly. So, the technological fix framing is related to the buying time argument. But it is also clear from the above quote that a technological fix cannot serve as a substitute for a social solution – the same has been shown for CE and mitigation. Scott has dubbed the fear associated with a misinterpreted technological fix the 'techno-fix'-objection (Scott 2012): The reckless use of CE, without dealing with the underlying social problem, or even worsening it. This is the danger that lies within the technological fix argument and arguably also with the buying time argument, if the BT-requirements are not met or watered down, and CE is used as a substitute for mitigation (Section 6.1).

Some form of SAI deployment might be used as a stopgap measure to buy more time for social transition – this is what the BT-argument shows. But when does the BT-argument support a 'cheap techno-fix'? Techno-fixes are deemed to be a way of delaying the problem, not solving it, since they do not address the root cause of the problem, and advocates of the techno-fix framing would reject CE deployment as inherently flawed.

"[Solutions] to environmental problems like climate change could never be powerful technologies like SRM. Western societies must challenge their fundamental values and worldview to find solution to climate change."

(Scott 2012, p. 161)

This 'western worldview' that ought to be changed, arguably amounts to the notion that business as usual can be continued and that our modes of production and consumption can be upheld, though possibly by different means. Concepts like green growth or sustainable consumption are accompanied with the promise that a lifestyle change is not needed, but rather that economic, yet sustainable growth guarantees perpetual, environmental friendly wealth. The notion of 'green growth' has been challenged frequently. Brand calls green growth the "next oxymoron" (Brand 2012), and Grunwald argues that sustainable consumerism is no silver bullet to "save the environment", but instead that the paradigm of consumption and growth has to be changed altogether by political and legal means (Grunwald 2015). A different economic setup like the degrowth framing could enable a comprehensive technology assessment of such technologies, in order to rule out the naïve technophile believe

that technological progress can take center stage in solving the ecological crisis at hand (Grunwald 2016).

This line of thought resonates with the idea of Section 6.1, which identified the transition anyway thesis to be essential to the buying time argument, in that a transition of society mustn't be delayed by possible CE deployment. Only if a fundamental transition of society is underway, CE-deployment might not be suspected to be a 'techno-fix'. The finitude requirement in the BT-argument serves to ensure this, as it both assumes and enables a general transition of modes of production and consumption. If it can be plausibly met, BT-deployment of CE might not be object to the general suspicion of 'techno-fixing' our way out of climate change⁴⁰.

The techno-fix objection is a general critique of the attitudes supposedly visible in CE deployment. In this view, CE deployment would in each case be an inadmissible delay of real solutions to climate change, namely fast and far reaching mitigation. If, on the other hand, it were possible to reach the two-degree target with mitigation alone, CE might indeed not be needed at all:

"(The discourse critical of CE) emphasizes that there is no need for geoengineering: existing technologies and renewables would surely be sufficient to considerably decrease CO_2 emissions if they were deployed worldwide. However, this would require radical systemic change, which the governments of the world are unwilling or unable to accept. This is where geoengineering presents itself as a pseudo solution, maintaining the status quo or, in the long run, possibly even worsening the global climate crisis"

(Anshelm 2014, p. 139)

However, especially in light of sufficient mitigation, a buying time deployment can be justified. As I have argued in Chapter 3, the BT-argument also addresses the secondary climate goal of reducing pressure. Arguably, reducing the pressure for mitigation only then makes sense, when mitigation is indeed undertaken.

⁴⁰ While I have frequently used the terms decarbonization, transition of society and the like, decarbonization arguably does not necessarily include a redistribution of wealth, or generate a just form of production. To further substantiate the techno-fix argument, different assessment patterns could be applied. The degrowth framing might be helpful, when evaluating CE in front of different social and economic assumptions (Muraca and Neuber 2017).

Paradoxically, it might not be valid to reject CE deployment in face of sufficient mitigation.

To sum up, the techno-fix argument does not object the plausible SAI-instantiations of the BT-argument (argument a2 and a3), because 1) the design of a plausible BT-deployment incorporates a transformation towards decarbonization and 2) CE in this instantiation aims at realizing the secondary climate goal of reducing pressure.

6.3 Results

The possibility of acceptable substitution between BECCS and mitigation has been examined in Section 6.1. Due to the magnitude of BECCS deployment in order to close the emission gap visible in current mitigation pledges, BECCS would have to be deployed large-scale. The associated side effects, uncertainties and risks rule out that large-scale BECCS can count as a perfect substitute for mitigation. Even a weak version of the BT-argument, which would allow for some form of substitution, does not become plausible for climate option B (BECCS and insufficient mitigation).

On the other hand, two general moral constraints, the hubris argument and the techno-fix argument, were shown to not automatically lead to a rejection of the plausible instantiations of the strong BT-argument. Regarding the plausible instantiations of the BT-argument, the 'techno-fix' objection can be met if it is made sure that:

- 1. No decline in current mitigation efforts takes place;
- 2. Serious mitigation as part of a shift in society is already underway;
- 3. The termination of CE at a defined point in time will be guaranteed.

Those are arguably quite ambitious goals. But if the BT-argument is to work, it should not leave room for empty promises. This leads to the tentative conclusion: Beneficial and acceptable BT-deployment of CE technologies might theoretically be possible, but the practical limitations and the imperfect moral nature of our political decisions set the bar fairly high.

Some form of negative emission technologies might be necessary to reach the twodegree target. We might find ourselves in the situation where we might have to use BECCS (and even SAI) in order to manage climate change and avoid catastrophic impacts. This might even be done in a beneficial manner. But we shouldn't fool ourselves by calling it a buying time deployment. In this case, CE would no longer serve as a stopgap measure, as so many authors would like to believe, but as a substitute for our failure to change our ways of production and consumption in time. The euphemism 'buying time' might then only serve to make us feel better, but it should not deceive about the fact that we have existentially failed in our mission to live on this planet without 'fouling our nest'.

Chapter 7 Conclusion and Outlook

In the research thesis at hand, I have evaluated the buying time argument in favor of CE-deployment. I was able to identify two plausible instantiations of the argument with respect to SAI deployment and have discussed the option of beneficial BECCS deployment as a means to buy time. The main results can be briefly summarized as follows:

- Moderate, finite SAI can be a plausible instantiation of the BT-argument.
- Only if mitigation efforts guarantee the realization of the two-degree target does the BT-argument become plausible. The BT-argument becomes implausible if its central assumption about decarbonization is dropped or watered down.
- BECCS in order to reduce mitigation pressure in light of insufficient mitigation is not an instance of the BT-argument, as it violates the finitude requirement as well as the no-impediment requirement.
- Such BECCS deployment could nevertheless be acceptable as well as plain necessary it just will not be an instantiation of the BT-argument.
- A weak version of the BT-argument is no BT-argument at all. The finitude requirement as well as the no-impediment requirement, which both depend on the assumption of decarbonizing society, are essential to the buying time idea.

This brings us back to the initial research questions: Can climate engineering help provide more time for an ambitious mitigation program? And if so, is a buying time deployment of climate engineering morally acceptable? The answer to the first question must be no. A strong version of the buying time argument becomes plausible only if serious mitigation is already underway. CE cannot buy time for an ambitious *mitigation* program, as exactly this mitigation program must be presupposed for the BT-argument to become plausible. CE (SAI specifically) can merely help buy time for *adaptation*, as it might reduce the rate of global temperature change. This result indicates that the buying time framing of CE is overly optimistic and deceives about its underlying assumptions. While a naïve BT-framing might make the evaluation of CE *in general* more positive (Section 1.1), a scrutinized BT-argument is very clear about the limitations, difficulties and drawbacks of a possible BT-deployment.

The answer to the second question, however, is yes. There are at least two instantiations of the BT-argument that are both plausible and morally acceptable. This result should not be overstrained, though. The two plausible instantiations of the BT-argument (argument a2 and a3) are rather limited in their scope and length of time, and assume drastic emission cuts that remain within the two-degree budget (without an overshoot).

A large number of questions still remain unresolved:

- There is the need for further research into BECCS scenarios that fully resonate with the BT-approach. Under what circumstances can BECCS realize its associated goal to reduce mitigation pressure without turning into a mere substitute for mitigation? Under what circumstances is BECCS only a 'cheap techno-fix', and under what circumstances a helpful stopgap measure?
- Research on moderate and realistic SAI deployment should equally continue. Particular attention should be given to the legal, including mechanisms to ensure compliance with the no impediment-requirement (that is, continued mitigation efforts even in light of SAI).
- Similarly, I have argued that the techno-fix-objection can be met, if mitigation efforts stay constantly high and the termination of CE is ensured. To guarantee this, certain legal and political measures ought to be established. A climate engineering treaty that involves conditions for CE deployment might be conceivable.
- Establishing a strong mitigation obligation continues to be the focus of climate ethics. Particularily in light of the BT-argument, the mitigation obligation can further be strengthened. A connection to the trade-off argument (see section 2.3.) in light of the BT-argument can also be drawn and should be further explored.
- The virtue ethical stance and the weighting of arguments play an important role in assessing the BT-argument. Research here might prove to be of interest for argumentation theory in general as well as practical philosophy.

There are numerous links to current philosophical research areas, which I will selectively highlight below:

Argumentation theory. I have written above that the virtue ethical hubris argument against CE deployment urges us to be moderate, even humble about our knowledge of natural systems, and especially the lack thereof. The virtue ethical framing of the hubris

argument gives a special weight to the argument about irreducible uncertainties (Section 6.2.1). In this sense, it might serve as a meta-ethical guideline about the status of arguments. This opens up a new line of thoughts: Which role do attitudes play in weighing reasons?

A meta-theoretical argument-principle to incorporate this idea could be established. It would be applicable to the theory of dialectical structures (Betz 2010). Such a principle could reside on the macro-level of the debate, as it would describe a weighting-relation between (groups of) arguments.

(Weighting-principle) Given attitude H in debate τ , argument a1 in τ is preferred to argument a2 in τ .

Attitude H is in essence a preference relation on a given dialectical structure τ (for the nomenclature see Section 1.2). It might go further than stance attribution (Betz 2009), and might also give a different insight than the concept of degree of justification (Betz 2012b). The degree of justification is an internal assessment mechanism that yields a numerical result for given finite dialectical structure. The preference relation H, on the other hand, might serve as a qualitative measure, taking the semantic structure of the arguments into account. H might then be further specified, determining its attributes like transitivity and completeness.

Such research could be part of the ongoing extension of knowledge in argumentation theory.

Virtue ethics. Naturally, such meta-arguments belong to the realm of virtue ethics. Virtue ethics seem to play an important role in a number of fields of practical ethics, such as environmental ethics (Di Paola 2015).

If what has been demonstrated in terms of the hubris argument is true, virtue ethical arguments might serve as meta-arguments that enable proponents of a debate to weight single arguments in a certain way. If the hubris-argument in its meta-argumentative form is believed to be true, arguments about uncertainty gain *prima facie* more force, than, say, effectiveness considerations.

However, when does a proponent adhere to such virtue ethical meta-considerations? The problem about virtue ethical arguments is the question of how convincing they are to a person who does not share the same virtue ethical intuitions as the person advancing the respective virtue ethical argument (Levine 2014). Which virtue ethical argument a person finds compelling arguably depends on her attitude (*hexis* in Aristotelian terms). This does not mean that attitudes are arbitrary or that they belong to the realm of a kind of moral 'anything goes'. How far one can argue in favor of certain attitudes has been of focus in recent philosophical research (e.g. Kurbacher 2017).

Education. The discussion of attitudes, or of *hexis*, is part of current philosophical research. Attitudes shape our evaluation patterns, they influence our thinking at a very fundamental level, and eventually enable us to formulate a judgment. Certainly, attitudes are not arbitrary, and desirable attitudes ought to be identified. But how does one *obtain* one's attitude? How does one argue in favor of a certain attitude? And lastly, if it is true that attitudes are rather subconscious, and in a way inaccessible to rational thinking, how can attitudes even be addressed through reasoning? Eventually, this might transgress the realm of philosophical research and might fall into the practical sphere of education. Successful education can enable the individual to formulate well-founded judgments developing her powers to decide on which attitudes she adopts. Hence, a virtue-ethical approach might also be concerned with the question of education, both on the theoretical as well as on the practical level.

Degrowth. One central assumption of the buying time argument is the inevitable decarbonization of society. However, in what terms this change in production and consumption is to come about is undetermined in this argument. In particular, further political implications of decarbonization are not addressed, such as the fair distribution of wealth. Shifting current economies towards net zero does not necessarily imply that such economies will be more just, for example regarding the standard of living, or that they will have a higher level of nature conservation or be more environmentally friendly. Pollution and exploitation of the vulnerable (the environment, animals, native people or people of the third world) might still occur in a global carbon-neutral economy. This gives rise to concern. As I have argued elsewhere (Muraca and Neuber 2017), a strong mitigation approach ought to be embedded in a comprehensive scheme of transforming society. Questions about distributional justice in mitigation ought to be addressed as well as environmental

issues. This also resonates with the techno-fix objection against CE (Section 6.2.2), which warns against a delayed shift in society, if CE is applied.

A promising framework to address those issues is arguably the degrowth approach. Certain well-defined criteria can be established to assess the deployment of CE-technologies, such as the viability-criteria and the conviviality-criteria (Muraca and Neuber 2017). An assessment of CE in light of an alternative social and economic background might prove to provide new insights. After all, it is conventional economic thinking which provokes a number of fundamental arguments against CE such as the techno-fix argument. The need for a shift in economic paradigms might be due in order to cope with the specific difficulties posed by CE. As the saying goes: Problems cannot be solved by the same level of thinking that created them.

References

- Anand, P. (1995). *Foundations of Rational Choice Under Risk*. Oxford: Oxford University Press.
- Anshelm, J. & Hansson, A. (2014). Battling Promethean dreams and Trojan horses: Revealing the critical discourses of geoengineering. Energy Research & Social Science, 2, pp. 135-144.
- Armeni, C. & Redgwell, C. (2015). International legal and regulatory issues of climate geoengineering governance: rethinking the approach. Climate Geoengineering Governance Working Paper Series: 021. Retrieved from: http://www.geoengineering-governance-research.org/cgg-working-papers.php.
- Baatz, C. & Ott, K. (2016). Why Aggressive Mitigation Must Be Part of Any Pathway to Climate Justice. In C. Preston (Ed.), Climate Justice and Geoengineering. Ethics and Policy in the Atmospheric Anthropocene (pp. 93-108). London, New York : Rowman & Littlefield.
- Baatz, C. (2016). Can We Have It Both Ways? On Potential Trade-Offs Between Mitigation and Solar Radiation Management. Environmental Values, 25/1, pp. 29-49.
- Barrett, S. (2008). *The incredible economics of geoengineering*. Environmental and Resource Economics, 39, pp. 45-54.
- Bauer, N. (2005). Carbon Capture and Sequestration: An Option to Buy Time? PhD-Thesis, Potsdam University Germany. Retrieved from: <u>https://www.researchgate.net/publication/291383259_Carbon_capture_and_sequestr</u> ation_an_option_to_buy_time.
- Betz, G. (2009). *Evaluating Dialectical Structures*. Journal for Philosophical Logic, 38, pp. 283-312.
- Betz, G. (2010). Theorie dialektischer Strukturen. Frankfurt a.M.: Klostermann.
- Betz, G. (2012a). *The case for climate engineering research: An analysis of the 'arm the future' argument*. Climatic Change, 111, pp. 473-485.
- Betz, G. (2012b). On Degrees of Justification. Erkenntnis, 77, pp. 237-272.
- Betz, G. (2015). Are climate models credible worlds? Prospects and limitations of possibilistic climate prediction. European Journal for Philosophy of Science, 5, pp. 191-215.
- Betz, G. & Cacean, S. (2012). *Ethical Aspects of Climate Engineering*. Karlsruhe: KIT Scientific Publishing.
- Betz, G. & Brun, G. (2016). Analyzing Practical Argumentation. In S.O. Hansson & G. Hirsch Hadorn (Ed.), The Argumentative Turn in Policy Analysis. Reasoning about Uncertainty (pp. 39-77). Springer International Publishing.
- Bodansky, D. (2016). *The Legal Character of the Paris Agreement*. RECIEL, 25, pp. 142-150.
- Boucher, P. & Gough, C. (2012). *Mapping the ethical landscape of carbon capture and storage*. Poiesis & Praxis, 9, pp. 249-270.
- Boucher, O., Forster, P., Gruber, N., Ha-Duong, M., Lawrence, M., Lenton, T., Maas, A., & Vaughan, N. (2014) *Rethinking climate engineering categorization in the context of climate change mitigation and adaptation*. WIREs Climate Change, 5/1, pp. 23-35.
- Brand, U. (2012). *After Sustainable Development: Green Economy as the Next Oxymoron?* GAIA, 21, pp. 28-32.
- Bundesministerium für Wirtschaft und Energie (BMWE) (2017). *Energiedaten: Gesamtausgabe*. Retrieved from: <u>http://www.bmwi.de/Redaktion/DE/Artikel/Energie/energiedaten-gesamtausgabe.html</u>.
- Bunzl, M. (2009): *Researching geoengineering: should not or could not?* Environmental Research Letters, 4, 045104 (3pp).
- Canadell, J.G., & Schulze, E.D. (2014). *Global potential of biospheric carbon management for climate mitigation*. Nature Communications, 5, doi:10.1038/ncomms6282.
- Chen, C., & Tavoni, M. (2013). *Direct air capture of CO₂ and climate stabilization: A model based assessment*. Climatic Change, 118, pp. 59-72.
- Climate Action Tracker (CAT), 2017. Colaboration of Ecofys, Climate Analytics, NewClimate Institute and Potsdam Institute for Climate Impact Research. Team: Bill Hare, Niklas Höhne, Prof. Kornelis Blok. Online <u>http://climateactiontracker.org/</u> (last accsess 10.07.2017)
- Corner, A., & Pidgeon, N. (2010). *Geoengineering the Climate: The Social and Ethical Implications*. Environment, 52, pp. 24-37.
- Crutzen, P. (2006). *Albedo Enhancement by Stratospheric Sulfur Injections: A Contribution to Resolve a Policy Dilemma?* Climatic Change, 77, pp. 211-220.
- Dieckhoff, C., & Leuschner, A. (Eds.) (2016). *Die Energiewende und ihre Modelle. Was uns Energieszenarien sagen können - und was nicht.* Bielefeld: transcript Verlag.
- Edmonds, J., Luckow, P., Calvin, K., & Wise, M. (2013). *Can radiative forcing be limited to 2.6 Wm⁻² without negative emissions from bioenergy and CO₂ capture and storage*? Climatic Change, 118/1, pp. 29-43.

- Effiong, U., & Neitzel, R.L. (2016). Assessing the direct occupational and public health impacts of solar radiation management with stratospheric aerosols. Environmental Health, 15, pp. 7-16.
- Elliott, K. (2016). Climate Geoengieering. In S.O. Hansson & G. Hirsch Hadorn (Eds.) The Argumentative Turn in Policy Analysis. Reasoning about Uncertainty (pp. 305-324). Springer International Publishing.
- Fleming, J.R. (2010). *Fixing the Sky: Checkered History of Weather and Climate Control*. New York: Columbia University Press.
- Fuss, S.; Canadell, J.; Peters, G., Tavoni, M., Andrew, R. M., Clais, P., Jackson, R. B. et al. (2014). *Betting on negative emissions*. Nature Climate Change, 4, pp. 850–853.
- Gardiner, S. (2010). Is 'Arming the Future' with Geoengineering Really the Lesser Evil? Some Doubts About the Ethics of Intentionally Manipulating the Climate System (p. 284-312). In Gardiner, S., Caney, S., Jamieson, D., & Shue, H. (Eds.), Climate Ethics: Essential Readings, Oxford: Oxford University Press.
- Gardiner, S. (2012). Are we the Scum of the Earth? Climate Change, Geoengineering and Humanity's Challenge. In A. Thompson & J. Bendik-Keymer (Eds.), Ethical Adaptation to Climate Change. Human Virtues of the Future (pp. 241-260). Cambridge, London: MIT Press.
- Gardiner, S. (2013). Geoengineering and Moral Schizophrenia: What Is the Question? In W. Burns & A. Strauss (Eds.), Climate Change Geoengineering: Legal, Political and Philosophical Perspectives (pp. 11-38). Cambridge: Cambridge University Press.
- Goodell, J. (2010). *How to Cool the Planet: Geoengineering and the Audacious Quest to Fix Earth's Climate.* New York: Houghton Mifflin.
- Grunwald, A. (2008). Technik und Politikberatung. Frankfurt am Main: Shurkamp.
- Grunwald, A. (2009). Technology assessment: concepts and methods. In Meijers, A.
 (Ed.), Philosophy of Technology and Engineering Sciences (pp. 1103-1146), vol. 9.
 Amsterdam: North Holland.
- Grunwald, A. (2015). Ende einer Illusion: Warum ökologisch korrekter Konsum die Umwelt nicht retten kann. oecom verlag.
- Grunwald, A. (2016). *Diverging pathways to overcoming the environmental crisis: A critique of eco-modernism from a technology assessment perspective*. Journal of Cleaner Production, <u>http://dx.doi.org/10.1016/j.jclepro.2016.07.212</u>.
- Hale, B. (2012). The world that would have been. Moral Hazard Arguments against geongineering. In C. Preston (Ed.), Engineering the Climate: The Ethics of Solar Radiation Management (pp. 113-132). Lanham: Rowman and Littlefield.
- Hamilton, C. (2013). *Earthmasters: The Dawn of the Age of Climate Engineering*. New Haven: Yale University Press.

- Hampicke, U. (2000). Naturschutz ökonomisch gesehen. In K.-H. Erdmann, T. J. Mager. Innovative Ansätze zum Schutz der Natur. Visionen für die Zukunft (pp. 127-150). Berlin, Heidelberg: Springer-Verlag.
- Hansen, J., Sato, M., Hearty, P., Ruedy, R., Kelley, M., et al. (2016). *Ice melt, sea level rise and superstorms: evidence from paleoclimate data, climate modeling and modern observations that 2°C global warming could be dangerous*. Atmospheric Chemistry and Physics, 16, pp. 3761 3812.
- Hansson, S., Hirsch Hadorn, G., (2016a). *The Argumentative Turn in Policy Analysis*. *Reasoning about Uncertainty*. Springer International Publishing.
- Hansson, S.O., & Hirsch Hadorn, G. (2016b). Introducing the argumentative turn in policy analysis. In S.O. Hansson & G. Hirsch Hadorn (Eds.), The Argumentative Turn in Policy Analysis. Reasoning about Uncertainty (pp. 11-35). Springer International Publishing.
- Henchion, M., McCarthy, M., Resconi, V., & Troy, D. (2014). *Meat consumption: Trends and quality matters*. Meat Science, 98/3, pp. 561-568.
- Hester, R.E., & Harrison, R.M. (2014). *Geoengineering of the climate system*. Cambridge: Royal Society of Chemistry.
- Heyward, C. (2013). Situating and Abandoning Geoengineering: A Typology of Five Responses to Dangerous Climate Change. Political Science and Politics, 46/1, pp. 23-27.
- Horton, J. (2015). *The emergency framing of solar geoengineering: Time for a different approach*. The Anthropocene Review, pp. 1-5.
- Huttunen, S., & Hildén, M. (2013): *Framing the Controversial: Geoengineering in Academic Literature*. Science Communication, 36/1, pp. 3-29.
- IPCC (2007),; Pachauri, R.K; Reisinger, A., eds., Climate Change 2007: Synthesis Report, Contribution of Working Groups I, II and III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, IPCC, Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- IPCC (2013): Summary for Policymakers. In: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Stocker, T.F., D. Qin, G.-K. Plattner, M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P.M. Midgley (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- IPCC (2014a): Summary for Policymakers. In: Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Edenhofer, O., R. Pichs-Madruga, Y. Sokona, E. Farahani, S. Kadner, K. Seyboth, A. Adler, I. Baum, S. Brunner, P. Eickemeier, B. Kriemann, J. Savolainen, S. Schlömer, C. von Stechow,

T. Zwickel and J.C. Minx (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA

- IPCC (2014b): Summary for policymakers. In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Field, C.B., V.R. Barros, D.J. Dokken, K.J. Mach, M.D. Mastrandrea, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B. Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L.White (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, pp. 1-32.
- IPCC (2014c): Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Core Writing Team, R.K. Pachauri and L.A. Meyer (eds.)]. IPCC, Geneva, Switzerland, 151 pp.
- Jacquet, J., & Jamieson, D. (2016). *Soft but significant power in the Paris Agreement*. Nature Climate Change, 6, pp. 643–646.
- Jamieson, D. (1996). *Ethics and Intentional Climate Change*. Climatic Change, 33, pp. 323-336.
- Jones, A., J. Haywood, O. Boucher, B. Kravitz, & A. Robock (2010). Geoengineering by stratospheric SO2 injection: Results from the Met Office HadGEM2 climate model and comparison with the Goddard Institute for Space Studies ModelE. Atmospheric Chemistry and Physics, 10, pp. 5999-6006.
- Jones, A., Haywood, J., & Jones, A. (2016). *Climatic impacts of stratospheric geoengineering with sulfate, black carbon and titania injection*. Atmospheric Chemistry and Physics, 16, pp. 2843–2862.
- Kartha, S. & Dooley, K. (2016). *The risks of relying on tomorrow's 'negative emissions' to guide today's mitigation action*. Stockholm Environment Institute, Working Paper 2016-08. Retrieved from: https://www.sei-internationalorg.
- Keith, D. (2000). *Geoengineering The Climate: History and Prospects*. Annual Review of Energy and the Environment, 25, pp. 245-285.
- Keith, D., Ha-Duong, M., & Stolaroff, J. (2006). *Climate Strategy with Co2 Capture from the air*. Climatic Change, 74, pp. 17-45.
- Keith, D. (2013). A Case for Climate Engineering. Boston: Boston Review Books.
- Keith, D., MacMartin, D. (2015). *A temporary, moderate and responsive scenario for solar geoengineering*. Nature climate change, 5, pp. 201-206.
- Kling, G., Clark, M.A., Wagner, G.N., Compton, H.R., Humphrey, A.M., Devine, J.D., Evans, W.C., et al. (1987). *The 1986 Lake Nyos Gas Disaster in Cameroon, West Africa.* Science, 236/ 4798, pp. 169-175.

- Kriegler, E., Riahi, K., Bosetti, V., Capros, P, Petermann, N., van Vuuren, D., Weyant, J., Edenhofer, O. (2015). *Introduction to the AMPERE model intercomparison studies on the economics of climate stabilization*. Technological Forecasting and Social Change, 90, pp. 1-7.
- Kravitz, B., Robock, A., Oman, L., Stenchikov, G., & Marquardt, A. (2009). Sulfuric acid deposition from stratospheric geoengineering with sulfate aerosols. Journal of Geophysical Research: Atmospheres, 114, D14109 (7pp).
- Kravitz, B., MacMartin, D., & Caldeira, K. (2012). *Geoengineering: Whiter skies?* Geophysical Research Letters, 39/11, L11801 (6pp).
- Kravitz, B., Robock, A., Tilmes, S., Boucher, O., English, J.M., Irvine, P.J., Jones, A., et al. (2015). *The Geoengineering Model Intercomparison Project Phase 6* (*GeoMIP6*): simulation design and preliminary results. Geoscientific Model Development, 8, pp. 3379–3392
- Kurbacher, F.A. (2017). Zwischen Personen. Eine Philosophie der Haltung. Würzburg: Könighausen & Neumann.
- Lempert, R., Nakicenovic, N., Sarewitz, D., Schlesinger, M. (2004). Characterizing Climate-Change Uncertainties for Decision-Makers. An Editorial Essay. Climatic Change 65, pp. 1-9
- Lenton, T., Held, H., Kriegler, E., Hall, J., Lucht, W., Rahmstorf, S., & Schellnhuber, H.J. (2008). *Tipping elements in the Earth's climate system*. Proceedings of the National Academy of Sciences of the United States of America, 105/6, pp. 1786– 1793.
- Lenton, T., Vaugham, N. (Eds.) (2013). Geoengineering Responses to Climate Change. Selected Entries from the Encyclopedia of Sustainability Science and Technology. New York, Heidelberg, Dordrecht, London: Springer.
- Levine, G. L. (2014). *Has It Really Come to This? An Assessment of Virtue Ethical Approaches to Climate Engineering*. BA-Thesis, Yale University. Retrieved from http://politicalscience.yale.edu/sites/default/files/files/Levine Gabriel.pdf.
- Lin, A. (2009). *Geoengineering Governance*. Issues in Legal Scholarship, 8/3, Retrieved from <u>http://www.bepress.com/ils/vol8/iss3/art2</u>.
- Lin, A. (2013). *Does Geoengineering Present a Moral Hazard?* Ecology Law Quarterly, UC Davis Legal Studies Research Paper No. 312, pp. 673 712.
- Long, J. (2016). Bringing Geoengineering into the Mix of Climate Change Tools. In Preston, C. (Ed.) Climate Justice and Geoengineering. Ethics and Policy in the Atmospheric Anthropocene (pp. 109-120). London, New York : Rowman & Littlefield.

- Loukkanen, M., Huttunen, S., Hildén, M. (2013). Geoengineering, newsmedia and metaphors: framing the controversial Public Understanding of Science, 23/8, pp. 966–981.
- MacCracken, M. (2009). On the possible use of geoengineering to moderate specific climate change impacts. Environmental Research Letters, 4, 045107 (14pp).
- MacMartin DG, Caldeira K, Keith DW (2014): *Solar geoengineering to limit the rate of temperature change*. Philosophical Transactions of the Royal Society A, 372: 20140134 (13pp).
- McNutt, M., Abdalati, W., Caldeira, K., Doney, S. et al. 2015a. *Climate Intervention: Reflecting Sunlight to Cool Earth*. The National Academies of Sciences, Engineering, and Medicine, Washington DC.
- McNutt, M., Abdalati, W., Caldeira, K., Doney, S. et al. 2015b. *Climate Intervention: Carbon Dioxide Removal and Reliable Sequestration*. The National Academies of Sciences, Engineering, and Medicine, Washington DC.
- Medvecky, F., Lacey, J., & Ashworth, P. (2014). *Examining the Role of Carbon Capture and Storage Through an Ethical Lens*. Science and Engineering Ethics, 20/4, pp. 1111-1128.
- Meyer, K., Uhle, C. (2015): *Geoengineering and the Accusation of Hubris*. THESys Discussion Paper No. 2015-3. Humboldt-Universität zu Berlin, Berlin, Germany. Retreived from: edoc.hu-berlin.de/series/thesysdiscpapers.
- Morrow, D. (2014). *Ethical aspects of the mitigation obstruction argument against climate engineering research*. Philosophical Transactions of the Royal Society, 372:20140062.
- Moss R.H., Edmonds J.A., Hibbard K.A., Manning M.R., Rose S.K., van Vuuren D.P., Carter T.R. et al. (2010). *The next generation of scenarios for climate change research and assessment*. Nature 463, pp. 747-756.
- Muraca, B., Neuber, F. (2017). Viable and convivial technologies. Considerations on Climate Engineering from a degrowth perspective. Journal of Cleaner Production. DOI: 10.1016/j.jclepro.2017.04.159
- Muratori, M., Calvin, K., Wise, M., et al. (2016). *Global economic consequences of deploying bioenergy with carbon capture and storage (BECCS)*. Environmental Research Letters, 11, 095004.
- Nordhaus, W. (2008). A Question of Balance: Weighing the Options on Global Warming Policies. New Haven: Yale University Press.
- Ott, K. (2010). Argumente für und wider "Climate Engineering". Versuch einer Kartierung. Technikfolgenabschätzung Theorie und Praxis, 19, pp. 32-41.

- Ott, K. (2012). Might Solar Radiation Management Constitute a Dilemma? In C. Preston (Ed.), Engineering the Climate: The Ethics of Solar Radiation Management (pp. 33 – 42). Lanham: Rowman and Littlefield.
- Owen, R. (2014): *Solar Radiation Management and the Governance of Hubris*. Issues in Environmental Science and Technology, 38, pp. 212-248.
- di Paola, M. (2015). *Virtues for the Anthropocene*. Environmental Values, 24, pp. 183-207.
- Parson, E., Ernst, L. (2013): *International Governance of Climate Engineering*. Theoretical Inquiries in Law, 14/1, pp. 307-337.
- Pourhashem, G., Adler, P., Spatari, S. (2016). *Time effects of climate change mitigation strategies for second generation biofuels and co-products with temporary carbon storage*. Journal of Cleaner Production 112(4), pp. 2642–2653.
- Rasch, P. J., Tilmes, S., Turco, R.P., Robock, A., Oman, L., Chen, C.C., Stenchikov,
 G.L. & Garcia, R.R. (2008). *An overview of geoengineering the climate using stratospheric sulphate aerosols*. Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences, 366/1882, pp. 4007-4037.
- Raven, J. A., Caldeira, K., Elderfield, H., Hoegh-Guldberg, O., et al. (2005). Ocean acidification due to increasing atmospheric carbon dioxide. Royal Society, London, UK. Retrieved from: <u>https://royalsociety.org/topics-policy/publications/2005/oceanacidification/</u>.
- Rayner, S., Heyward, S., Kruger, T., Pidgeon, N., Redgwell, C., Savulescu, J. (2013). *The Oxford Principles*. Climatic Change, 121/3, pp. 499–512.
- Riahi, K., Kriegler, E., Johnson, N., Bertram, C., den Elzen, M., Eom, J., Schaeffer, M et al. (2015). Locked into Copenhagen pledges – Implications of short-term emission targets for the cost and feasibility of long-term climate goals. Technological Forecasting & Social Change 90, pp. 8–23.
- Rickels, W., Klepper, G., Dovern, J., Betz, G., Brachatzek, N., Cacean, S., Güssow, K.,
- Heintzenberg J., Hiller, S., Hoose, C., Leisner, T., Oschlies, A., Platt, U., Proelß, A., Renn, O., Schäfer, S., Zürn M., (2011). *Large-Scale Intentional Interventions into the Climate System? Assessing the Climate Engineering Debate*. Kiel: Kiel Earth Institute.
- Reynolds, J. (2014). *The International Regulation of Climate Engineering: Lessons from Nuclear Power*. Journal of Environmental Law, 26/2, pp. 269-289.
- Reynolds, J. (2015). *The International Legal Framework for Climate Engineering*. Geoengineering Our Climate? Ethics, Politics and Governance, Working paper Series. Retrieved from: <u>https://geoengineeringourclimate.com/2015/03/26/the-</u> international-legal-framework-for-climate-engineering-working-paper/

- Robock, A., 2008. 20 Reasons Why Geoengineering May be a Bad Idea. Bulletin of the Atomic Scientists 64, pp. 14-18.
- Robock, A., L. Oman, and G. L. Stenchikov. 2008. *Regional climate responses to geoengineering with tropical and Arctic SO₂ injections*. Journal of Geophysical Research: Atmospheres 113(D16).
- Robock, A.; Marquardt, A.; Kravitz, B.; Stenchikov, G. (2009). *Benefits, risks, and costs of stratospheric geoengineering*. Geophysical Research Letters, 36/19, L19703 (9pp).

Royal Society (2011): *Solar radiation management: The Governance of research*. Online: <u>https://royalsociety.org/~/media/Royal_Society_Content/policy/projects/solar-</u> radiation-governance/DES2391_SRMGI%20report_web.pdf

- Rogelj, J., Schaeffer, M., Friedlingstein, P., Gillett, N., van Vuuren, D., Riahi, K., Allen, M., & Knutti, R. (2016). *Differences between carbon budget estimates unravelled*. Nature Climate Change, 6, pp. 245-252.
- Schäfer, S., Maas, A., Stelzer, H., & Lawrence, M.G. (2014). Earth's future in the Anthropocene: Technological interventions between piecemeal and utopian social engineering. Earth's Future, 2, pp. 239–243
- Schäfer, S.; Lawrence, M. G.; Stelzer, H.; Born, W.; Low, S., (2015). *Removing Greenhouse Gases from the Atmosphere and Reflecting Sunlight away from Earth.* The European Transdisciplinary Assessment of Climate Engineering (EuTRACE), Potsdam.
- Schellnhuber, J., Messner, D., Leggewie, C. (2009), Kassensturz für den Weltklimavertrag - Der Budgetansatz. Wissenschaftlicher Beirat Der Bundesregierung Globale Umweltveränderungen, Sondergutachten. Retrieved from: http://www.wbgu.de/sg2009/
- Scott, D. (2012). Insurance policy or Technological Fix? The ethical implications of Framing Solar Radiation Management. In C. Preston (Ed.), Engineering the Climate. The Ethics of Solar Radiation Management (pp. 151-168). Lanham: Rowman and Littlefield.
- Shepherd, J. (2016): What does the Paris Agreement mean for geoengineering? In The Royal Society Blog. Retrieved from: <u>http://blogs.royalsociety.org/in-</u><u>verba/2016/02/17/what-does-the-paris-agreement-mean-for-geoengineering/,</u> accessed: 15.07.2017.
- Shue, H. (2010). Global Environment and International Inequality. In S.M. Gardiner, S. Caney, D. Jamieson, & H. Shue (Eds.), Climate Ethics: Essential Readings (pp. 101-111), Oxford: Oxford University press.
- Sillmann, J., Lenton, T.M., Ott, K., et al. (2015): *Climate emergencies do not justify engineering the climate*. Nature climate change, 5, pp. 290–292.

- Smith, L.J., Torn, M.S. (2013). Ecological limits to terrestrial biological carbon dioxide removal Climatic Change 118/1, pp. 89-103.
- Solomon, S., Plattner, G-K., Knutti, R., Friedlingstein, P. (2009). *Irreversible climate change due to carbon dioxide emissions*. Proceedings of the National Academy of Sciences of the United States of America, 106, pp. 1704–1709.
- Stocker, T.F., Schmittner, A. (1997). *Influence of CO*₂ emission rates on the stability of the thermohaline circulation. Nature 388, pp. 862-865.
- Tjiputra, J. F., A. Grini, H. Lee (2016). *Impact of idealized future stratospheric aerosol injection on the large-scale ocean and land carbon cycles*. Journal of Geophysical Research, 121, pp. 2-27.
- Uther, S. (2013): Diskurse des Climate Engineering Argumente, Akteure und Koalitionen in Deutschland und Großbritannien. Heidelberg, New York: Springer.
- United Nations Framework Convention on Climate Change (UNFCCC 1998). *Kyoto Protocol To The United Nations Framework Convention On Climate Change*. Retrieved from <u>https://unfccc.int/kyoto_protocol/items/2830.php</u>
- United Nations Framework Convention on Climate Change (UNFCCC 2015). Adoption of the Paris Agreement, 21st Conference of the Parties, Paris: United Nations. Retreived from <u>http://unfccc.int/paris_agreement/items/9485.php</u>
- UNFCCC 2016. Unites Nations Framework Convention on Climate Change, Conference of the Parties, Twenty-first session Paris, 30 November to 11 December 2015, "Adoption Of The Paris Agreement". http://unfccc.int/resource/docs/2015/cop21/eng/109.pdf/ (access 12.07.2016).
- van Vuuren, D.P., Edmonds, J., Kainuma, M. et al. (2011). *The representative concentration pathways: an overview*. Climatic Change, 109, pp. 5-31.
- van Vuuren, D., Deetman, S., et al. (2013): The role of negative CO₂ emissions for reaching 2 °C - insights from integrated assessment modeling. Climatic Change, 118, pp. 15-27.
- Wagner, G., Zeckhauser, R.J. (2012). *Climate policy: hard problem, soft thinking* Climatic Change, 110/3, pp. 507–521.
- Walton, D. (2001). *Abductive, presumptive and plausible arguments*. Informal Logic, 21/2, pp. 141-169.
- Weale, A., Perry, H., Brown, S., Dr Amanda Burls, Professor Robin Gill, Professor Sian Harding, Professor Ray Hill et al. (2011): Biofuels: ethical issues. Nuffield Council on Bioethics, Retrieved from: <u>http://nuffieldbioethics.org/project/biofuels-0</u> (access 29. June 2016).
- Weiss, Edith (1989): In Fairness to Future Generations: International Law, Common Patrimony, and Intergenerational Equity. Dobbs Ferry, N.Y.: Transnational Publishers.

- Weiss, E. (1992). Intergenerational Equity: A Legal Framework For Global Environmental Change. In E.B. Weiss (Ed.), Environmental Change And International Law: New Challenges And Dimensions (pp. 385-412), Tokyo: United Nations University Press.
- Weinberg, A. (1991). Can Technology Replace Social Engineering? In E. Katz, A. Light & W.B. Thompson (Eds.), Controlling technology: contemporary issues (pp. 41-48). Buffaloe, NY: Promethus books.
- Wennersten, R., Sun, Q., Li, H. (2015). *The future potential for Carbon Capture and Storage in climate change mitigation – an overview from perspectives of technology, economy and risk.* Journal of Cleaner Production 103, pp. 724-736.
- Whyte, K. (2012). Indigenous Peoples, Solar Radiation Management and Consent. In Preston, C. (Ed.), Engineering the Climate: The Ethics of Solar Radiation Management (pp. 65 – 76). Lanham: Rowman & Littlefield.
- Wigley, T.M.L; Richels, R; Edmonds, J A (1996). Economic and Environmental Choices in the Stabilization of Atmospheric CO2 Concentrations. Nature, pp. 240-243.
- Wigley, T. M. L. (2006). A combined mitigation/geoengineering approach to climate stabilization. Science, 314, pp. 452-454.
- Wissenschaftlicher Beirat der Bundesregierung Globale Umweltveränderung (WBGU) (2014). Zivilisatorischer Fortschritt innerhalb planetarischer Leitplanken – Ein Beitrag zur SDG-Debatte. Politikpapier 8, WBGU. Retrieved from http://www.wbgu.de/pp8/
- Zhanga, Z., Moore, J., Huisinghe, D., Zhao, Y. (2015). *Review of geoengineering approaches to mitigating climate change*. Journal of Cleaner Production, 103, pp. 898–907.
- Zürn, M., Schäfer, S. (2013). *The Paradox of Climate Engineering*. Global Policy, 4/3, pp. 223–324.